UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-Q

(Mark one)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE Х **ACT OF 1934**

For the quarterly period ended September 30, 2006

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE 0 **ACT OF 1934**

For the transition period from

Commission file number: 000-50307

to

FormFactor, Inc.

(Exact name of registrant as specified in its charter)

DELAWARE

(State or other jurisdiction of incorporation or organization)

13-3711155 (I.R.S. Employer Identification No.)

7005 Southfront Road, Livermore, California 94551 (Address of principal executive offices, including zip code)

(925) 290-4000

(Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Act: None

Securities registered pursuant to Section 12(g) of the Act: Common Stock

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer. See definition of "accelerated filer and large accelerated filer" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer \boxtimes

Accelerated filer o

Non-accelerated filer o

4

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No x

As of October 31, 2006, 46,781,211 shares of the registrant's common stock, par value \$0.001 were outstanding.

FORMFACTOR, INC.

FORM 10-Q FOR THE QUARTERLY PERIOD ENDED September 30, 2006

INDEX

Page **Financial Information** Part I. Item 1. Unaudited Condensed Consolidated Financial Statements: Unaudited Condensed Consolidated Statements of Income for the three and nine months ended September 30, 2006 and 3 September 24, 2005 Unaudited Condensed Consolidated Balance Sheets as of September 30, 2006 and December 31, 2005

	Unaudited Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2006 and	5
	September 24, 2005.	C
	Notes to Unaudited Condensed Consolidated Financial Statements	6
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	16
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	26
<u>Item 4.</u>	Controls and Procedures	26
<u>Part II.</u>	Other Information	
<u>Item 1.</u>	Legal Proceedings	27
Item 1A.	Risk Factors	27
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	30
<u>Item 3.</u>	Defaults Upon Senior Securities	30
<u>Item 4.</u>	Submission of Matters to a Vote of Security Holders	30
<u>Item 5.</u>	Other Information	30
<u>Item 6.</u>	Exhibits	31
<u>Signature</u>		32
<u>Exhibit Index</u>		33

2

PART I. FINANCIAL INFORMATION

Item 1. Unaudited Condensed Consolidated Financial Statements

FORMFACTOR, INC.

CONDENSED CONSOLIDATED STATEMENTS OF INCOME (In thousands, except per share amounts) (Unaudited)

	_	Three Months Ended September September			Nine Months Ended September Septemb			
		30, 2006		24, 2005		80, 2006		24, 2005
Revenues	\$	96,757	\$	62,374	\$	270,520	\$	165,676
Cost of revenues		46,492		34,088		130,699		93,484
Gross margin	_	50,265		28,286		139,821		72,192
Operating expenses:	—						_	
Research and development		11,994		7,881		33,397		19,461
Selling, general and administrative		19,321		11,871		53,034		31,283
Total operating expenses	_	31,315		19,752		86,431		50,744
Operating income	_	18,950		8,534		53,390		21,448
Interest income		4,485		1,116		10,196		2,912
Other income (expense), net		59		(630)		45		(655)
Income before income taxes	_	23,494		9,020		63,631		23,705
Benefit from (provision for) income taxes		(7,675)		758		(21,763)		(4,004)
Net income	\$	15,819	\$	9,778	\$	41,868	\$	19,701
Net income per share:	=						_	
Basic	\$	0.34	\$	0.25	\$	0.94	\$	0.50
Diluted	\$	0.33	\$	0.23	\$	0.90	\$	0.47
Weighted-average number of shares used in per share calculations:	=							
Basic		46,417		39,733		44,625		39,343
Diluted	_	48,494		41,762		46,690		41,492
	_		-				-	

The accompanying notes are an integral part of these condensed consolidated financial statements.

3

FORMFACTOR, INC.

CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, except share and per share amounts) (Unaudited)

	Sej	otember 30, 2006	De	cember 31, 2005
ASSETS				
Current assets:				
Cash and cash equivalents	\$	276,365	\$	31,217
Marketable securities		181,128		180,391

Accounts receivable, net of allowance for doubtful accounts of \$74 as of September 30, 2006 and December 31,			
2005	54,097		43,967
Inventories	26,317		18,404
Deferred tax assets	11,233		11,396
Prepaid expenses and other current assets	 12,674		7,169
Total current assets	561,814		292,544
Restricted cash	2,250		2,250
Property and equipment, net	87,398		81,588
Deferred tax assets	6,270		4,518
Other assets	994		461
Total assets	\$ 658,726	\$	381,361
LIABILITIES AND STOCKHOLDERS' EQUITY		_	
Current liabilities:			
Accounts payable	\$ 27,346	\$	26,369
Accrued liabilities	21,850		20,467
Income tax payable	7,653		9,697
Deferred rent	353		313
Deferred revenue and customer advances	6,556		3,588
Total current liabilities	63,758		60,434
Deferred rent and other long term liabilities	4,559		3,138
Total liabilities	68,317		63,572
Commitments and contingencies (Note 8)	 <u> </u>		
Stockholders' equity:			
Preferred stock, \$0.001 par value:			
10,000,000 shares authorized; no shares issued and outstanding at September 30, 2006 and December 31, 2005,			
respectively	_		
Common stock, \$0.001 par value:			
250,000,000 shares authorized; 46,761,252 and 40,236,686 shares issued and outstanding at September 30, 2006			
and December 31, 2005, respectively	47		40
Additional paid-in capital	496,296		268,291
Deferred stock-based compensation	_		(2,495)
Accumulated other comprehensive loss	(114)		(359)
Retained earnings	94,180		52,312
Total stockholders' equity	590,409		317,789
Total liabilities and stockholders' equity	\$ 658,726	\$	381,361

The accompanying notes are an integral part of these condensed consolidated financial statements.

4

FORMFACTOR, INC.

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

	Nine Mon	ths Ended
	September 30, 2006	September 24, 2005
Cash flows from operating activities:		
Net income	\$ 41,868	\$ 19,701
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	15,919	12,184
Stock-based compensation expense	15,015	3,011
Deferred income taxes	(1,650)	(4,262)
Increase in allowance for doubtful accounts	—	33
Tax benefit from employee stock option plans	—	4,379
Excess tax benefits from equity based compensation plans	(13,448)	—
Provision for excess and obsolete inventories	9,222	8,085
Loss on disposal of property and equipment	378	80
Changes in assets and liabilities:		
Accounts receivable	(10,122)	(10,447)
Inventories	(16,382)	(11,270)
Prepaid expenses and other current assets	(5,472)	(2,859)
Other assets	(599)	
Accounts payable	4,867	(4,138)
Accrued liabilities	16,853	(320)
Income tax payable	(2,038)	5,896
Deferred rent	1,464	862
Deferred revenues and customer advances	2,968	1,183
Net cash provided by operating activities	58,843	22,118
Cash flows from investing activities:		
Acquisition of property and equipment	(26,862)	(21,275)

Purchase of marketable securities	(202,929)	(129,135)
Proceeds from sales and maturities of marketable securities	201,340	106,910
Acquisition of intangible research and development asset	—	(400)
Net cash used in investing activities	 (28,451)	(43,900)
Cash flows from financing activities:		
Proceeds from issuances of common stock	201,289	10,562
Excess tax benefits from equity based compensation plans	13,448	
Net cash provided by financing activities	 214,737	10,562
Effect of exchange rate changes on cash and cash equivalents	 19	74
Net increase (decrease) in cash and cash equivalents	245,148	(11,146)
Cash and cash equivalents, beginning of the period	31,217	34,836
Cash and cash equivalents, end of the period	\$ 276,365	\$ 23,690
Supplemental disclosure of significant non-cash investing activities:		
Change in accounts payable and accrued liabilities for purchase of property and equipment	\$ (5,949)	\$ 5,443

The accompanying notes are an integral part of these condensed consolidated financial statements.

5

FORMFACTOR, INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 1 — Basis of Presentation

Basis of presentation. The accompanying unaudited condensed consolidated financial statements of FormFactor, Inc. and its subsidiaries (the "Company") have been prepared in accordance with accounting principles generally accepted in the United States of America and pursuant to the instructions to Form 10-Q and Article 10 of Regulation S-X of the Securities and Exchange Commission ("the SEC"). Accordingly, the interim financial statements do not include all of the information and footnotes required by generally accepted accounting principles for annual financial statements. In the opinion of management, all adjustments (consisting only of normal recurring adjustments) considered necessary for a fair statement have been included. Operating results for the three and nine months ended September 30, 2006 are not necessarily indicative of the results that may be expected for the year ending December 30, 2006, or for any other period. The balance sheet at December 31, 2005 has been derived from the audited consolidated financial statements at that date but does not include all of the information and footnotes required by accounting principles generally accepted in the United States of America for complete financial statements. These financial statements and notes should be read with the consolidated financial statements and notes thereto for the year ended December 31, 2005 included in the Company's Annual Report on Form 10-K filed with the SEC.

Reclassifications. Certain prior period balances have been reclassified to conform to the current financial statement presentation. These reclassifications had no impact on previously reported results of operations or stockholders' equity.

Fiscal Year. The Company operates on a 52- 53 week fiscal year, whereby the year ends on the Saturday nearest December 31. Fiscal year 2006 will end on December 30, 2006, and will consist of 52 weeks. The fiscal year ended December 31, 2005 consisted of 53 weeks.

Note 2 — Significant Accounting Policies

The Company's significant accounting policies are disclosed in the Company's Annual Report on Form 10-K for the year ended December 31, 2005. The Company's significant accounting policies reflect the adoption of the provisions of Statement of Financial Accounting Standards ("SFAS") No. 123 (revised 2004), "Share-Based Payment" ("SFAS 123 (R)"); and have otherwise not materially changed during the three and nine months ended September 30, 2006.

During the first quarter of fiscal 2006, the Company implemented SFAS 123 (R) with regard to equity based compensation. Beginning January 1, 2006, the Company began accounting for stock options and shares issued under its employee stock purchase plan ("ESPP") under SFAS 123 (R), which requires the recognition of the fair value of equity based compensation. The fair value of stock options and ESPP shares was estimated using a Black-Scholes option valuation model. This model requires the Company to make subjective assumptions in implementing SFAS 123 (R), including expected stock price volatility, estimated life and estimated forfeitures of each award. The fair value of equity-based awards is amortized over the vesting period of the award, and the Company has elected to use the straight-line method. The Company makes quarterly assessments of the adequacy of the additional paid-in capital pool ("APIC pool") to determine if there are any tax shortfalls which require recognition in the condensed consolidated income statements. Prior to the implementation of SFAS 123 (R), the Company accounted for stock options and ESPP shares under the provisions of Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees" and made pro forma footnote disclosures as required by SFAS No. 148, "Accounting for Stock-Based Compensation." Pro forma net income and pro forma net income per share disclosed in the footnotes to the condensed consolidated financial statements were estimated using a Black-Scholes option valuation model. Under APB Opinion No. 25, SFAS 123 and SFAS 123 (R), the fair value of restricted stock units was calculated based upon the fair market value of the Company's common stock at the date of grant.

The Company has elected to adopt the alternative transition method provided under the provisions of Financial Accounting Standards Board ("FASB") Staff Position No. FAS 123 (R)- 3 "Transition Election Related to Accounting for Tax Effects of Share-Based Payment Awards." The alternative transition method includes simplified methods to establish the beginning balance of the APIC pool related to the tax effects of employee stock-based compensation, and to determine the subsequent impact on the APIC pool and consolidated statements of cash flows of the tax effects of employee stock-based compensation awards that are outstanding upon adoption of SFAS 123 (R). See Note 6 — Stock-Based Compensation.

Note 3 — Public Offering of Common Stock

On March 15, 2006, the Company completed an offering of 5,000,000 shares of its common stock. The Company received net proceeds of \$182.0 million after the payment of an aggregate of \$8.1 million of underwriting discounts and commissions and other offering expenses.

Note 4 — Inventories

Inventories are stated at the lower of cost (principally standard cost which approximates actual cost on a first-in, first-out basis) or market value. The Company provides inventory provisions based on excess and obsolete inventories determined primarily by future demand forecasts. The provision is measured as the difference between the cost of the inventory and market based upon assumptions about future demand and charged to the provision for inventory, which is a component of cost of revenues. At the point of the loss recognition, a new, lower cost basis for that inventory is established, and subsequent changes in facts and circumstances do not result in the restoration or increase in that newly established cost basis.

Inventories consisted of the following:

	Sep	tember 30, 2006	Dec	ember 31, 2005
		(In thousands)		
Raw materials	\$	10,640	\$	7,686
Work-in-progress		12,563		9,971
Finished goods		3,114		747
	\$	26,317	\$	18,404

Note 5 — Warranty

The Company offers warranties on certain products and records a liability for the estimated future costs associated with customer claims, which is based upon historical experience and the Company's estimate of the level of future costs. Warranty costs are reflected in the income statement as a cost of revenues. A reconciliation of the changes in the Company's warranty liability (included in accrued liabilities) for the three and nine months ended September 30, 2006 and September 24, 2005, respectively, follows:

	Three Months Ended				Nine Months Ended			
	September 30, 2006				otember 30, 2006	Sej	otember 24, 2005	
	(In thousa				isands)			
Warranty accrual beginning balance	\$	1,012	\$	518	\$	511	\$	560
Accrual for warranties issued during the period		227		215		1,051		679
Settlements made during the period		(533)		(213)		(856)		(719)
Warranty accrual ending balance	\$	706	\$	520	\$	706	\$	520

Note 6 — Stock-Based Compensation

Effective January 1, 2006, the Company adopted the provisions of SFAS No. 123 (R), using the modified prospective transition method. SFAS 123 (R) requires companies to recognize the cost of employee services received in exchange for awards of equity instruments based upon the grant-date fair value of those awards. Using the modified prospective transition method, the Company began recognizing compensation expense for equity-based awards granted after December 31, 2005 plus unvested awards granted prior to December 31, 2005. Stock-based compensation expense for unvested awards granted prior to December 31, 2005 is amortized based on the measurement of fair value under SFAS No. 123, while awards granted after December 31, 2005 are measured under the guidance of SFAS No. 123 (R). Under this method of implementation no restatement of prior periods has been made. The cumulative effect related to the implementation of this new accounting principle as of January 1, 2006 was not material.

7

The application of SFAS No. 123 (R) had the following effects on reported amounts for the three and nine months ended September 30, 2006:

Septembe			
\$	5,195	\$	13,672
	880		1,820
	92		276
	(191)		(753)
	5,976		15,015
	(1,952)		(5,132)
\$	4,024	\$	9,883
\$	0.09	\$	0.22
\$	0.08	\$	0.21
	<u>Septembe</u> (In the second seco	\$ 5,195 880 92 (191) 5,976 (1,952) \$ 4,024 \$ 0.09	September 30, 2006 Septemb

- (1) Stock-based compensation expense of \$164,000 and \$895,000 for the three and nine months ended September 30, 2006, respectively, related primarily to pre-initial public offering "cheap stock" would have been recorded under the provisions of APB No. 25.
- (2) Stock-based compensation expense of \$92,000 and \$276,000 for the three and nine months ended September 30, 2006, respectively, related to restricted stock units would have been recorded under the provisions of APB No. 25.

Prior to January 1, 2006, the Company measured compensation expense for its employee equity-based compensation plans using the intrinsic value method under APB No. 25 and related interpretations. In connection with the grant of stock options to employees in fiscal 2001, fiscal 2002 and fiscal 2003 through the Company's initial public offering, the Company recorded stock-based compensation expense under the provisions of APB No. 25 as these options were considered compensatory because the fair value of the Company's stock determined for financial reporting purposes was greater than the fair value determined at the date of the grant. As of December 31, 2005, the Company had an aggregate of \$1.5 million of stock-based compensation remaining to be amortized related to these options under the intrinsic valuation method.

In addition, the Company recorded stock-based compensation expense related to the issuance of restricted stock. As of December 31, 2005, the Company had an aggregate of \$1.0 million of unamortized stock-based compensation related to restricted stock.

Prior to fiscal 2006, the Company applied the disclosure-only provisions of SFAS No. 123. The following table illustrates the effect on net income and earnings per share for the three and nine months ended September 24, 2005 if the fair value recognition provisions of SFAS No. 123 had been applied to options granted under the Company's equity-based employee compensation plans. For purposes of this pro forma disclosure, the estimated value of the options is recognized over the options' vesting periods.

8

	e Months Ended tember 24, 2005 (In thousands, except	Se	ine Months Ended eptember 24, 2005 are amounts)
Net income, as reported	\$ 9,778	\$	19,701
Add: Stock-based compensation expense included in reported net income, net of tax	1,029		2,085
Deduct: Total stock-based compensation expense determined under the minimum and fair-value-based			
method for all awards, net of tax	 (3,165)		(8,249)
Pro forma net income	\$ 7,642	\$	13,537
Net income per share			
Basic:			
As reported	\$ 0.25	\$	0.50
Pro-forma	\$ 0.19	\$	0.34
Diluted:			
As reported	\$ 0.23	\$	0.47
Pro-forma	\$ 0.18	\$	0.33

For purposes of the weighted-average estimated fair value calculations, the fair value of each stock option grant and employee purchase right is estimated on the date of grant using the Black-Scholes option pricing model and the following assumptions:

	Three Months Ended September 24, 2005	Nine Months Ended September 24, 2005
Stock Options:		
Dividend yield	—	—
Expected volatility	44.2%	47.8%
Risk-free interest rate	3.86%	3.82%
Expected life (in years)	4.50	4.50
ESPP:		
Dividend yield		_
Expected volatility	44.2%	47.7%
Risk-free interest rate	3.23%	3.23%
Expected life (in years)	1.25	1.25

Stock Options

The exercise price of each stock option equals the market price of the Company's stock on the date of grant. Most options are scheduled to vest over four years and expire in either seven or ten years from the grant date. The fair value of each option grant is estimated on the date of grant using the Black-Scholes option pricing model. In addition, the Company estimates forfeitures when recognizing compensation expense, and will adjust its estimate of forfeitures over the requisite service period based on the extent to which actual forfeitures differ, or are expected to differ, from such estimates. Changes in estimated forfeitures will be recognized as a change in estimate in the period of change and will also impact the amount of compensation expense to be recognized in future periods.

The following weighted average assumptions were used in the estimated grant-date fair value calculations for stock options:

	Three Months Ended September 30, 2006	Nine Months Ended September 30, 2006
Stock Options:		
Dividend yield	—	—
Expected volatility	52.9%	50.2%
Risk-free interest rate	4.93%	4.92%

Expected life (in years)	4.75	4.82

The Company's computation of expected volatility for the three- and nine-month periods ended September 30, 2006 was based on a combination of historical and market-based implied volatility from traded options on the Company's common stock. The Company believes that including market-based implied volatility in the calculation of expected volatility results in

a more accurate measure of the volatility expected in future periods. Prior to 2006, the computation of expected volatility was based entirely on historical volatility. The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of the grant for periods corresponding with the expected life of an option. When establishing the expected life of a newly granted option, the Company applies the simplified method approach as outlined in Staff Accounting Bulleting No. 107. The simplified method is based on the vesting period and the contractual term for each grant, or for each vesting-tranche for awards with graded vesting. The mid-point between the vesting date and the expiration date is used as the expected term under this method.

During the three and nine months ended September 30, 2006, the Company granted approximately 178,100 and 2,067,227 stock options with an estimated total grant-date fair value of \$3.9 million and \$38.5 million, respectively. During the three and nine months ended September 30, 2006, the Company recorded stock-based compensation expense related to stock options of \$5.2 million and \$13.7 million, respectively. As of September 30, 2006, the unamortized stock-based compensation balance related to stock options was \$43.6 million after estimated forfeitures which will be recognized over an estimated period of 3.0 years based on the weighted-average unamortized stock-based compensation balance. Approximately \$0.8 million of stock-based compensation was capitalized in inventory for the nine months ended September 30, 2006.

Equity Incentive Plans

The Company has four incentive plans: 1996 Stock Option Plan, Incentive Option Plan, Management Incentive Option Plan (collectively, the "Plans") and 2002 Equity Incentive Plan ("2002 Plan"), which became effective in June 2002. As a result of the effectiveness of the 2002 Plan, the Company ceased granting any options under the Plans.

Activity under the Plans and the 2002 Plan is set forth below:

	Shares Awards Available Outstanding			hted- Average ercise Price
	(In thousan	ds, except share and per sha	are data)	
Balances, December 25, 2004	3,201,452	5,822,746	\$	11.88
Additional shares reserved	1,944,281			_
Options granted	(2,476,543)	2,476,543		24.89
Restricted stock awards granted	(17,000)	_		_
Options exercised	—	(1,042,373)		8.36
Options expired		(15,000)		0.10
Options forfeited	653,939	(653,989)		15.99
Balances, December 31, 2005	3,306,129	6,587,927	\$	16.91
Additional shares reserved	2,011,884	_		
Options granted	(2,067,227)	2,067,227		38.61
Restricted stock awards granted	_	_		
Options exercised	—	(1,296,669)		23.90
Options forfeited	260,263	(260,263)		11.63
Balances, September 30, 2006	3,511,049	7,098,222	\$	23.94

The options outstanding and exercisable at September 30, 2006 were in the following exercise price ranges:

		Options Ou	itstanding			Optio	ns Exercisa	ıble	
Range of Exercise Prices	Number of Options Outstanding	Weighted Average Remaining Contractual <u>Term</u> (in years)	Weighted Average Exercise Price	Aggregat Intrinsio Value (in thousan	Vested and Exercisable	A E	eighted verage xercise Price		Aggregate Intrinsic Value 1 thousands)
\$0.50 - \$6.00	352,649	3.54	\$ 4.72	\$ 13	,194 352,649	\$	4.72	\$	13,194
\$6.01 - \$6.50	960,544	5.43	6.50	34	,224 738,896		6.50		26,327
\$6.51 - \$19.50	1,202,150	7.03	17.43	29	,698 538,431		15.86		14,145
\$19.51 - \$23.56	935,818	8.25	22.49	18	,382 312,013		22.39		6,159
\$23.57 - \$25.08	84,656	8.69	24.51	1	,491 16,132		24.54		284
\$25.09 - \$25.39	926,039	9.10	25.39	15	,502 151,418		25.39		2,535
\$25.40 - \$30.46	741,509	8.69	26.60	11	,513 189,839		26.17		3,030
\$30.47 - \$38.38	761,817	6.62	38.19	3	,004 10,000		36.31		58
\$38.39 - \$39.84	876,640	6.61	39.58	2	,233 —				
\$39.85 - \$47.63	256,400	6.75	42.59						
\$0.50 - \$47.63	7,098,222	7.16	\$ 23.94	<u>\$ 129</u>	,241 2,309,378	\$	13.67	\$	65,732

The aggregate intrinsic value in the table above represents the total pre-tax intrinsic value based on the Company's closing stock price of \$42.13 on September 29, 2006, which would have been received by the option holders had all option holders exercised their options as of that date. The total number of in-the-money options vested and exercisable as of September 30, 2006 was 2,309,378. The total intrinsic value of options exercised during the nine months ended September 30, 2006 was \$38.5 million.

The weighted average grant-date fair value of options granted during the three and nine months ended September 30, 2006 was \$21.66 and \$18.62. Cash received from stock option exercises was \$8.0 million and \$15.1 million during the three and nine months ended September 30, 2006, respectively. In connection with these exercises, the gross tax benefits realized by the Company for the three and nine months ended September 30, 2006 was \$8.1 million and \$15.9 million, respectively.

The Company settles employee stock option exercises with newly issued common shares.

Employee Stock Purchase Plan

The ESPP provides that eligible employees may contribute up to 15% of their eligible earnings toward the semi-annual purchase of the Company's common stock. Under the ESPP, employees may purchase the Company's common stock through payroll deductions at a price equal to 85% of the lower of the fair market value at the beginning of the applicable offering period or at the end of each applicable purchase period. Offering periods are generally two years in length. During the nine months ended September 30, 2006, 209,789 shares were issued under the ESPP. As of September 30, 2006, the Company had \$2.4 million of total unrecognized deferred stock-based compensation, net of estimated forfeitures related to ESPP grants, which will be recognized over the weighted average period of 0.8 years. Compensation expense is calculated using the fair value of the employees' purchase rights under the Black-Scholes model. The following weighted average assumptions were used in the estimated fair value calculations for the employees' purchase rights:

	Three Months Ended September 30, 2006	Nine Months Ended September 30, 2006
ESPP:		
Dividend yield	—	
Expected volatility	49.1%	48.0%
Risk-free interest rate	4.34%	4.19%
Expected life (in years)	1.43	1.39

Restricted Stock Units

Restricted stock units are converted into shares of the Company's common stock upon vesting on a one-for-one basis. The vesting of restricted stock units is subject to the employee's continuing service to the Company. The cost of these awards

1	1
т	т

is determined using the fair value of the Company's common stock on the date of the grant, and compensation cost is recognized over the vesting period. Restricted stock units generally vest over four years.

Activity of the restricted stock units under the Company's equity compensation plans during the nine months ended September 30, 2006 is set forth below:

	Shares	Weighted Average Grant Date Fair Value
Restricted stock units at December 31, 2005	55,432	\$ 25.27
Granted	_	_
Vested	(18,108)	25.27
Forfeited	_	_
Restricted stock units at September 30, 2006	37,324	\$ 25.27

The total aggregate intrinsic value of restricted stock units outstanding as of September 30, 2006 is \$1.6 million. Aggregate intrinsic value is calculated using the closing price of the Company's common stock on September 30, 2006 multiplied by the number of restricted stock units outstanding at September 30, 2006.

As of September 30, 2006, the Company had \$0.7 million of unrecognized stock-based compensation costs related to restricted stock unit grants, which will be recognized over the weighted average remaining contractual term of 1.3 years. As of September 30, 2006, the Company expects 37,324 restricted stock units to vest.

Note 7 — Net Income per Share

Basic net income per share is computed by dividing net income by the weighted-average number of common shares outstanding for the period. Diluted net income per share is computed giving effect to all potential dilutive common stock, including stock options, warrants, restricted stock units and common stock subject to repurchase.

A reconciliation of the numerator and denominator used in the calculation of basic and diluted net income per share follows:

		Three Months Ended				Nine Months Ended			
	_		mber 30, 2006	Se	ptember 24, 2005	Sep	otember 30, 2006	Se	eptember 24, 2005
Numerator:	-								
Net income	9	\$	15,819	\$	9,778	\$	41,868	\$	19,701
Denominator:	=					-			

Weighted-average common stock outstanding	46,417	39,733	44,625	39,356
Less:				
Weighted-average shares subject to repurchase	—	—	—	(13)
Weighted-average shares used in computing basic net income				
per share	46,417	39,733	44,625	39,343
Dilutive potential common shares used in computing diluted				
net income per share	2,077	2,029	2,065	2,149
Total weighted-average number of shares used in computing				
diluted net income per share	48,494	41,762	46,690	41,492

The following outstanding options to purchase common stock and restricted stock units were excluded from the computation of diluted net income per share as they had an antidilutive effect (in thousands):

	Three Montl	hs Ended	Nine Month	ıs Ended
	September 30, 2006	September 24, 2005	September 30, 2006	September 24, 2005
Options to purchase common stock	1,932	305	1,986	553
Restricted stock units		_	_	13
	12			

Note 8 — Commitments and Contingencies

Environmental Matters

The Company is subject to U.S. federal, state and local, and foreign governmental laws and regulations relating to the protection of the environment, including those governing the discharge of pollutants into the air and water, the management and disposal of hazardous substances and wastes, the clean-up of contaminated sites and the maintenance of a safe workplace. The Company believes it complies with all material environmental laws and regulations that apply to the Company.

While the Company believes it is in compliance with all material environmental laws and regulations that apply to the Company, the Company has received, and in the future the Company may receive, environmental violation notices, and if received, final resolution of the violations identified by these notices could impact the Company's operations. New laws and regulations could have application to, and impact the Company's operations. The Company cannot predict what future environmental laws, rules or regulations will be enacted or how existing or future laws, rules or regulations will be administered or interpreted. The Company also cannot predict the amount of future expenditure that may be required to comply with such laws, rules or regulations. The discovery of previously unknown contamination at the Company's or others' sites or the imposition of new cleanup requirements could result in violations of certain applicable laws or regulations and adversely impact the Company's business, financial condition or result of operations.

Legal Matters

From time to time, the Company may be subject to legal proceedings and claims in the ordinary course of business. As of the filing date of this Quarterly Report on Form 10-Q, the Company was not involved in any material legal proceedings, other than patent litigation as set forth below. In the future we may become parties to additional legal proceedings which require us to spend significant resources including proceedings designed to protect our intellectual property rights.

On February 24, 2004, the Company filed in the Seoul Southern District Court, located in Seoul, South Korea, two separate complaints against Phicom Corporation, a Korean corporation, alleging infringement of a total of four Korean patents issued to the Company. One complaint alleges that Phicom is infringing the Company's Korean Patent Nos. 252,457, entitled "Method of Fabricating Interconnections Using Cantilever Elements and Sacrificial Substrates," and 324,064, entitled "Contact Tip Structures for Microelectronic Interconnection Elements and Methods of Making Same". The other complaint alleges Phicom is infringing the Company's Korean Patent Nos. 278,342, entitled "Method of Altering the Orientation of Probe Elements in a Probe Card Assembly," and 399,210, entitled "Probe Card Assembly". Both complaints seek injunctive relief. The court actions are part of the Company's ongoing efforts to protect the intellectual property embodied in its proprietary technology, including its MicroSpring interconnect technology.

On or about March 19, 2004, Phicom filed in the Korean Intellectual Property Office, or KIPO, invalidity actions challenging the validity of some or all of the claims of each of the four Company patents at issue in the Seoul infringement actions. KIPO dismissed Phicom's challenges against all four of the patentsat-issue. Phicom appealed the dismissals of the challenges to the Korean Patent Court.

On or about October 27, 2005, the Korean Patent Court issued rulings holding invalid certain claims of two of the Company's Korean patents. The two Korean patents affected by the decisions are Nos. 278,342, entitled "Method of Altering the Orientation of Probe Elements in a Probe Card Assembly," and 399,210, entitled "Probe Card Assembly". The Company is appealing these decisions to the Korean Supreme Court. The Company is also continuing its enforcement action against Phicom under these patents in the Seoul Southern District Court, including alleging infringement of certain claims from the patents that were not addressed by the Korean Patent Court decisions.

On or about February 9, 2006, the Korean Patent Court invalidated ten claims of the Company's Korean Patent No. 324,064, entitled "Contact Tip Structures for Microelectronic Interconnection Elements and Methods of Making Same," but did not address some sixty-one other claims of the 324,064 patent that were not before the Patent Court. The Company is appealing the decision on its Korean Patent No. 324,064 to the Korean Supreme Court. On or about June 15, 2006, the Korean Patent Court issued a decision upholding the validity of all of the claims of the Company's Korean Patent No. 252,457, entitled "Method of Fabricating Interconnections Using Cantilever Elements and Sacrificial Substrates." Phicom has appealed this ruling on the 252,457 patent to the Korean Supreme Court. The Company is also continuing its enforcement action against Phicom under both the 252,457 and 324,064 Korean patents in the Seoul Southern District Court, including certain claims from the 324,064 patent that were not addressed by the Korean Patent Court decision.

On or about August 7, 2006 the Company filed in the Seoul Central District Court, located in Seoul, South Korea, two actions against Phicom alleging infringement of certain claims of the Company's Korean Patent No. 252,457, entitled "Method of Fabricating Interconnections Using Cantilever Elements and Sacrificial Substrates." The actions include an

"injuction" action, which seeks preliminary injuctive relief, and a "merits" action. For each and all of the actions pending in the Seoul Southern District Court and the Seoul Central District Court, Phicom is asserting defenses to the Company's claims.

On March 4, 2005, the Company filed a patent infringement lawsuit in the United States District Court for the District of Oregon against Phicom charging that it is willfully infringing four U.S. patents that cover key aspects of the Company's wafer probe cards. The complaint in this action alleges that Phicom has incorporated the Company's proprietary technology into its products and seeks both injunctive relief and monetary damages. The U.S. patents identified in the complaint are U.S. Patent No. 5,974,662, entitled "Method of Planarizing Tips of Probe Elements of a Probe Card Assembly", U.S. Patent No. 6,246,247, entitled "Probe Card Assembly and Kit, and Methods of Using Same", U.S. Patent No. 6,624,648, entitled "Probe Card Assembly" and U.S. Patent No. 5,994,152, entitled "Fabricating Interconnects and Tips Using Sacrificial Substrates". Three of the patents at issue in the U.S. are substantially similar to those at issue in the Company's litigation with Phicom in Korea. On or about August 2, 2005, Phicom answered the complaint by denying infringement, alleging defenses and asserting counterclaims seeking adjudications on the validity and enforceability of the Company's patents and whether Phicom is infringing those patents. Phicom's motion with the Oregon District Court seeking that the lawsuit be transferred to the U.S. District Court for the Northern District of California was denied without prejudice by the District Court. On or about February 7, 2006, the District Court issued a scheduling order as jointly proposed by the parties that culminates in a pretrial conference in or about August 2007, followed by a two to four week trial at a date to be set by the Court. As of the date of this Quarterly Report on Form 10-Q, discovery has begun and the parties have exchanged written preliminary contentions regarding infringement and validity.

The Company has incurred and could in the future incur material legal expenses in connection with these legal proceedings.

One or more third parties have initiated challenges in foreign patent offices against other of the Company's patents. For example, on or about October 6, 2004, Micronics Japan Co., Ltd. ("MJC") filed an invalidation proceeding with KIPO relating to the Company's Korean Patent No. 312,872. After briefing, KIPO dismissed the challenge and upheld the validity of all of the claims of the Company's Korean Patent No. 312,872. The matter was appealed by MJC to the Korean Patent Court, which rendered a decision finding invalid all of the claims of Patent No. 312,872, The Company has appealed this decision to the Korean Supreme Court. MJC also filed a new proceeding in KIPO challenging the validity of the 312,872 patent. On or about April 22, 2005, an action was filed by MJC with KIPO against the Company's Korean Patent No. 467,997. On or about November 1, 2006, KIPO dismissed MJC's challenge and upheld all of the claims of the Company's Korean Patent No. 467,997. By way of further example, challenges against four of the Company's Taiwan patents, Taiwan Patent Nos. 83,716 (two challenges), 189,155, 198,158 and 121,535, have been filed in the Taiwan Intellectual Property Office, or TIPO. TIPO has not issued rulings in any of the validity challenge proceedings. While the Company believes that it does not have a material monetary damages exposure in these various invalidity proceedings, it is possible the Company will incur material attorneys' fees in defending its intellectual property at issue in these challenges.

Indemnification Obligations

The Company from time to time in the ordinary course of its business enters into contractual arrangements with third parties that include indemnification obligations. Under these contractual arrangements, the Company has agreed to defend, indemnify and hold the third party harmless from and against certain losses. These arrangements may limit the time within which an indemnification claim can be made, the type of the claim and the total amount that the Company can be required to pay in connection with the indemnification obligation. In addition, the Company has entered into indemnification agreements with its directors and certain of its officers, and the Company's bylaws contain indemnification obligations in favor of the Company's directors, officers and agents. It is not possible to determine or reasonably estimate the maximum potential amount of future payments under these indemnification obligations due to the varying terms of such obligations, the history of prior indemnification. The Company has not had any requests for indemnification under these arrangements. The Company has not recorded any liabilities for these indemnification arrangements on the Company's condensed consolidated balance sheet as of September 30, 2006.

Lease Obligations

In August 2006 the Company signed an amendment to an existing lease for the remaining 37,439 square feet of the building. The term of the lease was extended to 15-years. The total rent obligation over the amended term of the lease is \$8.5 million and is accounted for as an operating lease. The Company also signed a five-year lease for an additional 39,478 square feet of office space in September 2006. The total rent obligation over the term of the lease is \$2.4 million and is accounted for as an operating lease.

Note 9 — Asset Retirement Obligation

The Company accounts for the retirement of tangible long-lived assets and the associated asset retirement costs in accordance with SFAS No. 143, "Accounting for Asset Retirement Obligations". SFAS No. 143 requires that the fair value of a liability for an asset retirement obligation be recognized in the period in which it is incurred if a reasonable estimate of fair value can be made. In accordance with SFAS No. 143, the fair value of the liability is added to the carrying amount of the associated asset and this additional carrying amount is amortized over the life of the asset. The Company's asset retirement obligation is associated with its commitment to return property subject to operating leases in Jubei City Hsinchu, Taiwan, and Tokyo and Yokohama, Japan to original condition upon lease termination. The Company estimated that as of September 30, 2006, gross expected future cash flows of approximately \$0.7 million would be required to fulfill these obligations. The Company has recorded the respective asset retirement obligations and a corresponding increase in leasehold improvements of \$0.7 million. This amount represents the present value of expected future cash flows associated with returning the leased property to original condition. The leasehold improvements are being amortized to depreciation and amortization expense over the term of the lease.

Note 10 — Stockholders' Equity

Comprehensive Income (Loss)

Comprehensive income (loss) includes foreign currency translation adjustments and unrealized gains (losses) on available-for-sale securities, the impact of which has been excluded from net income and reflected as components of stockholders' equity.

Components of comprehensive income were as follows:

	Three Months Ended				Nine Months Ended				
	September 30, 2006							September 24, 2005	
				(In tho	usands				
Net income	\$	15,819	\$	9,778	\$	41,868	\$	19,701	
Change in unrealized gain (loss) on marketable securities		722		207		199		177	
Foreign currency translation adjustments		(29)		(20)		46		333	
Comprehensive income	\$	16,512	\$	9,965	\$	42,113	\$	20,211	

Components of accumulated other comprehensive loss were as follows:

	tember , 2006	December 31, 2005	
	 (In thous	ands)	
Unrealized loss on marketable securities	\$ (136)	\$ (3	335)
Foreign currency translation adjustments	22	((24)
Accumulated other comprehensive loss	\$ (114)	\$ (3	359)

Note 11 — Derivative Financial Instruments

The Company purchases forward exchange contracts to hedge certain existing foreign currency denominated accounts receivable. These hedges do not qualify for hedge accounting treatment in accordance with the provisions of SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities." The Company recognizes gains or losses from the fluctuation in foreign exchange rates and the valuation of these hedge contracts in other expense. The Company does not use derivative financial instruments for trading or speculative purposes. As of September 30, 2006, the Company had one outstanding foreign exchange forward contract to sell 2,600,000,000 Japanese Yen for \$22,090,059 with a contract rate of 117.70 Japanese Yen per U.S. Dollar.

Note 12 — Recent Accounting Pronouncements

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements" ("SFAS No. 157"). SFAS No. 157 defines fair value, establishes a framework for measuring fair value in accordance with generally accepted accounting principles, and expands disclosures about fair value measurements. This statement does not require any new fair value measurements; rather, it applies under other accounting pronouncements that require or permit fair value measurements. The provisions of this statement are to be applied prospectively as of the beginning of the fiscal year in which this statement is initially applied, with any transition adjustment recognized as a cumulative-effect adjustment to the opening balance of retained earnings. The provisions of SFAS No. 157 are effective for the fiscal years beginning after November 15, 2007; therefore, the Company anticipates adopting this standard as of January 1, 2008. The Company has not determined the effect, if any, the adoption of this statement will have on its consolidated financial statements.

In September 2006, the SEC issued Staff Accounting Bulletin No. 108, "Considering the Effects of Prior Year Misstatements when Quantifying Misstatements in Current Year Financial Statements" ("SAB No. 108"), to eliminate the diversity of practice surrounding how public companies quantify financial statement misstatements. Traditionally, there have been two widely-recognized methods for quantifying the effects of financial statement misstatements: the "roll-over" method and the "iron curtain" method. The roll-over method focuses primarily on the impact of a misstatement on the income statement, including the reversing effect of prior year misstatements, but its use can lead to the accumulation of

1	5
т	J

misstatements in the balance sheet. The iron-curtain method, on the other hand, focuses primarily on the effect of correcting the period-end balance sheet with less emphasis on the reversing effects of prior year errors on the income statement. In SAB No. 108, the SEC Staff established an approach that requires quantification of financial statement misstatements based on the effects of the misstatements on each financial statement and the related financial statement disclosures. This model is commonly referred to as a "dual approach" because it requires quantification of errors under both the iron curtain and the roll-over methods. The Company is required to adopt SAB No. 108 in the fourth quarter of 2006 and is currently evaluating the impact of this interpretation on its consolidated financial statements.

In July 2006, the FASB issued FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes, an Interpretation of FASB Statement No. 109" ("FIN 48"). FIN 48 provides guidance on the financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. FIN 48 requires that we recognize in the financial statements the impact of a tax position if that position will more likely than not be sustained on audit, based on the technical merits of the position. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosures, and transition provisions. FIN 48 is effective for fiscal years beginning after December 15, 2006, and the Company will adopt FIN 48 as of January 1, 2007. The Company is currently evaluating the impact of this interpretation on its consolidated financial statements.

In July 2006, the FASB issued EITF Issue No. 06-3, "How Taxes Collected from Customers and Remitted to Governmental Authorities Should be Presented in the Income Statement (that is, Gross versus Net Presentation)." The adoption of EITF No. 06-3 did not have an impact on the Company's consolidated financial statements. The Company's accounting policy has been to present above mentioned taxes on a net basis, excluded from revenues.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Cautionary Statement Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of the Securities Exchange Act of 1934 and the Securities Act of 1933, which are subject to risks, uncertainties and assumptions that are difficult to predict. The forward-looking statements include statements concerning, among other things, our business strategy, including anticipated trends and developments in and management plans for our business and the markets in which we operate, financial results, operating results, revenues, gross margin, operating expenses, products, projected costs and capital expenditures, research and development programs, sales and marketing initiatives, and competition. In some cases, you can identify these statements by forward-looking words such as "may," "might," "could," "should," "expect," "plan," "anticipate," "believe," "estimate," "predict," "intend" and "continue," the negative or plural of these words and other comparable terminology.

The forward-looking statements are only predictions based on our current expectations and our projections about future events. All forward-looking statements included in this quarterly report are based upon information available to us as of the filing date of this quarterly report. You should not place undue reliance on these forward-looking statements. We undertake no obligation to update any of these statements for any reason. These forward-looking statements involve known and unknown risks, uncertainties and other factors that may cause our actual results, levels of activity, performance or achievements to differ materially from those expressed or implied by these statements. These factors include the matters discussed in the section titled "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2005 and elsewhere in this quarterly report. You should carefully consider the numerous risks and uncertainties described under such sections.

The following discussion and analysis should be read in conjunction with our condensed consolidated financial statements and the accompanying notes contained in this quarterly report. Unless expressly stated or the context otherwise requires, the terms "we," "our," "us" and "FormFactor" refer to FormFactor, Inc. and its subsidiaries.

Overview

We design, develop, manufacture, sell and support precision, high performance advanced semiconductor wafer probe cards. Semiconductor manufacturers use our wafer probe cards to perform wafer probe test on the whole semiconductor wafer, prior to singulation, in the front end of the semiconductor manufacturing process. After the fabrication of a semiconductor wafer, the chips on the wafer are again subject to wafer probe test. During wafer probe test, a wafer probe card is mounted in a prober, which is in turn connected to a semiconductor tester, and the wafer probe card is used as an interface to connect electronically with and test individual chips on a wafer. At the core of our product offering are our proprietary technologies, including our MicroSpring interconnect technology includes a resilient contact element manufactured at our production facilities in Livermore, California. We operate in a single industry segment and have derived our revenues primarily from the sale of wafer probe cards incorporating our MicroSpring interconnect technology.

We work closely with our customers to design, develop and manufacture custom wafer probe cards. Each wafer probe card is a custom product that is specific to the chip and wafer designs of the customer. Our customers, in turn, operate in the highly cyclical semiconductor industry and are subject to fluctuations in the demand for their products. Because of the nature of our customers, and our business, our revenue growth is driven in significant part by the number of new semiconductor designs that our customers develop, the technology transitions involved in these designs and our customers' production volumes. In the past, this has resulted in our being subject to demand fluctuations that have resulted in significant variations of revenues, expenses and results of operations in the periods presented. We expect these fluctuations, and the resulting variations in our financial results, to continue in future periods.

Revenues. We derive substantially all of our revenues from product sales of wafer probe cards. Wafer probe card sales, including service and non-recurring engineering revenue associated with wafer probe card sales, accounted for virtually all of our revenues in the first nine months of fiscal 2006 and 2005. Revenues from licensing of our design and manufacturing technologies have historically been insignificant. Increases in revenues have resulted from increased demand for our existing products, the introduction of new, more complex products and the penetration of new markets. Revenues from our customers are subject to both quarterly, annual and other fluctuations due to design cycles, technology adoption rates and cyclicality of the different end markets into which our customers' products are sold.

Cost of Revenues. Cost of revenues consists primarily of manufacturing materials, payroll and other manufacturing-related overhead. Our manufacturing operations rely upon a limited number of suppliers to provide key components and materials for our products, some of which are a sole source. We order materials and supplies based on backlog and forecasted customer orders. Tooling and setup costs related to changing manufacturing lots at our suppliers are also included in the cost of revenues. We expense all warranty costs and inventory write downs or write-offs as cost of revenues.

We design, manufacture and sell a fully custom product for wafer probe test to semiconductor manufacturers, which operate in an industry subject to significant cyclicality and demand fluctuations. Our wafer probe cards are complex products that are custom to a specific chip design and must be delivered on relatively short lead-times as compared to our overall manufacturing process. As our advanced wafer probe cards are manufactured in low volumes and must be delivered on relatively short lead-times, it is not uncommon for us to acquire production materials and start certain production activities based on estimated production yields and forecasted demand prior to or in excess of actual demand for our wafer probe cards. We record inventory write downs for estimated obsolete and non-saleable inventories equal to the difference between the cost of inventories and the estimated market value based upon assumptions about future demand and market conditions. If actual market conditions are less favorable than those projected by management, additional inventory write downs would be required. Once established, the original cost of our inventory less the related inventory write downs represents the new cost basis of such products. Reversal of these reserves is recognized only when the related inventory has been scrapped or sold.

Research and Development. Research and development expenses include expenses related to product development, engineering and material costs. Almost all research and development costs are expensed as incurred. We plan to continue to invest a significant amount in research and development activities to develop new technologies for current and new markets and new applications in the future.

Selling, General and Administrative. Selling, general and administrative expenses include expenses related to sales, marketing, and administrative personnel, internal and outside sales representatives' commissions, market research and consulting, and other sales, marketing, and administrative activities. These expenses also include costs for enforcing our patent rights and regulatory compliance costs. We expect that selling expenses will increase as revenues

increase, and we expect that general and administrative expenses will increase to support future revenue growth and build out of global infrastructure to support international customers.

Use of Estimates. Our discussion and analysis of our financial condition and results of operations are based upon our unaudited condensed consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States of America. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amount of assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities. On an on-going basis, we evaluate our estimates, including those related to uncollectible receivables, inventories, marketable securities, intangible assets, income taxes, warranty obligations, excess component and order cancellation costs, contingencies and litigation, and stock-based compensation. We base our estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources.

Stock-based Compensation. Effective January 1, 2006, we adopted the provisions of SFAS No. 123 (R), which require us to measure all employee stockbased compensation awards using a fair value method and record such expense in our consolidated financial statements. In connection with our implementation of SFAS No. 123 (R), we have included in this Management's Discussion and Analysis of Financial Condition and Results of Operations certain non-GAAP measures, which are not in accordance with, or an alternative for, generally accepted accounting principles and may be different from non-GAAP measures used by other companies. We believe that our financial presentation in this section, which includes non-GAAP measures and the corresponding GAAP measures, gives our stockholders and other interested persons the benefit of a consistent basis for assessing our financial results for the three and nine months ended September 30, 2006 relative to our historical GAAP financials. Prior historical periods included only stock-based compensation charges related to pre-initial public offering "cheap stock" and restricted stock units accounted for under the provisions of APB No. 25. Our management also uses certain non-GAAP measures when assessing business trends and performance, forecasting and planning future operations, and evaluating our financial performance when compared to prior periods. These non-GAAP measures should be considered along with our GAAP measures.

Results of Operations

The following table sets forth our operating results as a percentage of revenues for the periods indicated:

	Three Months	Ended	Nine Months I	Ended
	September 30, 2006	September 24, 2005	September 30, 2006	September 24, 2005
Revenues	100.0%	100.0%	100.0%	100.0%
Cost of revenues	48.1	54.7	48.3	56.4
Gross margin	51.9	45.3	51.7	43.6
Operating expenses:				
Research and development	12.4	12.6	12.4	11.7
Selling, general and administrative	19.9	19.0	19.6	19.0
Total operating expenses	32.3	31.6	32.0	30.7
Operating income	19.6	13.7	19.7	12.9
Interest income	4.6	1.8	3.8	1.8
Other income (expense), net	0.1	(1.0)	0.0	(0.4)
Income before income taxes	24.3	14.5	23.5	14.3
Benefit from (provision for) income taxes	(8.0)	1.2	(8.0)	(2.4)
Net income	16.3%	15.7%	15.5%	11.9%

Three Months Ended September 30, 2006 and September 24, 2005

Revenues

	Three Months Ended						
	Se	ptember 30, 2006	Sep	otember 24, 2005		Increase lecrease)	Change %
				(In tho	usand	s)	
Revenues by Market:							
DRAM	\$	70,386	\$	47,309	\$	23,077	48.8%
Flash		16,675		8,715		7,960	91.3
Logic		9,696		6,350		3,346	52.7
Total Revenues	\$	96,757	\$	62,374	\$	34,383	55.1%

Revenues increased 55.1% in the three months ended September 30, 2006 compared with the three months ended September 24, 2005. Strong market demand for our advanced wafer probe cards continued in the third quarter of fiscal 2006 due to a variety of factors, including the increased demand for mobile and consumer applications, the transition to advanced technology nodes such as 90 nanometer and below, as well as the ongoing build-out of 300mm factories.

The majority of our revenues for the three months ended September 30, 2006 were generated by sales of wafer probe cards to manufacturers of DRAM devices. The increase was primarily due to the ongoing transition to advanced technology nodes, such as 90 nanometer and below, the ongoing build-out of 300mm factories and expanded relationships with new customers such as Hynix. Approximately 86% of our DRAM revenues in the third quarter of fiscal 2006 were derived from 90 nanometer and below technology products.

Revenues generated from sales to flash memory device manufacturers increased mainly due to increased demand for our NAND and NOR flash wafer probe cards. Demand for both our NAND and NOR flash wafer probe cards resulted from

growing demand for consumer applications which utilize multi-chip packages. Semiconductors that are integrated into multi-chip packages often benefit from increased wafer level testing to validate device performance before packaging.

Revenues from manufacturers of logic devices increased primarily due to increased demand by the computer data processing market and new customer engagements. The increased demand is driven by factors such as higher test parallelism migrations to flip-chip packaging and the need for higher device reliability. The majority of our logic revenues in the three months ended September 30, 2006 was derived from sales of wafer probe cards to test high performance flip-chip microprocessor and chipset applications used in personal computer, gaming and graphics applications.

Revenue by Geographic Region

The following table sets forth our revenues by geographic region for the periods indicated.

		Three Months Ended									
	September 30, 2006		% of Revenues	September 24, 2005	% of Revenues						
	(In thousands)										
Japan	\$	33,311	34.4%	\$ 19,079	30.6%						
North America		30,136	31.1	25,378	40.7						
Asia Pacific		29,479	30.5	12,336	19.8						
Europe		3,831	4.0	5,581	8.9						
Total Revenues	\$	96,757	100.0%	\$ 62,374	100.0%						

Geographic revenue information is based on the invoicing location of the customer. For example, certain Korean customers purchase through their North American subsidiaries. The increase in revenues in North America was primarily driven by demand for wafer probe cards used to test chips for consumer products. The increase in revenues in Japan was primarily due to increased sales to a manufacturer of DRAM devices. The increase in revenues in Asia Pacific was primarily due to growth in our business with both Taiwanese and Korean customers. The increase in revenues in Europe was primarily due to increased sales to a manufacturer of DRAM devices in this region.

The following customers accounted for more than 10% of our revenues for the three months ended September 30, 2006 and September 24, 2005:

	Three Months	Ended
	September 30, 2006	September 24, 2005
Elpida	25.0	29.0
Intel Corporation	13.8	12.6
Hynix	10.0	*
Samsung	*	19.9
Spirox Corporation	*	18.6%

* Less than 10% of revenues.

Revenues from Spirox Corporation decreased in the three months ended September 30, 2006 compared to the three months ended September 24, 2005 primarily due to our transition to a direct sales model in Taiwan. Until October 17, 2005, we relied upon Spirox Corporation to sell our products in Taiwan. While Spirox continues to serve as our distributor in Singapore, Philippines, Malaysia and China, a significant percentage of their sales occurred in Taiwan.

Gross Margin

		Three Months Ended						
	Se	eptember 30, 2006	% of Revenues	September 24, 2005		% of Revenues		
		(in thousands)						
Gross margin	\$	50,265	51.9.%	\$	28,286	45.3%		
Stock-based compensation expense		1,197	1.3		108	0.2		
Gross margin excluding stock-based compensation	\$	51,462	53.2%	\$	28,394	45.5 [%]		

The increase in gross margin for the three months ended September 30, 2006 compared with the three months ended September 24, 2005 was primarily due to factory productivity and both yield and through-put delivery improvements. These factory improvements and the increased revenue resulted in labor and material efficiencies that were the main drivers for the increased gross margin. Inventory write downs totaled \$3.2 million, or 3.4% of revenues for the three months ended September 30, 2006 as compared with \$3.2 million, or 5.1% of revenues for the three months ended September 24, 2005 primarily due to excess custom inventory quantities. As our advanced wafer probe cards are manufactured in low volumes and must be delivered on relatively short lead-times, it is not uncommon for us to acquire production materials and start certain production activities based on estimated production yields and forecasted demand prior to or in excess of actual demand for our wafer probe cards. Gross margin for the three months ended September 24, 2005 was impacted by non-recurring factory start up costs of \$2.3 million. Stock-based compensation expense increased to \$1.2 million for the three months ended September 30, 2006 due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006.

Research and Development

	Three Months Ended						
	September 30, 2006		% of Revenues	September 24, 2005	% of Revenues		
			(in thousand	ds)			
Research and development	\$	11,994	12.4% \$	7,881	12.6%		
Stock-based compensation expense		1,433	1.5	181	0.3		
Research and development excluding stock-based compensation	\$	10,561	10.9% \$	7,700	12.3%		

The increase in research and development expenses in absolute dollars was mainly due to an increase of approximately \$2.3 million in personnel costs and an increase of \$1.3 million in stock-based compensation expense due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006. Research and development expenses increased mainly due to our investment in the development of our Harmony architecture, fine pitch memory and logic products, advanced MicroSpring interconnect technology and new process technologies. We are also making incremental investments in new technologies and products as we focus on new market opportunities.

Selling, General and Administrative

	Three Months Ended							
	Sep	tember 30, 2006	% of September 24, Revenues 2005			% of Revenues		
Selling, general and administrative	\$	19,321	19.9.%	\$	11,871	19.0%		
Stock-based compensation expense		3,346	3.4		1,216	1.9		
Selling, general and administrative excluding stock-based compensation	\$	15,975	16.5%	\$	10,655	17.1%		

The increase in selling, general and administrative expenses in absolute dollars was mainly due to an increase of approximately \$4.1 million in personnel related expenses and an increase of \$2.1 million in stock-based compensation expense due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006.

Interest and Other Income (Expense), Net

		Three Months Ended						
	Sep	tember 30, 2006	% of Revenues	September 24, 2005	% of Revenues			
			(in tho	ısands)				
Interest income	\$	4,485	4.6%	5 \$ 1,116	1.8%			
Other income (expense), net	\$	59	0.1%	6 \$ (630)	(1.0)%			

The increase in interest income was due to larger cash, cash equivalents and marketable securities balances throughout the third quarter of fiscal 2006 compared to the third quarter of fiscal 2005 and higher interest rates, resulting in higher interest income earned. Cash, cash equivalents, restricted cash and marketable securities increased to \$459.7 million at September 30, 2006 compared with \$203.8 million at September 24, 2005. We completed an equity follow-on offering in March 2006, which resulted in net proceeds of \$182.0 million. Other income (expense) for the three months ended September 30, 2006 and September 24, 2005 was mainly comprised of foreign currency gains (losses), primarily related to Japanese Yen.

20

Provision for Income Taxes

	Three Months Ended						
	September 30, 2006		Effective Tax Rate	September 24 2005	, Effective Tax Rate		
Benefit from (provision for) income taxes	\$	(7,675)	32.7%	6 \$ 758	8 (8.4)%		

Our effective tax rate for the three months ended September 30, 2006 and September 24, 2005 was 32.7% and (8.4%), respectively. The increase in the effective tax rate between this year and last was primarily due to lower federal research and development tax credits, due to the expiration of the credit on December 31, 2005, and non-deductible stock based compensation expense resulting from the adoption of SFAS No. 123 (R). In addition, our tax provisions for both 2006 and 2005 benefited from the expiration of the statute of limitations for certain previously provided tax reserves. The benefit provision for September 24, 2005 also reflected the impact of \$3.0 million related to a research and development tax credit study.

Nine Months Ended September 30, 2006 and September 24, 2005

Revenues

Nine Months Ended							
Ser	September 30, 2006		September 24, 2005			Change %	
			(In tho	usand	ls)		
\$	203,631	\$	128,135	\$	75,496	58.9%	
	41,247		23,545		17,702	75.2	
	25,642		13,996		11,646	83.2	
\$	270,520	\$	165,676	\$	104,844	63.3%	
		\$ 203,631 41,247 25,642	\$ 203,631 \$ 41,247 25,642	September 30, 2006 September 24, 2005 (In the \$ 203,631 \$ 128,135 41,247 23,545 25,642 13,996	September 30, 2006 September 24, 2005 (() (In thousand \$ 203,631 \$ 128,135 \$ 41,247 23,545 25,642 13,996	September 30, 2006 September 24, 2005 Increase (decrease) (In thousands) (In thousands) \$ \$ 203,631 \$ 128,135 \$ 75,496 41,247 23,545 17,702 25,642 13,996 11,646	

Revenues increased 63.3% in the nine months ended September 30, 2006 compared with the nine months ended September 24, 2005. The increase was mainly driven by strong market demand for our advanced wafer probe cards due to a variety of factors, including the increased demand for mobile and consumer applications, increased design activity and bit growth, the transition to advanced technology nodes such as 90 nanometer and below, and the ongoing build-out of 300mm factories.

The majority of our revenues for the nine months ended September 30, 2006 were generated by sales of wafer probe cards to manufacturers of DRAM devices. The increase was primarily due to the ongoing transition to advanced technology nodes, such as 90 nanometer and below, the conversion to DDR II and the ongoing build-out of 300mm factories. Approximately 77% of our DRAM revenues in the first nine months of fiscal 2006 were derived from 90 nanometer and below technology products.

Revenues generated from sales to flash memory device manufacturers increased for both our NAND and NOR flash wafer probe cards. Consumer applications which utilize multi-chip packages were a major driver for both categories of flash devices.

Revenues from manufacturers of logic devices increased primarily due to increased demand for high parallelism test products. The majority of our logic revenues in the nine months ended September 30, 2006 was derived from sales of wafer probe cards to test high performance flip chip microprocessor and chipset applications used in personal computer, gaming and graphics applications.

Revenue by Geographic Region

The following table sets forth our revenues by geographic region for the periods indicated.

		Nine Months Ended								
	Sej	otember 30, 2006	% of S Revenues	September 24, 2005	% of Revenues					
			(In thousa	nds)						
Asia Pacific	\$	90,357	33.4% \$	52,529	31.7%					
Japan		82,687	30.6	35,783	21.6					
North America		80,198	29.6	60,483	36.5					
Europe		17,278	6.4	16,881	10.2					
Total Revenues	\$	270,520	100.0% \$	165,676	100.0%					

Geographic revenue information is based on the invoicing location of the customer. For example, certain Korean customers purchase through their North American subsidiaries. The increase in revenues in North America was primarily driven by demand for wafer probe cards used to test chips for consumer and mobile products. The increase in revenues in Japan was primarily due to increased sales to a manufacturer of DRAM devices. The increase in revenues in Asia Pacific was primarily due to growth in our business with Taiwanese customers. The increase in revenues in Europe was primarily due to increased sales to a manufacturer of DRAM devices in this region.

The following customers accounted for more than 10% of our revenues for the nine months ended September 30, 2006 and September 24, 2005:

	Nine Mont	hs Ended
	September 30, 2006	September 24, 2005
Elpida	23.4	18.4
Powerchip	13.4	*
Intel Corporation	11.8	10.7
Samsung	*	18.4
Spirox Corporation	*	25.7%

* Less than 10% of revenues.

Revenues from Spirox Corporation decreased in the nine months ended September 30, 2006 compared to the nine months ended September 24, 2005 primarily due to our transition to a direct sales model in Taiwan. Until October 17, 2005, we relied upon Spirox Corporation to sell our products in Taiwan (including Powerchip). While Spirox continues to serve as our distributor in Singapore, Philippines, Malaysia and China, a significant percentage of their sales occurred in Taiwan.

Gross Margin

			Nine Months	s Ended	
	Sep	otember 30, 2006	% of Revenues	September 24, 2005	% of Revenues
			(in thousa	inds)	
Gross margin	\$	139,821	51.7%	\$ 72,192	43.6%
Stock-based compensation expense		2,735	1.0	379	0.2
Gross margin excluding stock-based compensation	\$	142,556	52.7%	\$ 72,571	43.8%

The increase in gross margin in the nine months ended September 30, 2006 compared with the nine months ended September 24, 2005 was primarily due to factory productivity, yield improvements and product mix enabling revenue growth, which in turn improved gross margin. The increase was offset in part by an increase in our inventory write downs. Inventory write downs increased to \$9.2 million, or 3.4% of revenues for the nine months ended September 30, 2006 as compared with \$8.1 million, or 4.9% of revenues for the nine months ended September 24, 2005 primarily due to excess custom inventory quantities. As our advanced wafer probe cards are manufactured in low volumes and must be delivered on relatively short lead-times, it is not uncommon for us to acquire production materials and start certain production activities based on estimated production yields and forecasted demand prior to or in excess of actual demand for our wafer probe cards. Gross margin for the nine months ended September 24, 2005 was impacted by non-recurring factory start up costs of \$11.5 million. Stock-based compensation expense increased by \$2.4 million for the nine months ended September 30, 2006 due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006.

Research and Development

			Nine Months	5 Ended	
	Sep	otember 30, 2006	% of Revenues	September 24, 2005	% of Revenues
			(in thousa	nds)	
Research and development	\$	33,397	12.4%	\$ 19,461	11.7%
Stock-based compensation expense		3,702	1.4	577	0.3
Research and development excluding stock-based compensation	\$	29,695	11.0%	\$ 18,884	11.4%

The increase in research and development expenses in absolute dollars was mainly due to an increase of approximately \$7.1 million in personnel costs, an increase of approximately \$3.7 million in development related costs and an increase of \$3.1 million in stock-based compensation expense due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006. We plan to continue to invest in the development of our next generation Harmony architecture and products, fine pitch memory and logic products, advanced MicroSpring interconnect technology and new process technologies. We are also making incremental investments in new technologies and products as we focus on new market opportunities.

Selling, General and Administrative

		Nine Months Ended			
	Sep	otember 30, 2006	% of Revenues	September 24, 2005	% of Revenues
			(in thousa	nds)	
Selling, general and administrative	\$	53,034	19.6% \$	\$ 31,283	19.0%
Stock-based compensation expense		8,579	3.2	2,055	1.3
Selling, general and administrative excluding stock-based compensation	\$	44,455	16.4%	\$ 29,228	17.7%

The increase in selling, general and administrative expenses in absolute dollars was mainly due to an increase of approximately \$12.5 million in personnelrelated expenses, an increase of approximately \$1.1 million in outside professional services that primarily related to patent enforcement proceedings and other consulting services, including compliance and regulatory matters, and an increase of \$6.5 million in stock-based compensation expense due to the adoption of SFAS No. 123 (R) in the first quarter of fiscal 2006.

Interest and Other Income (Expense), Net

			Nine Mo	nths End	led	
	Sep	tember 30, 2006	% of Revenues		mber 24, 2005	% of Revenues
			(in the	ousands)		
Interest income	\$	10,196	3.89	%\$	2,912	1.8%
Other income (expense), net	\$	45	0.09	%\$	(655)	(0.4)%

The increase in interest income was due to larger cash, cash equivalents and marketable securities balances throughout the first nine months of fiscal 2006 compared to the first nine months of fiscal 2005 and higher interest rates, resulting in higher interest income earned. Cash, cash equivalents, restricted cash and marketable securities increased to \$459.7 million at September 30, 2006 compared with \$203.8 million at September 24, 2005. We completed an equity follow-on offering in March 2006, which resulted in net proceeds of \$182.0 million. Other expense for the nine months ended September 30, 2006 and September 24, 2005 was mainly comprised of foreign currency losses, primarily related to Japanese Yen.

Provision for Income Taxes

		Nine Months Ended			
	Sep	tember 30, 2006	Effective Tax Rate	September 24, 2005	Effective Tax Rate
			(in thou	sands)	
Provision for income taxes	\$	21,763	34.2%	\$ 4,004	19.9%

Our effective tax rate for the nine months ended September 30, 2006 and September 24, 2005 was 34.2% and 19.9%, respectively. The increase in the tax rate between this year and last was primarily due to lower federal research and development tax credits, due to the expiration of the credit on December 31, 2005, and non-deductible stock based

compensation expense resulting from the adoption of SFAS No. 123 (R). In addition, our tax provisions for both 2006 and 2005 benefited from the expiration of the statute of limitations for certain previously provided tax reserves.

We expect our effective tax rate for fiscal 2006 to be between 35% and 36%, however it could fluctuate due to lower than expected foreign sales, disqualifying dispositions of incentive stock options and employee stock purchase plan shares, tax exempt interest income and earnings.

Critical Accounting Policies and Estimates

For a description of the critical accounting policies that affect our more significant judgments and estimates used in the preparation of our condensed consolidated financial statements, refer to our Annual Report on Form 10-K filed with the SEC. There have been no changes to our critical accounting policies since December 31, 2005, with the exception of equity based compensation.

During the first quarter of fiscal 2006, we implemented SFAS No. 123 (R) as a new critical accounting policy related to equity based compensation. Beginning January 1, 2006, we began accounting for stock options and ESPP shares under the provisions of SFAS No. 123 (R), which requires the

recognition of the fair value of equity based compensation. The fair value of stock options and ESPP shares was estimated using the Black-Scholes option valuation model. This model requires the input of subjective assumptions in implementing SFAS No. 123 (R), including expected stock price volatility, estimated life and estimated forfeitures of each award. The fair value of equity-based awards is amortized over the vesting period of the award, and we have elected to use the straight-line method. We make quarterly assessments of the adequacy of the tax credit pool to determine if there are any deficiencies which require recognition in the condensed consolidated income statements. Prior to the implementation of SFAS No. 123 (R), we accounted for stock options and ESPP shares under the provisions of APB No. 25, "Accounting for Stock Issued to Employees" and made pro forma footnote disclosures as required by SFAS No. 148, "Accounting for Stock-Based Compensation – Transition and Disclosure," which amended SFAS No. 123, "Accounting for Stock-Based Compensation." Pro forma net income and pro forma net income per share disclosed in the footnotes to the condensed consolidated financial statements were estimated using the Black-Scholes option valuation model. The fair value of restricted stock units was calculated based upon the fair market value of our common stock at the date of grant.

We have elected to adopt the alternative transition method provided under the provisions of FASB Staff Position No. FAS 123 (R)- 3 "Transition Election Related to Accounting for Tax Effects of Share-Based Payment Awards." The alternative transition method includes simplified methods to establish the beginning balance of the additional paid-in capital pool, or APIC pool, related to the tax effects of employee stock-based compensation, and to determine the subsequent impact on the APIC pool and consolidated statements of cash flows of the tax effects of employee stock-based compensation awards that are outstanding upon adoption of SFAS No. 123 (R).

Liquidity and Capital Resources

As of September 30, 2006, we had \$457.5 million in cash, cash equivalents and marketable securities compared to \$211.6 million as of December 31, 2005.

Net cash provided by operating activities was \$58.8 million and \$22.1 million for the nine months ended September 30, 2006 and September 24, 2005, respectively. Net cash provided by operating activities increased for the first nine months of fiscal 2006 compared to the first nine months of fiscal 2005 primarily as a result of the \$22.2 million increase in net income in the first nine months of fiscal 2006 as well as better working capital management.

Accounts receivable increased by \$10.1 million and \$10.4 million for the nine months ended September 30, 2006 and September 24, 2005, respectively, due to an increase in worldwide sales. Our days sales outstanding from receivables, or DSO, remained fairly flat at 42 days at September 30, 2006 compared to 43 days at September 24, 2005, respectively.

Inventories were \$26.3 million and \$14.4 million at September 30, 2006 and September 24, 2005, respectively. The increase in inventories was a result of increased volume in business and strong demand for our products. Inventory turns were 8.4 and 9.7 in the third quarter of fiscal 2006 and the third quarter of fiscal 2005, respectively.

Net cash used in investing activities was \$28.5 million for the nine months ended September 30, 2006 compared with net cash used by investing activities of \$43.9 million for the nine months ended September 24, 2005. Net cash used in investing activities resulted primarily from the net purchase of marketable securities and capital expenditures in each of these periods. Capital expenditures were \$26.9 million for the nine months ended September 30, 2006 and \$21.3 million for the nine months ended September 24, 2005. The increase in capital expenditures was due primarily to our investments in factory equipment to increase our capacity, as well as investments in our new offices in Japan.

24

Net cash provided by financing activities was \$214.7 million for the nine months ended September 30, 2006 compared with \$10.6 million for the nine months ended September 24, 2005. Net cash provided by financing activities for the nine months ended September 30, 2006 was mainly due to \$182.0 million of net proceeds received from an equity follow-on offering executed in March 2006 as well as proceeds of \$15.1 million received from the exercise of employee stock options and \$4.5 million received from ESPP purchases. Tax benefits related to the exercise of stock options during the nine months ended September 30, 2006 were \$13.4 million. Net cash provided by financing activities for the nine months ended September 24, 2005 was mainly due to proceeds received from the exercise of employee stock options.

We expect capital expenditures for fiscal 2006 to average between 8% and 10% of our revenues.

We believe that cash generated from operations, together with the liquidity provided by our existing cash, cash equivalents and marketable securities, will be sufficient to meet our anticipated cash needs for at least the next 12 months. Our future capital requirements will depend on many factors, including the timing and extent of spending to support product development efforts, the expansion of sales and marketing activities, and the cost of increasing manufacturing capacity to meet projected demand. Although we are currently not a party to any agreement or letter of intent with respect to potential investments in, or acquisitions of, complementary businesses, products or technologies, we may enter into these types of arrangements in the future, which could also require us to seek additional equity or debt financing. Additional funds may not be available on terms favorable to us or at all.

Off-Balance Sheet Arrangements

As part of our ongoing business, we do not participate in transactions that generate relationships with unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities, or SPEs, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. As of September 30, 2006 we are not involved in any off-balance sheet arrangements as defined in Regulation S-K, Item 303 (a)(4).

Recent Accounting Pronouncements

In September 2006, the FASB issued SFAS No. 157, "Fair Value Measurements" ("SFAS No. 157"). SFAS No. 157 defines fair value, establishes a framework for measuring fair value in accordance with generally accepted accounting principles, and expands disclosures about fair value measurements. This statement does not require any new fair value measurements; rather, it applies under other accounting pronouncements that require or permit fair value measurements. The provisions of this statement are to be applied prospectively as of the beginning of the fiscal year in which this statement is initially applied, with any transition adjustment recognized as a cumulative-effect adjustment to the opening balance of retained earnings. The provisions of SFAS

No. 157 are effective for the fiscal years beginning after November 15, 2007; therefore, we anticipate adopting this standard as of January 1, 2008. We have not determined the effect, if any, the adoption of this statement will have on our consolidated financial statements.

In September 2006, the SEC issued Staff Accounting Bulletin No. 108, "Considering the Effects of Prior Year Misstatements when Quantifying Misstatements in Current Year Financial Statements" ("SAB No. 108"), to eliminate the diversity of practice surrounding how public companies quantify financial statement misstatements. Traditionally, there have been two widely-recognized methods for quantifying the effects of financial statement misstatements: the "roll-over" method and the "iron curtain" method. The roll-over method focuses primarily on the impact of a misstatement on the income statement, including the reversing effect of prior year misstatements, but its use can lead to the accumulation of misstatements in the balance sheet. The iron-curtain method, on the other hand, focuses primarily on the effect of correcting the period-end balance sheet with less emphasis on the reversing effects of prior year errors on the income statement. In SAB No. 108, the SEC Staff established an approach that requires quantification of financial statement misstatements based on the effects of the misstatements on each financial statement and the related financial statement disclosures. This model is commonly referred to as a "dual approach" because it requires quantification of errors under both the iron curtain and the roll-over methods. We are required to adopt SAB No. 108 in the fourth quarter of 2006 and are currently evaluating the impact of this interpretation on our consolidated financial statements.

In July 2006, the FASB issued FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes, an Interpretation of FASB Statement No. 109" ("FIN 48"). FIN 48 provides guidance on the financial statement recognition and

0	-
,	ъ
-	-

measurement of a tax position taken or expected to be taken in a tax return. FIN 48 requires that we recognize in the financial statements the impact of a tax position if that position will more likely than not be sustained on audit, based on the technical merits of the position. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosures, and transition provisions. FIN 48 is effective for fiscal years beginning after December 15, 2006, and we will adopt FIN 48 as of January 1, 2007. We are currently evaluating the impact of this interpretation on our consolidated financial statements.

In July 2006, the FASB issued EITF Issue No. 06-3, "How Taxes Collected from Customers and Remitted to Governmental Authorities Should be Presented in the Income Statement (that is, Gross versus Net Presentation)." The adoption of EITF No. 06-3 did not have an impact on our consolidated financial statements. Our accounting policy has been to present above mentioned taxes on a net basis, excluded from revenues.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Foreign Currency Exchange Risk. Our revenues, except in Japan, and our expenses, except those expenses related to our Germany, United Kingdom, Japan, Korea and Taiwan operations, are denominated in U.S. dollars. Revenues and accounts receivable from our Japanese customers are denominated in Japanese Yen. We may purchase from time to time forward exchange contracts to hedge certain existing foreign currency denominated accounts receivable. Gains and losses on these contracts are generally recognized in income when the related transactions being hedged are recognized. As of September 30, 2006, we had one outstanding foreign exchange forward contract to sell 2,600,000,000 Japanese Yen for \$22,090,059 with a contract rate of 117.70 Japanese Yen per U.S. Dollar. The fair value on this foreign currency forward exchange contract as of September 30, 2006 would have changed by \$2,209,006 if the foreign currency exchange rate for the Japanese Yen to the U.S. Dollar on this forward contract had changed by 10%. We do not use derivative financial instruments for trading or speculative purposes.

Interest Rate Risk. The primary objective of our investment activities is to preserve principal while at the same time maximizing the income we receive from our investments without significantly increasing risk. Some of the securities in which we invest may be subject to market risk. This means that a change in prevailing interest rates may cause the principal amount of the investment to fluctuate. For example, if we hold a security that was issued with an interest rate fixed at the then-prevailing rate and the prevailing interest rate later rises, the principal amount of our investment will probably decline. To minimize this risk, we maintain our portfolio of cash equivalents, and marketable securities in a variety of securities, including commercial paper, money market funds, government and non-government debt securities and certificates of deposit. The risk associated with fluctuating interest rates is limited to our investment portfolio and we do not believe that a 10% change in interest rates will have a significant impact on our consolidated statements of income and statements of cash flow. As of September 30, 2006, all of our investments were in money market accounts, certificates of deposit or high quality corporate debt obligations and U.S. government securities.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Our management, including our Chief Executive Officer and Chief Financial Officer, conducted an evaluation as of September 30, 2006 of the effectiveness of our "disclosure controls and procedures" as defined in Exchange Act Rule 13a-15(e). Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that as of September 30, 2006, our disclosure controls and procedures were effective to ensure that information we are required to disclose in reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in rules and forms of the SEC, and is accumulated and communicated to our management as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control Over Financial Reporting

Our management, including our Chief Executive Officer and Chief Financial Officer, conducted an evaluation of our "internal control over financial reporting" as defined in Exchange Act Rule 13a-15(f) to determine whether any changes in our internal control over financial reporting occurred during the third quarter of fiscal 2006 that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting. Based on that evaluation, there have been no such changes in our internal control over financial reporting during the third quarter of fiscal 2006.

CEO and CFO Certifications

We have attached as exhibits to this Quarterly Report on Form 10-Q the certifications of our Chief Executive Officer and Chief Financial Officer, which are required in accordance with the Exchange Act. We recommend that this Item 4 be read in conjunction with the certifications for a more complete understanding of the subject matter presented.

Limitations on the Effectiveness of Controls

Control systems, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control systems' objectives are being met. Further, the design of any control systems must reflect the fact that there are resource constraints, and the benefits of all controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within our company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty and that breakdowns can occur because of simple error or mistake. Control systems can also be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the controls. The design of any system of controls is based in part on certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Over time, controls may become inadequate because of changes in conditions or deterioration in the degree of compliance with policies or procedures.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

The information relating to "Legal Matters" set forth under Note 8 - Commitments and Contingencies of the Notes to Unaudited Condensed Consolidated Financial Statements of this Form 10-Q is incorporated herein by reference.

Item 1A. Risk Factors

In addition to the other information in this Quarterly Report on Form 10-Q, you should carefully consider the risk factors discussed in our Annual Report on Form 10-K for the year ended December 31, 2005, the risk factors set forth in our Quarterly Report on Form 10-Q for the period ending July 1, 2006 and the updated risk factors set forth below in evaluating FormFactor and our business. If any of the identified risks actually occur, our business, financial condition and results of operations could suffer. Accordingly, the trading price of our common stock would likely decline and you may lose all or part of your investment in our common stock. The risks and uncertainties described in our Annual Report on Form 10-K and below are not the only ones we face. Additional risks that we currently do not know about or that we currently believe to be immaterial may also impair our business operations.

We disclosed the risk factors below in our Annual Report on our Form 10-K and in our Form 10-Q for the quarter ended July 1, 2006. We have updated these risk factors to reflect changes in the third quarter of fiscal 2006 for the period ending September 30, 2006.

If demand for our products in the memory device and flip chip logic device markets declines or fails to grow as we anticipate, our revenues could decline.

We derive substantially all of our revenues from wafer probe cards that we sell to manufacturers of dynamic random access memory, or DRAM, memory and flash memory devices and manufacturers of microprocessor, chipset and other logic devices. In the microprocessor, chipset and other logic device markets, our products are primarily used for devices employing flip chip packaging, which are commonly referred to as flip chip logic devices. For the three months ended September 30, 2006 and September 24, 2005, sales to manufacturers of DRAM devices accounted for 72.8% and 75.8%, respectively, of our revenues, sales to manufacturers of logic devices accounted for 10.0% and 10.2%, respectively, of our revenues, and sales to manufacturers of flash memory devices accounted for 17.2% and 14.0%, respectively, of our revenues. For the nine months ended September 30, 2006 and September 24, 2005, sales to manufacturers of DRAM devices accounted for 75.3% and 77.4%, respectively, of our revenues, sales to manufacturers of logic devices accounted for 9.5% and 8.4%, respectively, of our revenues, and sales to manufacturers of flash memory devices accounted for 15.2% and 14.2 %, respectively, of our revenues. Therefore, our success depends in part upon the continued acceptance of our products within these markets and our ability to continue to develop and introduce new products on a timely basis for these markets. In particular, to continue to grow our business, we need to further penetrate the flash memory market and to gain additional market share with manufacturers in this market. To the extent that we are unable to do so, or if we are not able to deliver timely our products for testing flash memory device wafers, our ability to grow could suffer. If chip manufacturers fail to make architecture, node or technology transitions as we anticipate, or if anticipated or announced transitions push out or are delayed, it could adversely impact our revenues and operating results.

A substantial portion of these semiconductor devices is sold to manufacturers of personal computers and computer-related products and to manufacturers of personal electronic devices. Both the personal computer market and the personal electronic devices market have historically been characterized by significant fluctuations in demand and continuous efforts to reduce costs, which in turn have affected the demand for and price of memory devices and microprocessors. The personal computer market and the personal electronic devices market might not grow in the future at historical rates or at all and design activity in those markets might decrease, which could negatively affect our revenues and operating results.

27

We derive a substantial portion of our revenues from a small number of customers, and our revenues could decline significantly if any major customer cancels, reduces or delays a purchase of our products.

A relatively small number of customers has accounted for a significant portion of our revenues in any particular period. In the three and nine months ended September 30, 2006, four customers accounted for 57.7% and 57.2%, respectively, of our revenues. In the three and nine months ended September 24, 2005, four customers accounted for 80.1% and 73.2%, respectively, of our revenues. Our ten largest customers accounted for 89.0% and 88.6%, respectively, of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three and nine months ended September 30, 2006 and 98.6% and 97.4%, respectively of our revenues in the three a

manufacturers of DRAM, could reduce our customer base and lead to lost or delayed sales and reduced demand for our wafer probe cards. Industry consolidation also could result in pricing pressures as larger DRAM manufacturers could have sufficient bargaining power to demand reduced prices and favorable nonstandard terms. Additionally, certain customers may not want to rely entirely or substantially on a single wafer probe card supplier and, as a result, such customers could reduce their purchases of our wafer probe cards.

We manufacture all our products at our facilities in Livermore, California, and any disruption in the operations of these facilities could adversely impact our business and operating results.

Our processes for manufacturing our wafer probe cards require sophisticated and costly equipment and a specially designed facility, including a semiconductor clean room. We manufacture our wafer probe cards at our new facility located in Livermore, California. We also continue to use our old manufacturing facility for some steps in our overall manufacturing processes, which is also located in Livermore. Any disruption in the operation of our facilities, whether due to contamination in our manufacturing process, technical or labor difficulties, destruction or damage from fire or earthquake, infrastructure failures such as power or water shortage or any other reason, could interrupt our manufacturing operations, impair critical systems, disrupt communications with our customers and suppliers, and cause us to write off inventory, thereby potentially resulting in the loss of revenues. We could experience manufacturing disruptions if we close the old facility, or if we cease use of the old facility for any steps in our overall manufacturing processes. In addition, if the previous energy crises in California that resulted in disruptions in power supply and increases in utility costs were to recur, we might experience power interruptions and shortages, which could disrupt our manufacturing operations. This could subject us to loss of revenues as well as significantly higher costs of energy. Further, current and potential customers might not purchase our products if they perceive our lack of a fully operational alternate manufacturing facility to be a risk to their continuing source of supply.

If we do not effectively expand our manufacturing capacity, our business and operating results will be negatively impacted.

We completed the transition to our new manufacturing facility in Livermore and the first phase of our capacity ramp. We are presently further ramping production volume at our new facility to meet anticipated short term and long term demand for our existing products and for our contemplated new product introductions. We also plan to expand our manufacturing into Singapore. For this expansion, we currently plan to first expand our assembly and test and back-end manufacturing processes and then expand our Singapore manufacturing capabilities and capacities to include our front-end manufacturing processes. The costs of increasing manufacturing capacity at our current Livermore facilities and of starting up any new facility, including capital costs such as equipment, fixed costs such as rent, personnel and material costs required for any ramp and qualification, and redundancy costs of maintaining production sites in parallel, are substantial. Because Singapore is geographically separate from our Livermore campus, the start-up of any facility in Singapore could have additional logistical, supply, equipment and engineering issues as compared to the start-up of a facility near our Livermore campus. Any difficulties with expanding manufacturing capacity at our current facilities or bringing up any new facility, could result in significant expense, disrupt our manufacturing processes, and cause delays in product deliveries and lost sales, which would harm our business, financial condition and operating results. Some or all of our customers may also require a full qualification of any new facility. Any qualification process could take longer than we anticipate and negatively impact our operating results.

28

If we fail to protect our proprietary rights, our competitors might gain access to our technology, which could adversely affect our ability to compete successfully in our markets and harm our operating results.

If we fail to protect our proprietary rights adequately, our competitors might gain access to our technology. Unauthorized parties might attempt to copy aspects of our products or to obtain and use information that we regard as proprietary. Others might independently develop similar or competing technologies or methods or design around our patents. In addition, the laws of many foreign countries in which we or our customers do business do not protect our intellectual property rights to the same extent as the laws of the United States. As a result, our competitors might offer similar products and we might not be able to compete successfully. We also cannot assure that:

- · our means of protecting our proprietary rights will be adequate;
- · patents will be issued from our currently pending or future applications;
- our existing patents or any new patents will be sufficient in scope or strength to provide any meaningful protection or commercial advantage to us;
- any patent, trademark or other intellectual property right that we own will not be invalidated, circumvented or challenged in the United States or foreign countries; or
- others will not misappropriate our proprietary technologies or independently develop similar technology, duplicate our products or design around any
 patent or other intellectual property rights that we own, or attempt to manufacture and sell infringing products in countries that do not strongly
 enforce intellectual property rights.

We might be required to spend significant resources to monitor and protect our intellectual property rights. We presently believe that it is likely that one or more of our competitors are using methodologies or have implemented structures into certain of their products that are covered by one or more of our intellectual property rights. We have in the past brought claims to protect our rights, and, in certain cases, our competitors have initiated invalidity proceedings in foreign patent offices against certain of our patents. For example, in connection with our litigation with Phicom Corporation, one of our competitors, on or about October 27, 2005, the Korean Patent Court issued rulings holding invalid certain claims of two of our Korean patents. The two Korean patents affected by the rulings are Nos. 278,342, entitled "Method of Altering the Orientation of Probe Elements in a Probe Card Assembly," and 399,210, entitled "Probe Card Assembly", both of which had previously been upheld by the Korean Intellectual Property Office when it dismissed validity challenges in their entirety. On or about February 9, 2006, the Korea Patent Court issued a ruling holding invalid certain claims of our Korean Patent No. 324,064, entitled "Contact Tip Structures for Microelectronic Interconnection Elements and Methods of Making Same". On or about June 15, 2006, the Korea Patent Court issued a ruling holding invalid certain claims of our Korean Supreme Court the adverse decision on its Patent No. 278,342, 399,210 and 324,064; Phicom has appealed the June 15 ruling on Korea Patent No. 252,457 to the Korea Supreme Court. The outcome of any appeal of the rulings can not be definitively predicted, but will result in our incurring additional legal fess and expenses; which most likely will be material. See the "Legal Proceedings" section of this Form 10-Q for a description of the infringement actions we have brought against Phicom and other third parties are pursuing against certain of our patents.

While we do not have a material monetary damages exposure in these various invalidity proceedings, it is possible we will incur material expenses in our litigation with Phicom or in defending our intellectual property more broadly. Any litigation, whether or not it is resolved in our favor, could result in significant expense to us and divert the efforts of our management and technical personnel. In addition, while patents are territorial and a ruling on a certain given patent does not necessarily impact the validity or enforceability of a corresponding or related patent in a different country, an adverse ruling in one country might negatively impact our ability to enforce the corresponding or related patent in other countries. Finally, certain of our customer contracts contain provisions that require us to indemnify our customers for third party intellectual property infringement claims, which would increase the cost to us of an adverse ruling in such a claim. An adverse determination could also negatively impact our ability to license certain of our technologies and methods to others, and result in our competitors being allowed to sell products with, or add to their products, features and benefits contained in our products, thereby reducing our competitive advantages over these competing products.

If we do not effectively manage growth and other changes in our business, these changes could place a significant strain on our management and operations and, as a result, our business might not succeed.

Our ability to grow successfully requires an effective planning, implementation and management process. We are presently ramping production volume at our new manufacturing facility, expanding our product development efforts, increasing our global operations and hiring additional employees domestically and internationally, including for our direct

29

sales force. We also plan to expand our manufacturing into Singapore. Our growth could place a significant strain on our management systems, infrastructure and other resources. To manage our growth effectively, we must invest the necessary capital and continue to improve and expand our controls, systems and infrastructure in a timely and efficient manner. Those resources might not be available when we need them, which would limit our growth. Our controls, systems and procedures might not be adequate to support a growing public company. For example, if we do not implement in a timely manner scalable information technology systems, we may not be able to maintain or expand our current manufacturing capacity, which would, in turn, have a negative impact on our operating results. In addition, if our plans to expand our manufacturing capacity or our global operations involves the acquisition of businesses, we will need to invest the necessary resources, and to improve our corporate systems and infrastructure in order to enable the successful integration of any acquired businesses. If our management fails to plan effectively for our growth initiatives or to respond effectively to changes in our business, our business might not succeed.

Because we conduct some of our business internationally, we are subject to operational, economic, financial and political risks abroad.

Sales of our products to customers outside the United States have accounted for an important part of our revenues. Our international sales as a percentage of our revenues were 68.9% and 70.4%, respectively, for the three and nine months ended September 30, 2006 and 59.3% and 63.5%, respectively for the three and nine months ended September 24, 2005. Additionally, certain of our Korean customers purchase through their North American subsidiaries. In the future, we expect international sales, particularly in Europe, Japan, South Korea and Taiwan, to continue to account for a significant percentage of our revenues. Accordingly, we will be subject to risks and challenges that we would not otherwise face if we conducted our business only in the United States. These risks and challenges include:

- · compliance with a wide variety of foreign laws and regulations;
- · legal uncertainties regarding taxes, tariffs, quotas, export controls, export licenses and other trade barriers;
- · political and economic instability in, or foreign conflicts that involve or affect, the countries of our customers;
- · difficulties in collecting accounts receivable and longer accounts receivable payment cycles;
- · difficulties in staffing and managing personnel, distributors and representatives;
- · reduced protection for intellectual property rights in some countries;
- currency exchange rate fluctuations, which could affect the value of our assets denominated in local currency, as well as the price of our products relative to locally produced products;
- · seasonal fluctuations in purchasing patterns in other countries; and
- fluctuations in freight rates and transportation disruptions.

Any of these factors could harm our existing international operations and business or impair our ability to continue expanding into international markets.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Default Upon Senior Securities

None.

Item 4. Submission of Matters To A Vote of Security Holders

None.

Item 5. Other Information

30

Item 6. Exhibits

The following exhibits are filed herewith:

Exhibit Number	Exhibit Description	In Form	corporated by Re Date	ference Number	Filed Herewith
10.01	Pacific Corporate Center Lease dated as of September 7, 2004 by and between Greenville Holding Company LLC and the Registrant, as amended by First Amendment to Building 6 Lease dated as of August 16, 2006				Х
31.01	Certification of Chief Executive Officer pursuant to 15 U.S.C. Section 7241, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002				Х
31.02	Certification of Chief Financial Officer pursuant to 15 U.S.C. Section 7241, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002				Х
32.01*	Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002				Х

* This exhibit shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934, whether made before or after the date hereof and irrespective of any general incorporation language in any filings.

31

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

FORMFACTOR, INC.

By:

/s/ RON C. FOSTER

Ron C. Foster Chief Financial Officer (Principal Financial Officer and Duly Authorized Officer)

November 7, 2006

32

EXHIBIT INDEX

Exhibit		Ir	corporated by Re	eference	Filed
Number	Exhibit Description	Form	Date	Number	Herewith
10.01	Pacific Corporate Center Lease dated as of September 7, 2004 by and between Greenville Holding Company LLC and the Registrant, as amended by First Amendment to Building 6 Lease dated as of August 16, 2006				Х
31.01	Certification of Chief Executive Officer pursuant to 15 U.S.C. Section 7241, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002				Х
31.02	Certification of Chief Financial Officer pursuant to 15 U.S.C. Section 7241, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002				Х

- 32.01* Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
- * This exhibit shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934, whether made before or after the date hereof and irrespective of any general incorporation language in any filings.

PACIFIC CORPORATE CENTER LEASE

by and between

GREENVILLE INVESTORS, L.P.,

a California limited partnership

as "Landlord"

and

FORMFACTOR, INC.,

a Delaware corporation

as "Tenant"

Dated as of September 7, 2004

ARTICLE 1.	BASIC TERMS	1
ARTICLE 2.	TERM	2
ARTICLE 3.	BASE RENT	2
ARTICLE 4.	USE OF PREMISES	3
ARTICLE 5.	LETTERS OF CREDIT/SECURITY DEPOSIT	6
ARTICLE 6.	UTILITIES	8
ARTICLE 7.	REAL PROPERTY TAXES	9
ARTICLE 8.	CONSTRUCTION AND ACCEPTANCE	10
ARTICLE 9.	REPAIRS AND MAINTENANCE	11
ARTICLE 10.	OPERATING AND MAINTENANCE COSTS	13
ARTICLE 11.	TRADE FIXTURES AND SURRENDER	16
ARTICLE 12.	DAMAGE OR DESTRUCTION	17
ARTICLE 13.	EMINENT DOMAIN	18
ARTICLE 14.	INSURANCE	18
ARTICLE 15.	WAIVER OF SUBROGATION	20
ARTICLE 16.	RELEASE AND INDEMNITY	20
ARTICLE 17.	INSOLVENCY, ETC. OF TENANT	22
ARTICLE 18.	PERSONAL PROPERTY AND OTHER TAXES	22
ARTICLE 19.	SIGNS	23
ARTICLE 20.	ASSIGNMENT AND SUBLETTING	23
ARTICLE 21.	RIGHTS RESERVED BY LANDLORD	24
ARTICLE 22.	INTENTIONALLY DELETED	25
ARTICLE 23.	RIGHT OF LANDLORD TO PERFORM	25
ARTICLE 24.	LANDLORD DEFAULT	25
ARTICLE 25.	DEFAULT AND REMEDIES	26
ARTICLE 26.	PRIORITY OF LEASE AND ESTOPPEL CERTIFICATE	29
ARTICLE 27.	HOLDING OVER	30
ARTICLE 28.	NOTICES	31
ARTICLE 29.	LIENS	31
ARTICLE 30.	QUIET ENJOYMENT	32
ARTICLE 31.	ATTORNEYS' FEES	32
ARTICLE 32.	MISCELLANEOUS	32

i

SCHEDULE OF EXHIBITS

EXHIBIT A	SITE PLAN	A-1
EXHIBIT A-1	DEPICTION OF BUILDING 6 DRIP LINE	A1-1
EXHIBIT B	CENTER LEGAL DESCRIPTION AND PARCEL MAP	B-1
EXHIBIT C	WORK LETTER	C-1
EXHIBIT D	INTENTIONALLY DELETED	
EXHIBIT E	RULES AND REGULATIONS	E-1

ii

PACIFIC CORPORATE CENTER LEASE

THIS LEASE is made and entered into as of September 7, 2004 by and between **Greenville Investors**, **L.P.** a California limited partnership (hereafter, "Landlord"), and FormFactor, Inc., a Delaware corporation (hereafter "Tenant").

A. DEMISE. Landlord hereby leases, demises and lets to Tenant, and Tenant hereby leases, hires and takes from Landlord, those certain premises ("the **Premises**") described as follows:

Approximately 12,303 square feet of gross leasable area ("GLA"), at the west end of the commercial Building designated as Building 6 on the Site Plan attached hereto as Exhibit A ("Building 6") with improvements to be constructed by Tenant in accordance with <u>Article 8</u> and <u>Exhibit C</u> hereof. The precise GLA of the Premises shall be determined pursuant to Section 8.5 below. The exterior walls, roof, air space above and the area beneath the Premises are not demised and their use together with the right to install, maintain, use, repair and replace pipes, ducts, conduits and wires leading through the Premises in locations that will not materially interfere with Tenant's use and serving other parts of Building 6, are hereby reserved to the Landlord, except as otherwise expressly provided herein. The Premises also shall include the service yard area of approximately 1,418 square feet shown on the Site Plan attached as <u>Exhibit A</u> ("Tenant's Facility Yard") which is designated for Tenant's use as (i) a Facilities Pad; and (ii) an enclosed/gated storage yard similar to the storage yard attached to Building 3.

Building 6 is located at 7501 Lawrence Drive, Livermore, California on the real property more particularly described and shown on <u>Exhibit B</u> as Parcel 6 ("**Parcel 6**"). The address assigned to the Premises is 7545 Longard Rd, Livermore, California. Building 6 is a part of Pacific Corporate Center, a common interest development being developed by Landlord in the City of Livermore, Alameda County, California, (the "**Center**") which includes eight (8) parcels of real property together with all buildings and other structures and improvements constructed or to be constructed thereon and is more particularly described and shown on <u>Exhibit B</u>, the Center Legal Description and Parcel Map. All parcels of real property in the Center owned (in whole or in part) by Landlord from time to time are hereinafter collectively referred to as "**Landlord's Parcels**".

B. TERMS, COVENANTS AND CONDITIONS. The parties agree that this Lease is made upon the following terms, covenants and conditions:

ARTICLE 1. BASIC TERMS

In all instances, the basic terms set forth in this <u>Article 1</u> are subject to the main body of the Lease in general and those Articles noted in parentheses in particular.

 (b) Initial Monthly Base Rent: \$0.75 psf of GLA. Estimated Initial Monthly Base Rent (Art 3) is \$9,227. 25** (c) Security: None (Art. 5) (d) Tenant's Initial Estimated Monthly Operating Expense Payment: \$0.084 psf of GLA. Estimated Initial Estimated Monthly Operating Expense Payment is \$1,033.45** (e) Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated (Art. 7) Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated (Art. 7) Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Date) 	(a)	Term: Approximately Seven (7) Lease Years, plus Eight (8) months; four (4) options of 5 years each	(Art.2; Addendum A-2.1)
 (c) Security: None (Art. 5) (d) Tenant's Initial Estimated Monthly Operating Expense Payment: \$0.084 psf of GLA. Estimated Initial Estimated Monthly Operating Expense Payment is \$1,033.45** (e) Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art. 8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Date) 	(b)	5 1 5	
 (d) Tenant's Initial Estimated Monthly Operating Expense Payment: \$0.084 psf of GLA. Estimated Initial Estimated Monthly Operating Expense Payment is \$1,033.45** (e) Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated (Art. 7) Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Pate) 	(\cdot)		(A.++ E)
 GLA. Estimated Initial Estimated Monthly Operating Expense Payment is \$1,033.45** (e) Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Carter Science Science) 	• •	5	· /
 (e) Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated (Art. 7) Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Carter Science) 	(d)	Tenant's Initial Estimated Monthly Operating Expense Payment: \$0.084 psf of	(Art.10)
 Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Carter Science Scienc		GLA. Estimated Initial Estimated Monthly Operating Expense Payment is \$1,033.45 **	
 Initial Estimated Monthly Tax Payment is \$2,165.33** (f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Carter Science Scienc	(e)	Tenant's Initial Estimated Monthly Tax Payment: \$0.176 psf of GLA. Estimated	(Art. 7)
(f) Tentative Commencement Date: November 15, 2004 (Art.8) (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (Art. 2) (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (Art. 4) (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Carter)	~ /		
 (g) Base Rent Commencement Date: Base Rent shall commence on March 15, 2005. All Additional Rent shall commence on the Commencement Date of this Lease. (h) Use: Office and research and development facilities, "clean rooms", and related lawful purposes (i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit Commencement Commencement	(f)		(Art.8)
purposes(i)Tenant Improvement Allowance:\$25.00 psf of GLA. Estimated Tenant(Exhibit Control of Control o		,	(Art. 2)
(i) Tenant Improvement Allowance: \$25.00 psf of GLA. Estimated Tenant (Exhibit C	(h)	-	(Art. 4)
	(i)		(Exhibit C)
	~ ~ ~	Improvement Allowance is \$307,575. **	· · · ·
(j) Articles and Exhibits: This Lease consists of <u>Articles 1</u> through <u>32</u> , Addendum to	(i)	Articles and Exhibits: This Lease consists of Articles 1 through 32, Addendum to	
Lease, and <u>Exhibits A</u> , <u>B</u> , <u>C</u> , <u>D</u> , <u>E</u> , <u>F</u> , <u>F-1</u> , <u>G</u> , <u>H</u> , <u>I</u> and <u>J</u> attached hereto, which are by		Lease, and Exhibits A, B, C, D, E, F, F-1, G, H, I and J attached hereto, which are by	
•		this reference incorporated herein.	

(k) Tenant's Proportionate Share: GLA/49,742 (the GLA for Building 6). Estimated Tenant's Proportionate Share is 24.73%. **

^{**} Estimate based on GLA of 12,303 sf. The exact amount will be determined when the precise GLA is certified in accordance with Section 8.5 below.

ARTICLE 2. TERM

2.1 Landlord and Tenant have entered into a lease dated May 3, 2001 for Building 1 located on Parcel 1 in the Center. That Lease as amended is referred to as the (**"Building 1 Lease**"). The Term of this Lease shall commence on the date (**"Commencement Date**") that the Premises are delivered to Tenant in the Delivery Condition (as defined in Article 8) and the initial term shall terminate at midnight on the date of expiration of the initial term of the Building 1 Lease, i.e. **November 30, 2012**.

See Addendum A-2.1

2.2 The first "Lease Year" shall begin on the Commencement Date and shall expire on the last day of the month, twelve (12) full calendar months next following the Rent Commencement Date set forth in Paragraph 1(g). If the Rent Commencement Date occurs on the first day of the calendar month, then the first Lease Year shall end on the day immediately preceding the first anniversary of the Rent Commencement Date. Subsequent Lease Years shall be each consecutive twelve (12) calendar month period thereafter except for the last Lease Year which may be a partial Lease Year.

2.3 Promptly after the Rent Commencement Date, Landlord and Tenant shall execute a written acknowledgment of the Rent Commencement Date in the form attached hereto as **Exhibit J.**

ARTICLE 3. BASE RENT

3.1 Tenant agrees to pay without offset or deduction of any kind (except as expressly set forth in this Lease) the initial monthly Base Rent amount set forth in <u>Paragraph 1(b)</u> above, as adjusted pursuant to Section 3.2, in advance at Landlord's address on the first day of each calendar month during the Term of this Lease. Tenant's obligation to pay Base Rent shall commence on the Rent Commencement Date. If the Rent Commencement Date is not the first day of a calendar month, the first month's rent shall be prorated on the basis of a thirty (30) day month, and shall be payable with the first full monthly rental due hereunder. Landlord's address shall be as set forth below its signature, or as from time to time designated by Landlord to Tenant in writing.

3.2 As of the date of commencement of the second Lease Year and as of the commencement of each Lease Year during the initial Lease Term thereafter, the monthly Base Rent shall increase by four percent (4%) over the monthly Base Rent in effect immediately preceding the applicable adjustment date.

ARTICLE 4. USE OF PREMISES

4.1 The Premises shall be used and occupied only for the purposes described in <u>Paragraph 1(h)</u> above and for other uses permitted within the light industrial zoning district within which the Premises is located, unless prohibited by the Declaration, and provided Tenant's use otherwise complies with all applicable governmental requirements. Tenant shall not use the Premises for any other purposes without Landlord's prior written consent, which consent may be withheld in Landlord's sole discretion.

4.2 Tenant shall not do or permit to be done in or about the Premises anything which is illegal or unlawful; or which will cause cancellation of any insurance on the building of which the Premises are a part. Tenant shall not obstruct or interfere with the rights of any other tenants and occupants of the Center or their invitees, nor injure them, nor operate the Premises in a manner which unreasonably disturbs other tenants in the use of their premises in the Center. Tenant shall not cause, maintain or permit any nuisance on or about the Premises. Tenant shall not use nor permit the use of the Premises or any part thereof as living quarters.

4.3 Tenant acknowledges that although Landlord has permitted Tenant the use of Premises for the purpose described in this Article, neither Landlord nor any agent of Landlord has made any representation or warranty to Tenant with respect to the suitability of the present zoning of the Building for such use. Tenant assumes all responsibility for investigating the suitability of the zoning for its use and for compliance with all other laws and regulations governing such use.

4.4 Tenant shall have use of, and access to, the Premises twenty four (24) hours per day, three hundred sixty five (365) days per year, subject to the provisions of this Lease and ordinances and regulations of applicable governmental agencies.

4.5 Tenant agrees that, at its own cost and expense, it will comply with and conform to all Legal Requirements (as defined in Section 4.7(d) below) in any way relating to the use or occupancy of the Premises throughout the entire term of this Lease; including the Livermore Fire Code requiring all tenants to obtain fire extinguishers for the Premises and maintain them so that they are fully charged and operational at all times and inspected annually. Further, subject to Landlord's obligation to deliver the Premises to Tenant in the Delivery Condition, Tenant shall thereafter be obligated at its own cost and expense to take such action and perform such work (including structural alterations) to the Premises, as

2

required to comply with the Americans with Disabilities Act ("ADA") and other applicable handicapped access codes. Further, if, and to the extent, due to Tenant's use of, or alterations to, or work performed by Tenant in the Premises, changes, alterations or improvements to Building 6, Parcel 6 or other portions of the Center are required by any governmental agency, Tenant shall be responsible for the costs of such changes, alterations and improvements. Notwithstanding the foregoing, nothing contained herein shall limit or affect any representations, warranties or covenants of Landlord or any of Landlord's contractors with respect to any work performed pursuant to <u>Article 8</u> or <u>Exhibit C</u>. Except to the extent of Tenant's compliance obligations set forth above, Landlord shall be obligated to comply with all Legal Requirements, including, without limitation, the ADA and other applicable handicapped access codes, with respect to all portions of Parcel 6 outside of Building 6, subject to reimbursement as specifically set forth in this Lease and further subject to the terms of the Declaration.

4.6 Tenant shall place no loads upon the floors, walls, ceilings or roof of the Building in excess of the maximum designed load of Building 6 as specified in Exhibit C.

A. **Hazardous Substance; Reportable Uses:** As used herein, the terms "**Hazardous Substance**" and "**HS**" shall mean any product, substance, chemical, material or waste whose presence, nature, quantity and/or intensity of existence, use, manufacture, disposal, transportation, spill, release or effect, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment, or the Premises; (ii) regulated or monitored by any governmental authority; or (iii) a basis for potential liability of Landlord to any governmental agency or third party under any applicable statute. Hazardous Substance shall include, but not be limited to, hydrocarbons, petroleum, gasoline, crude oil or any products or by-products thereof. "**Reportable Use**" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and (iii) the presence in, on or about the Premises of a Hazardous Substance with respect to which any Applicable HS Requirements (as defined in subparagraph (F) hereinafter) require that a notice be given to persons entering or occupying the Premises or neighboring properties.

B. Tenant's Use of Hazardous Substances:

(1) Notice of Use of Hazardous Substances. Tenant may, without Landlord's prior consent, but upon notice to Landlord and in compliance with all Applicable HS Requirements and all other provisions of this Section 4.7, at Tenant's sole cost and expense, (i) operate a business on the Premises which is substantially similar to the business it is operating at its facilities in Livermore, California as of the Commencement Date, i.e. testing of solid state electronic components and research and development related thereto ("Permitted Use") and (ii) use any ordinary and customary Hazardous Substances reasonably required to be used by Tenant in the normal course of the Permitted Use.

(2) <u>Tenant's HS Use</u>. Tenant shall have the right to use the Hazardous Substances listed on Exhibit F without Landlord's prior consent and without the requirement of additional insurance. Tenant shall use all such Hazardous Substances in accordance with all Applicable HS Requirements and in compliance with all other provisions of this Section 4.7, specifically including the notice requirements and restrictions set forth below. Tenant's use of the substances referenced in Exhibit F may be referred to herein as "**Tenant's HS Use**".

(3) <u>Control of HS Hazards.</u>

(a) <u>Plans for Designated HS Areas</u>. Tenant shall use, store, or otherwise manage HS only in areas designated by Tenant for such use ("**Designated HS Areas**"). Prior to commencement of Tenant's HS Use on the Premises, and prior to modification of or addition to any Designated HS Areas, Tenant shall provide Landlord with written plans (such as architectural or engineering plans) regarding the design and planned operation of the Designated HS Areas. The plans shall include descriptions of the types and quantities of HS that will be used, stored, or otherwise managed in Designated HS Areas, the maximum design capacity of each Designated HS Area and descriptions of all equipment and structures that will be used to control environmental, health, and safety hazards associated with the HS, including, for example, secondary containment structures and air pollution control equipment. Tenant will also provide copies of all permits and other approvals required to be obtained to lawfully operate Tenant's business and Hazardous Substances on the Premises.

(b) <u>Commencement of Tenant's HS Use</u>. Tenant shall not commence Tenant's HS Use until Landlord has approved the plans submitted by Tenant pursuant to subparagraph (a) above, which approval shall not be unreasonably withheld or delayed. Landlord shall not withhold its approval, however, if Tenant's plans comply with the requirements of subparagraph (a) and the minimum standards set forth on Exhibit F-1. Landlord may (but without any obligation to do so) condition its consent upon Tenant's taking

3

such measures as Landlord, at its reasonable discretion, deems necessary to protect itself, the public, the Premises, the Center, and the environment against damage, contamination, injury, and/or liability, including, but not limited to the installation (and, at Landlord's option, removal on or before Lease expiration or earlier termination) of reasonably necessary protective equipment, structures, or modifications to the Premises.

(c) <u>Modification/Expansion of Designated HS Areas</u>. Tenant shall not modify or add to the Designated HS Areas until Landlord has approved the plans submitted by Tenant pursuant to subparagraph (a) above for the modification or addition, which approval shall not be unreasonably withheld or delayed. Landlord shall not withhold its approval, however, if Tenant's plans for the modification or addition comply with the requirements of subparagraph (a) and the minimum standards set forth on Exhibit F-1.

(4) <u>Notice of HS Use</u>. Tenant shall notify the Landlord in writing at least five (5) business days prior to any of the following:

(a) the date Tenant first commences Tenant's HS Use on the Premises; or

(b) the date Tenant commences to store or use any Hazardous Substance which is not listed on Exhibit F (a "**New HS** "), if the quantity of the New Hazardous Substance exceeds either (i) 55 gallons of liquid, 500 pounds of solid, 200 cubic feet of compressed gas at standard temperature and pressure, or (ii) the applicable Threshold Planning Quantity listed in 40 CFR Part 355.

After receipt of a notice pursuant to subparagraph (b) above, Landlord may require Tenant to obtain a policy of pollution liability insurance in a commercially reasonable form and amounts and with such insurer as may be reasonably approved by Landlord. For any insurance policy requirement, Landlord shall be named as an additional insured under such policy. Tenant shall deliver a certificate of any insurance required prior to bringing the Hazardous Substance into the Premises and Tenant shall maintain such insurance in effect until the closure requirements set forth in subparagraph (H) below have been satisfied or the New HS use ceases.

(5) <u>Contents of New HS Notice</u>. Each notice of a New HS shall specify the names and quantities of any New HS that Tenant intends to place on the Premises which exceeds the quantities described in subparagraph 4(b) above together with a copy of all permits and other approvals required to be obtained to lawfully use, store, or otherwise manage the New HS on the Premises. Tenant's notice shall also provide Landlord with information regarding the Designated HS Areas where the New HS will be used, stored, or otherwise managed, the new aggregate quantities of all Hazardous Substances in Designated HS Areas, and the maximum design capacities of the Designated HS Areas (if changed or modified from the Designated HS Areas as initially approved consistent pursuant to Section 4.7(B)(3)(b)above.

(6) <u>Increase in HS Quantities</u>. If, at any time during the term, Tenant intends to increase the quantity of existing Hazardous Substances and/or add New HS such that the aggregate quantity of all Hazardous Substances in any Designated HS Area on the Premises exceeds the maximum design capacity for the Designated HS Area, Tenant shall not increase quantities or add New HS until Landlord has consented to the modification of or addition to the Designated HS Areas, pursuant to Section 4.7(B)(3)(c) above.

(7) <u>Restrictions on Quantity or Use of HS</u>. Notwithstanding any other provision of this Lease, Tenant's use of Hazardous Substances at the Premises is subject to the following restrictions:

(a)	Tenant shall not use any HS in quantities such that Tenant would be subject to requirements for preparation of a Risk Management Plan, as set forth in 40 CFR Part 68.
(b)	Tenant shall not use any HS which emits odors unless the odors can be controlled to the extent they are not present at objectionable levels in any areas exterior to the Premises that are accessible to other tenants of the Center or the general public. In the absence of any legal thresholds for identifying objectionable odors, other odor standards may be used, provided they are generally accepted as being scientifically valid.
(c)	Tenant shall not use any HS in a manner that would result in "Significant Emissions". Significant Emissions are defined as air emissions originating from the Premises for which under applicable federal or state law (i) notices or warnings be given to other occupants of the Center or the general public or (ii) other occupants of the Center or the general public must receive special training

4

and/or use personal protective equipment.

C. **Plans/Reports**: Within ten (10) days after Tenant submits the same to any governmental authority, Tenant shall provide Landlord with copies of all hazardous materials business plans, permits and all other plans, reports and correspondence pertaining to storage/management of Hazardous Substances at the Premises, except waste manifests and routine monitoring reports.

D. **Duty to Inform Landlord:** If Tenant knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises or Building or Center, other than as previously permitted or consented to by Landlord or there has been a spill, release or discharge of any Hazardous Substances in the Premises (other than discharges permitted, authorized or otherwise approved by the applicable governmental agencies regulating the same), Tenant shall immediately give Landlord written notice thereof, together with a copy of any statement, report, notice, registration, application, permit, business plan, license, claim, action, or proceeding given to, or received from, any governmental authority or third party concerning the presence, spill, release, discharge of, or exposure to, such Hazardous Substance including but not limited to all such documents as may be involved in any Reportable Use involving the Premises. Tenant shall not cause or permit any Hazardous Substance to be spilled or released in, on, under or about the Premises (including, without limitation, through the plumbing, storm, or sanitary sewer system).

E. **Indemnification:** Tenant shall indemnify, protect, defend and hold Landlord, its agents, employees, lenders, and the Premises and Center, harmless from and against any and all damages, liabilities, judgments, costs, claims, liens, expenses, penalties, loss of permits and attorneys' and consultants' fees arising out of or involving any Hazardous Substance to the extent brought into the Premises and/or Center by or for Tenant, its employees, agents or contractors. Tenant's obligations under this Paragraph 4.7(E) shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Tenant, and the cost of investigation (including reasonable consultants' and attorneys' fees and testing), removal, remediation, restoration and/or abatement thereof, or of any contamination therein involved, and shall survive the expiration or earlier termination of this Lease. No termination, cancellation or release agreement entered into by Landlord and Tenant shall release Tenant from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Landlord in writing at the time of such agreement.

Tenant's Compliance with Requirements: Tenant shall, at Tenant's sole cost and expense fully, diligently and in a timely manner, comply with all "Legal Requirements", which term is used in this Lease to mean all laws, rules, regulations, ordinances, directives, covenants, easements and restrictions of record, permits, the requirements of any applicable fire insurance underwriter or rating bureau, relating in any manner to the Premises or Center (including but not limited to matters pertaining to (i) industrial hygiene, (ii) environmental conditions on, in, under or about the Premises, including soil and groundwater conditions, and (iii) the use, generation, manufacture, production, installation, maintenance, removal, transportation, storage, spill or release of any Hazardous Substance, which foregoing (ii) and (iii) Legal Requirements may be referred to as "Applicable HS Requirements"), now in effect or which may hereafter come into effect. Tenant shall, within twenty (20) business days after receipt of Landlord's written request made from time to time, provide Landlord with copies of all documents and information, including but not limited to permits, registrations, manifests, applications, reports and certificates, evidencing Tenant's compliance with all Applicable HS Requirements specified by Landlord, and shall within five (5) business days after receipt, notify Landlord in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving failure by Tenant or the Premises to comply with any Legal Requirements. Tenant shall be obligated to disclose to Landlord which Hazardous Substances are used at the Premises and how such Hazardous Substances are being handled (but in no event shall Tenant be required to disclose information regarding formulations or manufacturing processes or procedures related to such Hazardous Substances) notwithstanding that such information may be proprietary information or a trade secret. Landlord agrees to keep as confidential all such proprietary information delivered to Landlord (including, without limitation, Exhibit F) and which Tenant designates in writing as confidential, provided that Landlord may disclose the same when required by law or in litigation between Landlord and Tenant regarding such information or to Landlord's lenders or to prospective purchasers provided such parties have also agreed to keep the same confidential.

G. **Compliance with Law Governing Hazardous Substances:** Landlord, Landlord's agents, employees, contractors and designated representatives, and the holders of any mortgages, deeds of trust or ground leases on the Premises ("**Lenders**") shall have the right to enter the Premises at any time in case of an emergency, and otherwise at reasonable times (but not more often than annually for inspection of Tenant's "clean room" on the Premises, if any, or more often than quarterly for inspection of other parts of the Premises), and upon no less than 10 days' notice, unless an emergency exists, for the purpose of inspecting the condition of the Premises and for verifying compliance by Tenant with this Lease and all Legal Requirements, and Landlord shall be entitled to employ experts and/or consultants in connection therewith (provided that such experts and/or consultants are not engaged in a business competitive with Tenant, or consult or give advice to

any competitor of Tenant listed on Exhibit I) to advise Landlord with respect to Tenant's activities, including but not limited to Tenant's installation, operation, use, monitoring, maintenance, or removal of any Hazardous Substance on or from the Premises ("Landlord's Consultants"). Prior to engaging any Landlord's Consultants, Landlord shall provide Tenant with written notice of the name of the proposed consultant and Tenant shall have five (5) business days to object to the engagement based upon Tenant's reasonable belief that engagement of the particular individual as Landlord's Consultant, and the consequent access to Tenant's facilities and proprietary information and trade secrets, could result in competitive injury to Tenant. Landlord shall not engage with a consultant as to whom Tenant has objected. Tenant shall cooperate with Landlord's Consultants inspecting the Premises, including responding to interviews (for a time period not to exceed four (4) hours for the initial site visit and two (2) hours for site visits thereafter). Landlord's Consultants shall at all times be escorted by Tenant, unless Tenant agrees otherwise. This and all rights to enter except in the event of an emergency, are subject to Landlord, Landlord's agents, employees, contractors, designated representatives, prospective purchasers and/or Lenders, as the case may be, executing Tenant's standard non-disclosure agreement in the form attached hereto as Exhibit H. The costs and expenses of any such inspections shall be paid by the party requesting same and in no event shall be borne by or passed along to Tenant unless requested by Tenant, subject only to the proceeding sentence. If the inspection is performed due to a violation of Applicable HS Requirements, Tenant shall, upon request, reimburse Landlord or Landlord's Lender, as the case may be, as additional rent, for the costs and expenses of such inspections.

H. **Closure Requirements:** Prior to any termination of the Lease, Tenant, at its sole cost and expense (except as to those costs and expenses arising out of actions undertaken by Landlord or by a third party on behalf of Landlord), shall satisfy the following closure requirements with respect to the Hazardous Substances Tenant has used in the Premises during the term:

(1) Comply with all applicable federal, state and local closure requirements with respect to Hazardous Substances;

(2) Prepare a closure plan (the "**Closure Plan**") that specifies the final disposition of all Hazardous Substances and equipment which may be contaminated with Hazardous Substances; cleaning and decontamination activities, and confirmation sampling (e.g. wipe samples, soil/ground water samples and/or indoor air quality samples, to the extent warranted by the site conditions then existing).

(3) At least sixty (60) days prior to the Lease termination, provide to Landlord a copy of the Closure Plan for review and reasonable approval. Landlord may, after consultation with Tenant, require modification of the closure plan to include additional activities, including sampling activities, if the site conditions indicate that there is a reasonable probability that "Significant Residual Contamination" is present. **Significant Residual Contamination** shall mean residual contamination which: (i) exceeds standards or guidance levels typically used by regulatory agencies in California for evaluating potential threats to human health or the environment; or (ii) would result in notification requirements under applicable state law of potential health risks to individuals on the Premises, other tenants of the Center, and/or the general public; or (iii) would result in potential environmental liability to Tenant or Landlord; or (iv) would result in the need for conducting any type of additional decontamination activities prior to leasing the Premises to a new tenant. If Landlord fails to request modification of the Closure Plan within ten (10) business days after its receipt thereof, Tenant's Closure Plan shall be deemed accepted.

(4) Notify Landlord of closure schedule and allow access to Landlord and/or Landlord's Consultants for inspections prior to commencing and following completion of the cleaning/decontamination activities.

(5) Notify Landlord of all sample analysis results, if any. Landlord may require additional closure activities if sampling results disclose Significant Residual Contamination.

(6) Prepare and provide to Landlord closure report documenting closure activities consistent with the Closure Plan and sample results, if any, following completion of all closure activities.

Closure shall be deemed to be complete upon Landlord's reasonable approval of the closure report and, if applicable, Landlord's receipt of a copy of the written closure approval from the local environmental agency with jurisdiction over the Hazardous Substances at the Premises.

Survival of Obligations: Tenant's obligations under this Section 4.7 shall survive the termination of this Lease.

I.

See Addendum A-4.7.

6

4.8 Declaration. Tenant acknowledges and agrees that this Lease shall be subject to and subordinate to a recorded Declaration of Covenants, Conditions and Restrictions which, together with all amendments from time to time, are collectively referred to as the "**Declaration**". A true and correct copy of the Declaration is attached hereto as <u>Exhibit G</u>. Tenant agrees to bound by and comply with all provisions of the Declaration. See Addendum A-4.8.

ARTICLE 5. INTENTIONALLY DELETED

ARTICLE 6. UTILITIES

6.1 Tenant, at its own cost and expense, shall pay for all water, gas, heat, electricity, garbage disposal, sewer charges, telephone, and any other utility or service charge related to its occupancy of the Premises, including but not limited to any hook- up charges. Utilities will be separately metered to the Premises, except for water as to which Landlord has elected to have a single meter for the entire Building 6. As a part of Tenant's initial improvements to the Premises, Tenant shall install a sub-meter for water at its sole cost and expense. Tenant shall pay water charges for Building 6 based on the sub-meter reading. Tenant shall not be obligated to pay for any other water charges for Building 6, except that Tenant shall also pay its Proportionate Share of all water use and other water charges with respect to the landscaping and fire protection service on Parcel 6. Tenant shall also pay Tenant's Proportionate Share of the

5

electricity for all outdoor lighting on Parcel 6. Tenant shall not be responsible for any water or electrical service costs or expenses with respect to any other parcels in the Center.

6.2 Except to the extent arising out of Landlord's negligence or willful misconduct, Landlord shall not be liable in damages, consequential or otherwise, nor shall there be any rent abatement, arising out of any interruption or reduction whatsoever in utility services (i) which is due to fire, accident, strike, governmental authority, acts of God, acts of other tenants or other third parties, or other causes beyond the reasonable control of Landlord or any temporary interruption in such service, and (ii) which is necessary to the making of alterations, repairs, or improvements to the Center, or any part of it (all of which shall be conducted pursuant to <u>Article 9</u>), or (iii) to comply with energy conservation measures mandated by a governmental agency having jurisdiction over the Center.

See Addendum A-6.

ARTICLE 7. REAL PROPERTY TAXES

7.1 Commencing on the Commencement Date, Tenant shall pay as Additional Rent Tenant's Proportionate Share of all "**Taxes**" (as hereinafter defined) which may be levied, assessed or imposed against or become a lien upon Parcel 6, the tax parcel upon which Building 6 is located, which will be separately assessed. The term "**Taxes**" shall mean and include real estate taxes, assessments (special or otherwise), including impositions for the purpose of funding special assessment districts, water and sewer rents, rates and charges (including water and sewer charges which are measured by the consumption of the actual user of the item or service for which the charge is made) levies, fees (including license fees) and all other taxes, governmental levies and charges of every kind and nature whatsoever (and whether or not the same presently exist or shall be enacted in the future) which may during the term be levied, assessed, imposed, become a lien upon or due and payable with respect to, out of or for the Parcel 6 or any part thereof, or of any land, building or improvements thereon, or the use, occupancy or possession thereof; and imposed or based upon or measured by the rents receivable by Landlord for the Parcel 6, including gross receipts taxes, business taxes, business and occupation taxes.

"Taxes" shall also include interest on installment payments and all costs and fees (including reasonable attorney's and appraiser's fees) incurred by Landlord in contesting Taxes and negotiating with public authorities as to the same. Taxes shall not include, however, any franchise, estate, inheritance, corporation, transfer, net income, excess profits tax or any assessments levied by the Association pursuant to the Declaration. Association assessments shall be payable pursuant to the provisions of Section 10.4.

7.2 Tenant shall pay Tenant's Proportionate Share of Taxes with respect to any tax fiscal year during the term hereof. Landlord's estimate of Tenant's initial tax payment for Parcel 6 is that amount set forth in Paragraph 1(e) above.

7.3 Commencing on the first day of the month following the month in which the Lease Commencement Date occurs, Tenant shall pay Landlord monthly, on the first day of each month, the amount computed in accordance with <u>Paragraph 1(e)</u> above as an impound toward the Taxes. If the Commencement Date does not occur the first day of a month, Tenant's first monthly payment shall include the amount payable for the partial month at the beginning of the term. Tenant's actual obligation for Taxes shall be determined and computed by Landlord not less often than annually and at the time each such computation is made, Landlord and Tenant shall adjust for any difference between impounded amounts and Tenant's actual share. Tenant shall pay Landlord any deficiency (or Landlord shall pay Tenant any surplus) within thirty (30) days after receipt of Landlord's written statement. At the time of each such

7

computation, Landlord may revise the monthly payment for Taxes set forth in <u>Paragraph 1(e)</u> above by written notification to Tenant. Tenant shall pay its share of Taxes during each year of the Lease Term. Landlord shall furnish Tenant with a copy of the tax bills for the Parcel 6 supporting the amounts charged to Tenant by Landlord.

7.4 If this Lease shall commence or terminate on any date other than the last day of a tax fiscal year, the amount payable by Tenant during the applicable tax fiscal year shall be prorated on the basis which the number of days of the Term within said tax fiscal year bears to 365. The obligation of Tenant under this <u>Article 7</u> shall survive the termination of this Lease.

ARTICLE 8. CONSTRUCTION AND ACCEPTANCE

8.1 Landlord at its sole cost and expense shall construct "**Landlord's Work**" as described in <u>Exhibit C</u> attached hereto and incorporated by reference herein. Tenant shall construct "**Tenant's Work**" as specified in <u>Exhibit C</u> and Landlord shall provide a Tenant Improvement Allowance in the amount specified in <u>Paragraph 1 (i)</u> to be applied to the cost of the Tenant Improvements constructed by Tenant. If the actual cost of such Tenant Improvement Allowance, all excess costs shall be at Tenant's sole cost and expense. If the cost is less than the Tenant Improvement Allowance, the balance of the Tenant Improvement Allowance shall be applied to the cost of any Special Tenant Improvements described in <u>Exhibit C</u>, or if none are specified, to the cost of Tenant Improvements under any then existing lease between Landlord and Tenant for other premises in the Center. Landlord agrees to notify Tenant at least ten (10) days prior to the date Landlord anticipates substantial completion of the Base Building portion of Landlord's Work as set forth in <u>Exhibit C</u> ("Base Building Work"). The "**Delivery Date**" for the Premises shall be the date upon which Landlord has substantially completed the Base Building Work, as evidenced by a written certificate of substantial completion issued by Landlord's architect. The remaining Landlord's Work shall be substantially completed on or before the Rent Commencement Date. As used herein, "substantial completion" shall mean completed, except for minor punch list items which do not interfere with Tenant's ability to complete its improvements. The condition of the Premises in compliance with the requirements set forth above may sometimes be referred to herein as the "**Delivery Condition**."

8.2 Following delivery of the Premises to Tenant in the Delivery Condition, Tenant shall diligently proceed to complete Tenant's Work, including any Special Tenant Improvements and such other work as it may deem necessary for the conduct of its business in the Premises. Prior to commencing Tenant's Work, Tenant shall submit to Landlord for approval plans and specifications prepared by an architect selected by Tenant, which plans shall be subject to Landlord's prior reasonable approval. Once Tenant's plans are approved by Landlord, Tenant's contractors (which shall also be subject to prior reasonable approval by Landlord) shall obtain all necessary permits for the work set forth in the Approved Tenant Plans (the "Tenant 's Work") and proceed to complete Tenant's Work in compliance with all applicable governmental requirements.

8.3 Within thirty (30) days following Delivery Date and within thirty (30) days following the date of substantial completion of Landlord's Work, Landlord and Tenant shall mutually prepare a punch list of items to be corrected in the Base Building Work and other Landlord's Work, respectively, including any defects or non-conformance in Landlord's construction. Landlord shall cause its contractors to promptly complete all punch list items. Landlord's Work shall also be under warranty by Landlord's contractors for a period of one (1) year. Landlord hereby assigns to Tenant all warranties and guaranties received by Landlord from its contractors with respect to Landlord's Work. If Landlord's contractors shall fail to complete any punch list items within the 90-day period following completion of the applicable punchlist, and such failure continues after notice from Tenant and the cure period provided in Article 24, Tenant may at its option (but shall not be obligated to) complete the required work at Landlord's cost. Landlord shall pay to Tenant within thirty (30) days the amount shown on any statement describing the necessary work completed by Tenant accompanied by the invoices for such work.

8.4 Landlord's Work for Building 6 includes construction of improvements required for nitrogen truck access to the Premises. Landlord will use diligent efforts to complete said truck access on or before November 15, 2004, subject to delays in the City approval process or Force Majeure delays as described in Section 32.8 below. If Landlord has not completed the construction of the required improvements by November 15, 2004, Landlord, at its sole cost and expense, agrees to provide temporary nitrogen truck access to the Premises on or before November 15, 2004 to ensure access until the permanent improvements have been constructed.

8.5 After the Premises has been constructed, Landlord's architect shall measure the gross leasable area of the Premises, and shall certify to Landlord such measurement in writing. The GLA so certified will be deemed to be the GLA of the Premises for all purposes of this Lease. To compute the Premises GLA, Building 6 shall be measured to the drip line as shown on <u>Exhibit A-1</u>. The initial monthly Base Rent, estimated tax and operating expense payments, Tenant's Proportionate Share, and the Tenant

8

Improvement Allowance were based on the estimated GLA set forth in Paragraph A of the Recitals at the beginning of this Lease. In the event that the Premises GLA as determined pursuant this <u>Section 8.5</u> is different from the estimated GLA (which difference shall be certified by Landlord's architect and approved by Tenant), the Base Rent, estimated payments, Tenant's Proportionate Share, and Tenant Improvement Allowance amounts set forth in <u>Article 1</u> shall be adjusted accordingly.

8.6 In the event that Landlord, at its sole option, permits Tenant to take possession of the Premises prior to the Commencement Date for the purpose of constructing its Tenant Improvements, such possession shall be on all the terms and conditions of this Lease except for payment of Rent, specifically including the insurance and indemnity provisions in <u>Articles 14 and 16</u>. In the event that Landlord notifies Tenant that Tenant's early possession is causing a delay in Landlord's Work, Tenant shall promptly cease its construction activities and cause its contractor to remove its personnel, subcontractors and equipment from Premises until the Delivery Date or earlier date acceptable to Landlord.

ARTICLE 9. REPAIRS AND MAINTENANCE

9.1 Landlord, at its sole cost and expense, shall be responsible for the repair, maintenance and, if necessary, replacement of the structural elements, the roof structure, foundation and the structural integrity of floor slabs of Building 6, provided that Tenant shall pay for the cost of any such repairs to the extent occasioned by the negligent act, omission or willful misconduct of Tenant, its agents, employees, invitees, licensees or contractors, or by the construction of Tenant Improvements by Tenant, but only to the extent such cost is in excess of any proceeds received by Landlord from the insurance for Building 6 maintained by Landlord pursuant to Section 14.2.

9.2 Subject to reimbursement by Tenant as provided in <u>Article 10</u> hereof, Landlord shall keep and maintain in good repair (including replacement as necessary), the roof covering and the exterior surfaces of the exterior walls and window frames of Building 6 (exclusive of doors, door frames, door checks and other entrances and windows), all Outdoor Areas (defined in Section 10.1) on Parcel 6, all Shared Areas (as defined in the Declaration) for the use of Parcel 6 and all systems (including sewer, gas, electrical and water lines) serving the Premises to the point of connection to Building 6. Tenant shall give Landlord prompt written notice of any damage to the Premises requiring repair by Landlord.

9.3 Except to the extent of Landlord's obligations provided in Sections 9.1 and 9.2 hereof, Tenant shall, at its expense, keep and maintain the Premises and every part thereof in good order, condition and repair, including, without limiting the generality of the foregoing, all equipment or facilities specifically serving the Premises, such as plumbing, heating, air conditioning, ventilating, electrical, lighting facilities, boilers, fired or unfired pressure vessels, fire hose connections if within the Premises, fixtures, interior walls, interior surfaces of exterior walls, ceilings, floors, windows, doors, plate glass, and skylights. Notwithstanding the foregoing, Tenant shall not be required to make any such repairs to the extent occasioned by the negligent act or omission or willful misconduct of Landlord, its agents, employees, or contractors. Tenant shall keep its sewers and drains open and clear to the perimeter of the Premises, and shall keep the hallways and/or sidewalks and common areas adjacent to the Premises clean and free of debris created by Tenant. Tenant shall reimburse Landlord on demand for the cost of damage to the Premises, Building 6 or Landlord's Parcels caused by Tenant or its employees, agents, customers, suppliers, shippers, contractors, or invitees which is in excess of any proceeds received by Landlord from the insurance for Building 6 maintained by Landlord pursuant to Section 14.2. If Tenant shall fail to comply with the foregoing requirements within ten (10) days after notice from Landlord, Landlord may (but shall not be obligated to) effect such maintenance and repair, and the cost thereof together with interest thereon at the Interest Rate (defined below) shall be due and payable as Additional Rent to Landlord within thirty (30) days following receipt of Landlord's written statement of such costs.

See Addendum A-9.3

9.4 Tenant in keeping the Premises in good order, condition, and repair shall exercise and perform good maintenance practices including obtaining, at its expense, a contract for the repair and maintenance of the air conditioning and heating system, if any, exclusively serving the Premises and provide Landlord with a copy of said contract within thirty (30) days after Tenant takes possession of the Premises. The contract shall be for the benefit of Landlord and Tenant and in a form and placed with a licensed contractor satisfactory to Landlord. Tenant obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements thereon or a part thereof in good order, condition and state of repair, except the extent of Landlord's obligations expressly set forth in this Lease.

9.5 Tenant shall not make any exterior or structural alterations, changes or improvements in or to the Premises or material modifications to any of the Base Building operating systems within the

Premises without first obtaining Landlord's prior written consent (which may be withheld by Landlord in its sole discretion as to exterior alterations, and which shall not be unreasonably withheld or delayed with respect to structural or Base Building system modifications), and all of the same shall be at Tenant's sole cost. Landlord's consent shall not be required for any interior cosmetic alterations or alterations not affecting Base Building exterior, structure or systems as referenced above, or for any alterations, changes, replacements or improvements to any interior nonstructural Special Tenant Improvements or any other elements of Tenant's Work; provided that Tenant shall obtain required permits and comply with all other Legal Requirements and all requirements of Article 8 and Exhibit C regarding construction by Tenant and shall notify Landlord not less than ten (10) days prior to commencing any such alterations to give Landlord an opportunity to post a notice of non-responsibility. Landlord may impose as a condition of its consent (when required) such requirements as Landlord, in its reasonable discretion, may deem necessary, including but not limited to, the requirement that Tenant utilize for such purposes only contractors, materials, mechanics and materialmen approved by Landlord, and that good and sufficient plans and specifications be submitted to Landlord at such times as its consent is requested. Further, Landlord's consent to any alteration which Tenant proposes to make after the Commencement Date shall designate by written notice to Tenant any of the alterations, additions and improvements (collectively, "Alterations") which Landlord will require Tenant to remove at the expiration or termination of the Lease and those Alterations (if any) which Tenant is not permitted to remove. If Landlord so designates, Tenant shall prior to the expiration of the Term promptly remove the Alterations designated to be removed and repair all damage caused by such removal at its cost and with all due diligence, and shall surrender the Premises with all Alterations which Tenant is required to leave. Unless Landlord designates as a condition to granting its consent to any Alterations that removal by Tenant is required or prohibited, Tenant shall have the right, but not the obligation to remove from the Premises the Alterations for which consent was obtained so long as Tenant promptly repairs any damage resulting from such removal. Except as otherwise expressly provided herein, all Alterations made by Tenant (specifically excluding Tenant's furniture, trade fixtures and equipment) shall become the property of Landlord and a part of the realty and shall be surrendered to Landlord upon the expiration or sooner termination of the Term hereof.

See Addendum A-9.5.

ARTICLE 10. OPERATING AND MAINTENANCE COSTS

10.1 All Common Areas in the Center shall be operated and maintained by the Association pursuant to the Declaration. The term "Common Areas" as used in this Lease shall include all areas in the Center defined as Common Areas in the Declaration. Landlord agrees to operate and maintain or cause to be operated and maintained during the term of this Lease all "Outdoor Areas" on Parcel 6. The term "Outdoor Areas" as used in this Lease shall include all areas on each of Landlord's Parcels which are not Common Areas, or areas covered by buildings ("Building Areas") and are provided by Landlord for the convenience and exclusive use of tenants of each of Landlord's Parcels, their respective employees, customers, suppliers, shippers, contractors, and invitees.

10.2 The manner and method of operation, maintenance, service and repair of the Common Areas and the expenditures therefor, shall be determined in accordance with the provisions of the Declaration. The manner and method of operation, maintenance, service and repair of the Outdoor Areas shall be determined by Landlord and at minimum shall be comparable to similar projects in the general vicinity of the Center and shall be in accordance with all Legal Requirements. Except as otherwise expressly provided herein, Landlord reserves the right from time to time to make changes in, additions to and deletions from the Outdoor Areas and/or Common Areas including without limitation changes in the location, size, shape and number of driveways, entrances, parking spaces, parking areas, loading and unloading areas, ingress, egress, direction of traffic, landscaped areas, walkways and utility raceways and the purposes to which they are devoted. Notwithstanding the foregoing, in no event shall Landlord make or permit any modifications to Landlord's Parcels which materially and adversely affect Tenant's access to or from Parcel 6 as shown on Exhibit A or which would reduce the number of exclusive parking spaces on Parcel 6 available to Tenant, its agents, employees or contractors.

10.3 Tenant agrees to comply with such reasonable rules and regulations as the Association may adopt from time to time for the orderly and proper operation of the Common Areas. Tenant further agrees to comply with and observe all reasonable rules and regulations established by Landlord from time to time for use of the Outdoor Areas on Parcel 6, including, without limitation, the removal, storage and disposal of refuse and rubbish. The initial Rules and Regulations for the Center are attached hereto as <u>Exhibit E</u>. All rules and regulations adopted or amended after the date of this Lease shall be reasonable and non-discriminatory and shall be subject to the restrictions set forth in Section A-4.9 of the Addendum.

10.4 Commencing with the Commencement Date, during the Term of this Lease, Tenant shall pay to Landlord, as Additional Rent, at the time and in the manner specified in Section 10.6 below, Tenant's Proportionate Share of all costs and expenses of every kind and nature paid or incurred by Association and/ or Landlord in operating, policing, protecting, lighting, providing sanitation and sewer and other services to, insuring, repairing, replacing and maintaining in neat, clean, good order and

condition, the Common Areas of the Center and all Outdoor Areas on Landlord's Parcels and in operating, insuring and maintaining the Buildings on Landlord's Parcels ("**operating and maintenance costs**") that are allocable to Building 6 as provided in Section 10.5 below.

Subject to the exclusions set forth below, operating and maintenance costs shall include, but shall not be limited to, the following: water, gas and electricity to the Common Areas and Outdoor Areas, and security and guard services; salaries and wages (including employment taxes and so called "fringe benefits") or maintenance contracts of all persons and management personnel to the extent engaged in the regular operation, servicing, repair and maintenance, (specifically including the site coordinator and site superintendent, clerical, and on-site and off-site accounting staff), repair and replacement of roofs of Buildings on Landlord's Parcels, painting and cleaning the exterior surfaces of such Buildings, premiums for liability, property damage and Workers' Compensation insurance (which insurance Landlord, at all times during the Lease term, agrees to maintain with respect to Landlord's Parcels); all costs associated with obtaining such insurance or making any claims under such insurance policies, including the cost of any deductible portion payable with respect to claims (subject to subparagraphs (x) and (xxv)); personal property taxes, if any; charges, excises, surcharges, fees or assessments levied by a governmental agency by virtue of the parking facilities furnished; costs and expenses of planting, replanting and relandscaping; trash disposal, if any; lighting, including exterior building lights; utilities; maintenance and repair of utility lines, sewers and fire detection and suppression systems (including the water used in connection with such systems); sweeping, repairing and resurfacing the blacktop surfaces; repainting and restriping; exterior signs and any tenant directories for the Center as a whole, reserves set aside for maintenance and repair, the cost of any environmental inspections; fees for any licenses

and/or permits required for operation of the Common Areas and Outdoor Areas, or any part thereof; equipment rental or purchases, supplies, postage, telephone, service agreements, deliveries, promotion, dues and subscriptions, and reasonable legal fees.

The following costs shall be excluded from the operating and maintenance costs payable by Tenant:

- the costs of the initial construction of the Center, including the Buildings, roads, parking lots, utility lines and similar improvements shown on Exhibit A;
- debt service (including, without limitation, principal, interest, late fees, prepayment fees, principal, points, impound payments and all other charges) with respect to any financing relating to Landlord's acquisition or initial construction of the Center or any portion thereof or any refinancing of such costs;
- (iii) any fees or other amounts payable with respect to any ground lease now or hereafter affecting any portion of the Center;
- (iv) any costs, fines or penalties incurred as a result of any violation of laws, rules or regulations by Landlord, its agents, employees or contractors;
- (v) the cost of any items for which Landlord is reimbursed (or if Landlord fails to carry the insurance required by Section 14.2, would have been so reimbursed) by insurance proceeds, condemnation awards, other tenants of the Center, or for which Landlord is otherwise actually reimbursed;
- (vi) any real estate brokerage commissions or other costs (including, without limitation, finder's fees, legal fees, space planning fees and review and supervision fees) incurred in connection with the sale, leasing or subleasing of any portion of the Center, including the renewal, extension or modification of leases;
- (vii) any costs representing amounts paid to an entity or person which is an affiliate of Landlord which is in excess of the amount which would have been paid in the absence of the relationship, including, without limitation, any overhead or profit increment paid to subsidiaries or affiliates of Landlord for goods and/or services to any portion of the Center to the extent in excess of the amount which would be paid to unaffiliated third parties on a competitive basis;
- (viii) capital improvements and expenditures shall be amortized over the useful life of the capital item in accordance with GAAP.
- (ix) non-cash items, such as deductions for depreciation or obsolescence of any improvements or equipment within or used in connection with the Center, and reserves for future expenditures (except reserves maintained by the Association pursuant to the Declaration);
- (x) costs incurred by Landlord for the repair of damage to the Center caused by fire, windstorm, earthquake or other casualty, condemnation or eminent domain; provided that

11

an amount equal to the deductible under Landlord's insurance policy may be included, up to a maximum of \$5,000 for property damage and \$25,000 for liability insurance (collectively the "**Existing Deductibles**"), unless otherwise approved by Tenant, and specifically excluding any earthquake insurance deductible;

- (xi) Landlord's general corporate overhead and general administrative expenses (including memberships, travel, recruitment and marketing);
- (xii) any compensation or benefits paid to clerks or attendants for parking operations of the Center, including validated parking for any entity unless the revenues, if any, from such operations are used to reduce the operating and maintenance costs;
- (xiii) electric power, water or other utility costs for which any tenant or occupant of the Center directly contracts with the local public service company or for which any tenant is separately metered or submetered and pays Landlord directly;
- (xiv) penalties, late charges and interest incurred as a result of Landlord's failure or negligence to make payments and/or to file any returns (including tax or other informational returns) when due, unless due to Tenant's failure to timely pay the Rent hereunder;
- (xv) Landlord's charitable or political contributions, membership dues to organizations or expenses related to attendance at or travel to meetings of political, charitable or business organizations;
- (xvi) costs associated with the operation of the business of the corporation, partnership or other entity which constitutes Landlord as the same are distinguished from the costs of operation of the Center, including partnership accounting and legal matters, and costs of selling or mortgaging any of Landlord's interest in the Center;
- (xvii) any expenses for repairs or maintenance to the extent reimbursed through warranties, service contracts or recoveries from vendors;
- (xviii) any costs incurred in connection with the defense of Landlord's title to the Center or any portion thereof;
- (xix) fines and penalties incurred by Landlord due to the violation by Landlord or any tenant of the Center of the terms and conditions of any lease at the Center, or fines or penalties incurred by Landlord due to the violation by Landlord or any tenant of the Center of any law, code, regulation or ordinance;
- (xx) marketing, advertising and promotional expenditures ;
- (xxi) any bad debt or expense, rent loss or reserves for bad debt or rent loss;

- (xxii) any amounts constituting "Taxes" as defined in and to the extent payable pursuant to <u>Article 7</u> of this Lease;
- (xxiii) the costs of any building repairs, maintenance, replacement or casualty insurance for any buildings other than Building 6;
- (xxiv) any costs which would duplicate a cost included in the Association charges payable by Tenant with respect to Parcel 6;
- (xxv) any premiums for any policy of earthquake insurance with respect to the Center or any portion thereof or any deductible amount under such policies.

10.5 Tenant shall pay its share of the operating and maintenance costs described in Section 10.4 above as provided in this Section 10.5. Tenant's share of operating and maintenance costs for the Common Areas of the Center shall be Tenant's Proportionate Share of such costs allocated by the Association to Parcel 6 pursuant to the Declaration. Tenant's share of all other operating and maintenance costs shall be as follows: (i) costs related to repairs the Buildings on Landlord's Parcels shall be determined by the GLA of the Premises compared to the aggregate GLA of all Buildings on Landlord's Parcels, (ii) costs related to the Outdoor Areas on Landlord's Parcels shall be determined by the square footage of Parcel 6 compared to the total square footage of all of Landlord's Parcels, and then charging Tenant for Tenant's Proportionate Share thereof, and (iii) notwithstanding the foregoing, operating costs which benefit only one or a portion of all of Landlord's Parcels shall be allocated only among the Parcels benefited either by GLA or Parcel square footage, as applicable, and Tenant shall be charged only for

12

Tenant's Proportionate Share of that portion of the operating costs that is allocable to Building 6 or Parcel 6, as the case may be.

10.6 As Additional Rent, Tenant shall pay Landlord monthly on the first day of each month, following the Commencement Date and continuing on the first day of each month thereafter during the Term hereof, an operating and maintenance charge in an amount estimated by Landlord to be Tenant's share of the **"operating and maintenance costs**". If the Commencement Date does not occur the first day of a month, Tenant's first monthly payment shall include the amount payable for the partial month at the beginning of the term. The initial monthly operating and maintenance charge shall be the amount estimated by Landlord as set forth in <u>Paragraph 1(d)</u>. Landlord may adjust said monthly charge at the end of each calendar year thereafter on the basis of Landlord's reasonably anticipated costs for the following calendar year.

10.7 Within one hundred twenty (120) days after the end of each calendar year, Landlord shall furnish to Tenant a statement showing the total operating and maintenance costs, Tenant's share of such costs, and the total of the monthly payments made by Tenant to Landlord during the calendar year just ended. Landlord shall keep good and accurate books and records concerning the operation, maintenance and management of the Landlord's Parcels, and Tenant and its agents shall have the right, upon twenty (20) days' written notice given within nine (9) months after receipt of the statement for a calendar year, and at Tenant's sole cost and expense to audit, inspect and copy such books and records with respect to such calendar year at the office where the same are located. If such audit discloses that the annual statement has overstated the actual operating and maintenance expenses for the calendar year under review, Landlord shall rebate to Tenant the amount by which Tenant has been overcharged or, at Tenant's election, Tenant may offset such amount against operating and maintenance charges becoming due; and if the audit discloses that Landlord's annual statement has overstated such charges by more than five percent (5%), then, in addition to rebating to Tenant any overcharge, Landlord shall also pay the reasonable costs incurred by Tenant for such audit. If Landlord disputes the results of Tenant's audit, the parties shall submit the dispute for resolution by arbitration in accordance with the procedures set forth in Section 10.4 of the Declaration, which shall be deemed to be incorporated herein by this reference. The decision of the arbitrator shall be binding and conclusive on the parties.

10.8 If Tenant's share of the operating and maintenance costs for the accounting period exceeds the payments made by Tenant, Tenant shall pay Landlord the deficiency within ten (10) days after the receipt of Landlord's statement. If Tenant's payments made during the accounting period exceed Tenant's pro-rata share of the operating and maintenance costs, Tenant may deduct the amount of the excess from the estimated payments next due to Landlord. If a credit remains at the end of the Lease Term, such credit shall be refunded by Landlord to Tenant within twenty (20) business days thereafter. The obligations of Landlord and Tenant under this Section 10.8 shall survive the termination of this Lease.

ARTICLE 11. TRADE FIXTURES AND SURRENDER

11.1 Upon the expiration or sooner termination of the Term hereof, Tenant shall surrender the Premises including, without limitation, all apparatus and fixtures then upon the Premises, in good condition and repair, reasonable wear and tear excepted, broom clean and free of trash and rubbish, subject, however to the following:

a Tenant shall remove all Alterations which Landlord has designated to be removed pursuant to Section 9.5 above and shall leave all Alterations which Landlord has designated pursuant to that Section must remain;

b. If no consent was required or obtained, Tenant shall either remove or leave all Alterations which Landlord prior to the end of the Term designates in writing to Tenant must be removed or left in place;

c. Tenant at its election may remove or leave all Alterations with respect to which Landlord has not made a designation as described in (a) or (b) above.

d. Tenant shall remove all of Tenant's Personal Property (as defined in Section 11.3 below).

e. Tenant shall repair all damage caused by removal of its Personal Property and any Alterations Tenant is permitted to remove.

Notwithstanding anything to the contrary herein, Tenant Improvements and any Special Tenant Improvements shall be the property of Landlord throughout the Term to the extent of the amount of the Tenant Improvement Allowance, and such improvements may not be removed by Tenant without Landlord's prior written consent. To the extent the costs of Tenant Improvements and/or Special Tenant Improvements exceed the Tenant Improvement Allowance, such improvements shall be owned by Tenant

throughout the Term. At the end of the Term, all Tenant Improvements and Special Tenant Improvements which Tenant is not required to remove in accordance with the terms hereof shall be surrendered by Tenant without any injury, damage or disturbance thereto, and Tenant shall not be entitled to any payment therefor.

11.2 Within ninety (90) days prior to its Lease expiration, Tenant shall notify Landlord in writing of the manner and means in which it will remove any and all Hazardous Substances used in the Premises during its occupancy. Tenant shall also certify in writing upon delivery of Premises to Landlord on the date of the Lease expiration that all Hazardous Substances were removed in accordance with all governmental and regulatory laws.

11.3 Moveable trade fixtures, furniture and other personal property (collectively, Tenant's "**Personal Property**") installed in the Premises by Tenant at its cost shall be Tenant's property unless otherwise provided in Section 11.1 above and Tenant shall remove all of the same prior to the termination of this Lease and at its own cost repair any damage to the Premises and Parcel 6 caused by such removal. If Tenant fails to remove any of such property, Landlord may at its option retain such property as abandoned by Tenant and title thereto shall thereupon vest in Landlord, or Landlord may remove the same and dispose of it in any manner and Tenant shall, upon demand, pay Landlord the actual expense of such removal and disposition plus the cost of repair of any and all damage to said Premises and the building thereto resulting from or caused by such removal.

ARTICLE 12. DAMAGE OR DESTRUCTION

12.1 Except as otherwise provided in Section 12.2 below, if the Premises are damaged and destroyed by any casualty covered by fire and special extended coverage insurance policies which Landlord is required to provide pursuant to <u>Article 14</u>, Landlord shall repair such damage as soon as reasonably possible, to the extent of the available proceeds, and the Lease shall continue in full force and effect.

12.2 If the Premises are damaged or destroyed by any casualty covered by Landlord's fire and special extended coverage insurance policies which Landlord is required to provide pursuant to <u>Article 14</u>, to the extent of seventy-five percent (75%) or more of the replacement cost thereof, or to the extent of twenty-five percent (25%) or more of the replacement cost of the Premises if the damage occurs during the last twelve (12) months of the Term, or if the insurance proceeds which are received by Landlord, under the policies Landlord is required to provide, are not sufficient to repair the damage (specifically including any insufficiency due to payment of such proceeds to Landlord's lender, if required), then Landlord may, at Landlord's option, either (i) repair such damage as soon as reasonably possible, in which event this Lease shall continue in full force and effect, or (ii) cancel and terminate this Lease as of the date of the occurrence of such damage. Landlord shall deliver to Tenant written notice of Landlord's election within sixty (60) days after the date of the occurrence of the damage, which notice shall also specify the expected time to restore the Premises if Landlord elects to repair the damages.

12.3 If at any time during the Term the Premises are damaged and such damage was caused by a casualty not covered under the insurance policy Landlord is required to carry pursuant to Section 14.2, Landlord may, at its option, either (i) repair such damage as soon as reasonably possible at Landlord's expense, in which event this Lease shall continue in full force and effect, or (ii) cancel and terminate this Lease as of the date of the occurrence of such damage, by giving Tenant written notice of Landlord's election to do so within thirty (30) days after the date of occurrence of such damage, in which event this Lease shall so terminate unless within thirty (30) days thereafter Tenant agrees to repair the damage at its cost and expense or pay for Landlord's repair of such damage.

12.4 Notwithstanding anything to the contrary herein, if it is determined that the damage or destruction resulting from a casualty cannot be repaired within twelve (12) months following the date of casualty, Tenant may terminate this Lease by written notice delivered to Landlord within thirty (30) days following Tenant's receipt of Landlord's written notice given under Section 12.2 or 12.3 above.

12.5 In the event of any damage or destruction the Base Rent and all Additional Rent payable by Tenant hereunder shall be proportionately reduced from the date of casualty until the completion by Landlord of any repair or restoration pursuant to this <u>Article 12</u> (provided that the abatement period shall not exceed twelve (12) months). Said reduction shall be based upon the extent to which the damage or the making of such repairs or restoration shall interfere with Tenant's business conducted in the Premises.

12.6 Landlord shall in no event be required or obligated to repair, restore or replace any of Tenant's Personal Property. Landlord shall restore the Tenant Improvements and Special Tenant Improvements (if any) to the extent of insurance proceeds received by Landlord. In the event of a termination of this Lease pursuant to this Article 12, Landlord shall pay to Tenant from the proceeds of the

insurance payable to Landlord with respect to the Tenant Improvements and Special Tenant Improvements an amount equal to the unamortized cost of Tenant's ownership interest in the Tenant Improvements and the Special Tenant Improvements.

12.7 In the event of a dispute by the parties regarding the extent of damage, duration of repair or rights of termination under Articles 12 or 13 only of the Lease, either party can request arbitration within ninety (90) days after the date of the damage has occurred. In such event the dispute shall be resolved by arbitration in accordance with the procedures set forth in Section 10.4 of the Declaration. The decision of the arbitrator shall be binding and conclusive on the parties.

ARTICLE 13. EMINENT DOMAIN

13.1 If all or substantially all of the Premises shall be taken or appropriated by any public or quasi-public authority under the power of eminent domain (or similar law authorizing the involuntary taking of private property, which shall include a sale in lieu thereof to a public body), either party hereto shall have the right, at its option, to terminate this Lease effective as of the date possession is taken by said authority, and Landlord shall be entitled to any and all income, rent, award and any interest thereon whatsoever which may be paid or made in connection with such public or quasi-public use or purpose. Tenant shall have no claim against Landlord for any portion of Landlord's award and shall not make a claim for the value of any unexpired term of this Lease.

13.2 If only a portion of the Premises is taken such that the Premises are still accessible and usable for the operation of Tenant's business, then this Lease shall continue in full force and effect and the proceeds of the award shall be used by Landlord to restore the remainder of the improvements on the Premises so far as practicable to a complete unit of like quality and condition to that which existed immediately prior to the taking, and all Rent payable by Tenant hereunder shall be reduced in proportion to the floor area of the Premises which is no longer available for Tenant's use. Landlord's restoration work shall not exceed the scope of work done by Landlord in originally constructing the Premises and the cost of such work shall not exceed the amount of the award received by Landlord with respect to the Premises.

13.3 Nothing hereinbefore contained shall be deemed to deny to Tenant its right to seek a separate award from the condemning authority for the unamortized costs of Tenant's ownership interest in the Tenant Improvements and Special Tenant Improvements, damage to its trade fixtures and personal property, relocation expenses or loss of goodwill.

ARTICLE 14. INSURANCE

14.1 Tenant shall, at all times during the Term hereof, at its expense, carry and maintain insurance policies in the amounts and in the form hereafter provided:

(a) **Commercial Liability and Property Damage:** Commercial general liability insurance in an amount not less than One Million Dollars (\$1,000,000) per occurrence and Two Million Dollars (\$2,000,000) in the general aggregate of bodily injury and property damage insuring against liability of the insured with respect to the Premises or arising from the maintenance, use or occupancy thereof. All such insurance shall include contractual liability insurance for the bodily injury, personal injury and property damage liability assumed by Tenant in Article 16 hereof. Said insurance shall provide that Landlord is named as an additional insured and will have a "separation of insureds" clause. Landlord's recovery under Tenant's insurance as an additional insured shall apply to loss or damages resulting from Tenant's negligence and shall not be restricted due to any contributory negligence on the part of Landlord. However, Tenant's insurance shall not responsible for loss or damage that is determined to be due to the sole negligence of Landlord. The insurance by this policy shall be primary insurance. The liability insurance required to be provided by Tenant shall be applicable to claims incurred by reason of events with respect to the Premises or arising from the maintenance, use or occupancy thereof during the term of this Lease, regardless of when such claims shall be first made against Tenant and/or Landlord. Should any required liability insurance be written on a claims-made basis, Tenant shall continue to provide evidence of such coverage beyond the term of this Lease, for a period mutually agreed upon by Landlord and Tenant at the time of termination, but in no event for a period of less than five years. Not more frequently than once each year, if in the opinion of Landlord's lender or of the insurance consultant retained by Landlord, the amount of liability insurance coverage at that time is not adequate, Tenant shall increase the insurance coverage as either required by Landlord's lender or recommended by Landlord'

(b) **Tenant Personal Property**: Insurance covering all of Tenant's trade fixtures, merchandise and other personal property from time to time in the Premises in an amount equal to their full replacement cost from time to time, providing protection against the "risks of physical damage" as provided in the *ISO Causes of Loss – Special Form (CP 10 30)*, or equivalent insurance company form.

15

The proceeds of such insurance shall, so long as this Lease remains in effect, be used to repair or replace the property damaged or destroyed, as determined by Tenant.

(c) Worker's Compensation: Worker's Compensation insurance as required by the State of California

(d) **Policy Form:** All insurance to be carried by Tenant hereunder shall be in companies, on forms and with loss payable clauses satisfactory to Landlord. The commercial liability and property damage insurance carried by Tenant pursuant to Section 14.1(a) above shall name Landlord, its managers, their officers, directors, partners, employees and agents as additional insureds. Each policy shall include a notice of cancellation to additional insured on the Additional Insured endorsement providing that no such policy shall be canceled except upon thirty (30) days advance notice to all additional insureds by the issuing company in the event of cancellation. Tenant shall have the right to maintain required insurance under blanket policies provided that Landlord and such parties as Landlord may reasonably designate from time are named therein as additional insureds (as to Tenant's liability policies)and that the coverage afforded Landlord will not be reduced or diminished by reason thereof, including self funded insurance reserves.

(e) **Evidence of Insurance**: Concurrent with delivery of possession of the Premises to Tenant, Tenant shall provide Landlord with the following evidence of insurance:

and

(i)

Certificate evidencing that each of the insurance policies required in subparagraphs (a), (b) and (c) above are in full force and effect,

(ii) A copy of the applicable provision or endorsement from each of Tenant's policies specifying that Landlord and the parties designated by Landlord are additional insureds, that the insurer recognizes the waiver of subrogation set forth in <u>Article 15</u> hereof, and that the insurer agrees not to cancel the policy without the notice to Landlord specified in subparagraph (d) above.

14.2 Subject to reimbursement by Tenant as provided in <u>Article 10</u> herein, Landlord shall obtain and keep in force during the term hereof, a policy or policies of insurance covering loss or damage to Building 6 and improvements on Landlord's Parcels. Landlord's insurance shall cover the "risks of physical damage" as provided in the *ISO Causes of Loss – Special Form (CP 10 30)*, or equivalent insurance company form, together with an endorsement providing for rental income insurance covering all Rent payable by Tenant hereunder for a period of twelve (12) months.

14.3 Landlord's policy described in Section 14.2 shall also insure all Tenant Improvements and Special Tenant Improvements for one hundred percent of the replacement cost thereof, with an agreed amount endorsement in lieu of coinsurance. Tenant shall pay to Landlord the cost of the insurance covering the Tenant Improvements and Special Tenant Improvements as provided in <u>Article 10</u> herein. Tenant acknowledges that Landlord's insurance on the Tenant and Special Tenant Improvements will not include earthquake insurance. Upon Tenant's request, Landlord shall obtain such coverage at Tenant's sole cost and expense.

14.4 If Tenant shall fail to procure and maintain any insurance policy required herein, Landlord may (but shall not be obligated to), after reasonable written notice to Tenant procure the same on Tenant's behalf, and the cost of same shall be payable as Additional Rent within ten (10) business

days after written demand therefor by Landlord. Tenant's failure to pay such Additional Rent shall constitute an Event of Default of this Lease, and Landlord may, without any further notice, exercise its remedies specified in <u>Article 25</u> hereof.

ARTICLE 15. WAIVER OF SUBROGATION

Any fire and special extended coverage insurance and any other property damage insurance carried by either party with respect to Landlord's Parcels, the Common Areas, the Premises and property contained in the Premises or occurrences related to them shall include a clause or endorsement denying to the insurer rights of subrogation against the other party to the extent rights have been waived by the insured prior to occurrence of damage or loss. Each party, notwithstanding any provisions of this Lease to the contrary, waives any right of recovery against the other for injury or loss due to hazards covered by insurance containing such clause or endorsement to the extent that the damage or loss is covered by such insurance.

ARTICLE 16. RELEASE AND INDEMNITY

16.1 Tenant shall indemnify, defend and hold harmless Landlord against and from any and all claims, actions, damages, liability and expenses, including reasonable attorneys' fees, arising from or out of Tenant's use of the Premises or from the conduct of its business or from any activity, work, or other

16

things done, permitted or suffered by the Tenant in or about the Premises or Tenant's reserved parking spaces. Tenant shall further indemnify, defend and hold Landlord harmless from any and all claims arising from any negligent act or omission or willful misconduct of Tenant, or any officer, agent, employee, contractor, guest, or invitee of Tenant, and from all costs, damages, attorney's fees, and liabilities incurred in defense of any such claim of any action or proceeding brought thereon, including any action or proceeding brought against Landlord by reason of such claim. Tenant, as a material part of the consideration to Landlord, hereby assumes all risk of damage to property or injury to persons in the Premises, from any cause except to the extent arising out of or resulting from Landlord's (or its agents', employees' or contractors') negligent act or omission or willful misconduct. Tenant shall give prompt notice to Landlord in case of casualty or accidents in the Premises.

16.2 Landlord shall indemnify, defend and hold harmless Tenant against and from any and all claims, actions, damages, liability and expenses, including reasonable attorneys' fees, arising from or out of any activity, work, or other things done by Landlord, its agents, employees or contractors in or about the Outdoor Areas and Common Areas on Landlord's Parcels. Landlord shall further indemnify, defend and hold Tenant harmless from any and all claims arising from the negligent act or omission or willful misconduct of Landlord, or any officer, agent, employee, or contractor of Landlord while on any of Landlord's Parcels or Buildings, and from all costs, damages, attorney's fees, and liabilities incurred in defense of any such claim of any action or proceeding brought thereon, including any action or proceeding brought against Tenant by reason of such claim.

16.3 Except to the extent arising out of or resulting from Landlord's negligent act or omission or willful misconduct, Landlord shall not be liable for injury or damage which may be sustained by the person, goods, wares, merchandise or property of Tenant, its employees, invitees or customers, or by any other person in or about the Premises caused by or resulting from fire, building vibrations or movement of floor slab, steam, electricity, gas, water or rain which may leak or flow from or into any part of the Premises, or from the breakage, leakage, obstruction or other defects of the pipes, sprinklers, wires, appliances, plumbing, air conditioning or lighting fixtures of the same, whether said damage or injury results from conditions arising upon the Premises or from other sources. Landlord shall not be liable for any damages arising from any act or neglect of any other tenant of the Building. Notwithstanding the foregoing, nothing contained herein shall limit any representations, warranties or covenants of Landlord set forth in this Lease, or any warranties provided with respect to work performed by Landlord's contractors. Further, notwithstanding the foregoing, the terms of <u>Article 12</u> shall govern with respect to any events of casualty.

ARTICLE 17. INSOLVENCY, ETC. OF TENANT

17.1 The filing of any petition in bankruptcy whether voluntary or involuntary, or the adjudication of Tenant as bankrupt or insolvent, or the appointment of a receiver or trustee to take possession of all or substantially all of Tenant's assets, or an assignment by Tenant for the benefit of its creditors, or any action taken or suffered by Tenant under any State or Federal insolvency or bankruptcy act including, without limitation, the filing of a petition for or in reorganization, or the taking or seizure under levy of execution or attachment of the Premises or any part thereof, shall constitute a breach of this Lease by Tenant, and in any one or more of said events this Lease shall be deemed terminated to the extent such result is permitted by relevant bankruptcy laws and statutes.

17.2 Landlord shall be entitled, notwithstanding any provision of this Lease to the contrary, upon re-entry of the Premises in case of a breach under this Article, to recover from Tenant as damages, and not as a penalty, such amounts as are specified in <u>Article 25</u>, unless any statute governing the proceeding in which such damages are to be proved shall lawfully limit the amount thereof capable of proof, in which later event Landlord shall be entitled to recover as and for its damages the maximum amount permitted under said statute.

ARTICLE 18. PERSONAL PROPERTY AND OTHER TAXES

18.1 Tenant shall pay, before delinquency, any and all taxes and assessments, sales, use, business, occupation or other taxes, and license fees or other charges whatever levied, assessed or imposed upon its business operations conducted in the Premises. Tenant shall also pay, before delinquency, any and all taxes and assessments levied, assessed or imposed upon its equipment, furniture, furnishings, trade fixtures, merchandise and other personal property in, on or upon the Premises.

18.2 Tenant shall pay all taxes and assessments levied, assessed or imposed on Tenant's trade fixtures and its leasehold improvements, regardless of whether such improvements were installed and/or paid for by Tenant or by Landlord, and regardless of whether or not the same are deemed to be a part of the Building.

18.3 Tenant shall pay (or reimburse Landlord therefor forthwith on demand) any excise tax, gross receipts tax, or any other tax however designated, and whether charged to Landlord, or to Tenant, or to either or both of them, which is imposed on or measured by or based on the rentals to be paid under this Lease, or any estate or interest of Tenant, or any occupancy, use or possession of the Premises by Tenant.

18.4 Nothing hereinabove contained in this Article shall be construed as requiring Tenant to pay any inheritance, estate, succession, transfer, gift, franchise, income or profits tax or taxes imposed upon Landlord.

ARTICLE 19. SIGNS

Tenant shall not place, construct or maintain on the windows, doors or exterior walls or roof of the Premises or any interior portions that may be visible from the exterior of the Premises, any signs, advertisements, names, trademarks or other similar item without Landlord's consent, which consent shall not be unreasonably withheld or delayed so long as the signage Tenant installs complies with all Legal Requirements and the master sign program for the Center. Upon written notice from Landlord specifying the violation in reasonable detail, Tenant shall, at Tenant's cost, remove any item so placed or maintained which does not comply with the provisions of this Section. Landlord agrees that Landlord shall not install or permit the installation of signs or billboards on the exterior walls and/or the roof of the Premises.

See Addendum 32.26.

ARTICLE 20. ASSIGNMENT AND SUBLETTING

20.1 Subject to the terms of <u>Section 20.4</u>, Tenant shall not voluntarily, involuntarily, or by operation of law assign, transfer, hypothecate, or otherwise encumber this Lease or Tenant's interest therein, and shall not sublet nor permit the use by others of the Premises or any part thereof without first obtaining in each instance Landlord's written consent. If consent is once given by Landlord to any such assignment, transfer, hypothecation or subletting, such consent shall not operate as a waiver of the necessity for obtaining Landlord's consent to any subsequent assignment, transfer, hypothecation or sublease, and no assignment shall release Tenant from any liability hereunder. Any such assignment or transfer without Landlord's consent shall be void and shall, at Landlord's option, constitute an Event of Default of this Lease. This Lease shall not, nor shall any interest therein, be assignable as to Tenant's interest by operation of law, without Landlord's express prior written consent.

20.2 The consent of Landlord required under <u>Section 20.1</u> above shall not be unreasonably withheld or delayed. Should Landlord withhold its consent for any of the following reasons, the withholding shall be deemed to be reasonable:

- (a) Conflict of the proposed use with other uses in the Building or Center;
- (b) Financial inadequacy of the proposed subtenant or assignee;
- (c) A proposed use which would diminish the reputation of the Center or the other businesses located therein;
- (d) A proposed use which would have a detrimental impact on the common facilities or the other tenants in the Center.

20.3 Each assignee or transferee shall agree to assume and be deemed to have assumed this Lease and shall be and remain liable jointly and severally with Tenant for the payment of all rents due here under, and for the due performance during the term of all the covenants and conditions herein set forth by Tenant to be performed. No assignment or transfer shall be effective or binding on Landlord unless said assignee or transferee shall, concurrently, deliver to Landlord an assumption agreement by said assignee or transferee assuming all obligations of Tenant under this Lease.

20.4 Notwithstanding anything to the contrary herein, Landlord's consent shall not be required for any assignment, transfer or sublease to any entity which controls, is controlled by or under common control with Tenant, or to any entity resulting from a reorganization, merger or sale of substantially all of the assets of Tenant. The term "control" shall mean the ownership of at least 50% of the stock or assets of Tenant. Further, Landlord's consent shall not be required for any offering of the stock of Tenant on the public market or any open market transactions involving the stock of Tenant. If Tenant is not a publicly traded corporation, or if Tenant is an unincorporated association or a partnership, the transfer, assignment, or hypothecation or any stock or interest in such corporation, association or partnership in the aggregate of in excess of fifty percent (50%) shall be deemed an assignment within the meaning of this Article, except transfers in connection with Tenant becoming a publicly traded corporation. Tenant shall give Landlord

1	O.
т	0

prior written notice of all transfers, whether or not consent is required, and in no event shall Tenant be released from any of its obligations under this Lease.

20.5 If Tenant intends to assign this Lease and Landlord's consent to such assignment is required, Tenant shall give prior written notice to Landlord of each such proposed assignment or subletting specifying the proposed assignee or subtenant and the terms of such proposed assignment or sublease. Landlord shall, within fifteen (15) business days thereafter, notify Tenant in writing either, that (i) it consents (subject to any conditions of consent that may be imposed by Landlord) or does not consent to such transaction, or (ii) it elects to cancel this Lease in which event the parties would have no further obligations to each other except with respect to obligations which arose prior to the effective date of termination or which otherwise survive the termination of this Lease.

20.6 In the event of an assignment or subletting which requires Landlord's consent pursuant to this Article 20, Tenant shall assign to Landlord 75% of any and all consideration paid to Tenant directly or indirectly for the assignment by Tenant of its leasehold interest, and 75% of any and all subrentals payable by sublessees to Tenant which are in excess of the Rent payable by Tenant hereunder. Tenant's brokerage fees shall be paid by Tenant and deducted from excess proceeds on a pro rata basis monthly over the term of the sublease.

20.7 Tenant agrees to reimburse Landlord for Landlord's reasonable costs and attorneys fees' incurred in connection with the processing and documentation of any requested assignment, transfer, hypothecation or subletting of this Lease aforesaid, whether or not such consent is granted, in an amount not to exceed \$2500 in each instance.

ARTICLE 21. RIGHTS RESERVED BY LANDLORD

Subject to Tenant's reasonable security and trade secret requirements, upon reasonable prior notice, Landlord or its agents shall have the right to enter the Premises for the purposes of:

- (a) Inspection of the Premises and the equipment therein, not to exceed once per calendar quarter (or not to exceed once per year for inspections of any clean room), except in the event of an emergency or unless a known problem exists or Landlord is responding to a third party complaint involving the Premises;
- (b) Making repairs or improvements to the Premises and/or Building 6 which are the responsibility of Landlord under the terms of this Lease;
- (c) Performing remodeling, construction or other work incidental to any portion of the Building 6, including, without limitation, the premises of another tenant adjacent to, above or below the Premises. Landlord agrees to coordinate the timing and staging of any major construction program with Tenant
- (d) Showing the Premises to persons wishing to purchase or make a mortgage loan upon the same;
- (e) Posting notice of non-responsibility;
- (f) Posting "For Lease" signs and showing the Premises to persons wishing to rent the Premises during the last six (6) months of the term of this Lease.

ARTICLE 22. INTENTIONALLY DELETED

ARTICLE 23. RIGHT OF LANDLORD TO PERFORM

All covenants to be performed by Tenant hereunder shall be performed by Tenant at its sole cost and expense and without any abatement of any rent to be paid hereunder, subject to the terms and conditions set forth in this Lease. If Tenant shall fail to pay any sum, other than rent, required to be paid by it or shall fail to perform any other act on its part to be performed, and such failure shall continue beyond the applicable notice and grace period set forth in <u>Article 25</u>, Landlord may (but shall not be obligated to) and without waiving or releasing Tenant from any of its obligations, make any such payment or perform any such other act on Tenant's part to be made or performed as herein provided. All sums so paid by Landlord and all necessary incidental costs, together with interest at the Interest Rate from the date of such payment by Landlord shall be payable by Tenant as Additional Rent within thirty (30) days after Landlord's written demand therefor. Tenant's failure to pay such Additional Rent shall constitute an Event of Default of this Lease, and Landlord may, without any further notice, exercise its remedies specified in <u>Article 25</u> hereof.

19

ARTICLE 24. LANDLORD DEFAULT

24.1 If Landlord shall be in default of any covenant of this Lease to be performed by it, Tenant, prior to exercising any right or remedy it may have against Landlord on account thereof, shall give Landlord a thirty (30) day written notice of such default, specifying the nature of such default. Notwithstanding anything to the contrary elsewhere in this Lease, Tenant agrees that if the default specified in said notice is of such nature that it can be cured by Landlord, but cannot with reasonable diligence be cured within said thirty (30) day period, then such default shall be deemed cured if Landlord within said thirty (30) days period shall have commenced the curing thereof and shall continue thereafter with all due diligence to cause such curing to proceed to completion.

24.2 If Landlord shall fail to cure a default of any covenant of this Lease to be performed by it within the time period provided in <u>Section 24.1</u>, the same shall be deemed an Event of Default by Landlord and, subject to <u>Section 24.3</u>. Tenant may pursue all remedies available at law or in equity and may recover all costs and expenses incurred by Tenant by reason of such default by Landlord. Notwithstanding the foregoing, if Tenant shall recover a money judgment against Landlord, such judgment shall be satisfied solely out of the right, title and interest of Landlord in the Premises and its underlying realty and out of the rents, or other income from said property receivable by Landlord, or out of the consideration received by Landlord's right, title and interest in said property, but neither Landlord nor any partner or joint venture of Landlord shall be personally liable for any deficiency.

24.3 Tenant agrees to give any mortgagee and/or trust deed holders ("**Mortgagee**"), by registered mail, a copy of any notice of default served upon the Landlord, provided that prior to such notice Tenant has been notified in writing (by way of Notice of Assignment of Rents and Leases, or otherwise) of the address of such Mortgagee. Tenant further agrees that if Landlord shall have failed to cure such default within the time provided for in this Lease, then the Mortgagee shall have an additional sixty (60) days within which to cure such default or if such default cannot be cured within that time, then such additional time as may be necessary to cure such default shall be granted if within such sixty (60) days Mortgagee has commenced and is diligently pursuing the remedies necessary to cure such default (including, but not limited to, commencement of foreclosure proceedings, if necessary to effect such cure), in which event the Lease shall not be terminated while such remedies are being so diligently pursued.

ARTICLE 25. DEFAULT AND REMEDIES

25.1 The occurrence of any of the following shall constitute an "**Event of Default**" under this Lease by Tenant:

(a) Any failure by Tenant to pay when due any of the Rent required to be paid by Tenant hereunder where such failure continues for five (5) business days after Tenant's receipt of written notice that the same is overdue;

(b) A failure by Tenant to observe and perform any other provision of this Lease to be observed or performed by Tenant where such failure continues for thirty (30) days after written notice thereof from Landlord; provided, that if the nature of such default is such that the same cannot with due diligence be cured within said period, Tenant shall not be deemed to be in default if it shall within said period commence such during and thereafter diligently prosecutes the same to completion;

(c) Any default by Tenant under any other lease between Landlord and Tenant for other premises in the Center;

(d) The abandonment or vacation of the Premises, provided that if Tenant has vacated the Premises and is actively seeking a subtenant or assignee, no default shall be deemed to exist under this Lease so long as Tenant is paying the Rent required to be paid hereunder; and

(e) Any other event herein specified to be an Event of Default under this Lease.

25.2 In the event of any Event of Default by Tenant as aforesaid, in addition to any and all other remedies available to Landlord at law or in equity, Landlord shall have the right to immediately terminate this Lease and all rights of Tenant hereunder by giving written notice to Tenant of its election to do so. If Landlord shall elect to terminate this Lease, then it may recover from Tenant:

(a) The worth at the time of the award of the unpaid rent payable hereunder which had been earned at the date of such termination; plus

20

(b) The worth at the time of the award of the amount by which the unpaid rent which would have been earned after termination and until the time of the award exceeds the amount of such rental loss which Tenant proves could have been reasonably avoided; plus

(c) The worth at the time of the award of the amount by which the unpaid rent for the balance of the term after the time of the award exceeds the amount of such rental loss which Tenant proves could be reasonably avoided; plus

(d) Any other amounts necessary to compensate Landlord for all detriment proximately caused by Tenant's failure to perform its obligations hereunder or which, in the ordinary course of affairs, would likely result therefrom, and

(e) At Landlord's election, such other amounts in addition to or in lieu of the foregoing as may be permitted by applicable California law from time to time.

25.3 As used in subparagraphs (a) and (b) above, the "worth at the time of the award" is computed by allowing interest at the rate of twelve (12%) percent per annum (the "**Interest Rate**"). As used in subparagraph (c) above, the "worth at the time of the award" is computed by discounting such amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of the award plus one (1%) percent.

25.4 Following the occurrence of an Event of Default by Tenant, Landlord shall also have the right, with or without terminating this Lease, to reenter the Premises and remove all property and persons therefrom, and any such property may be removed and stored in a public warehouse or elsewhere at the cost and for the account of Tenant, all in accordance with all Legal Requirements.

25.5 If Landlord (in accordance with California Civil Code Section 1951.4) shall elect to re-enter as above provided or shall take possession of the Premises pursuant to legal proceedings or pursuant to any notice provided by law, and if Landlord has not elected to terminate this Lease, Landlord may continue this Lease and may either recover all rental as it becomes due or relet the Premises or any part or parts thereof for such term or terms and upon such provisions as Landlord, in its sole judgment, may deem advisable and shall have the right to make repairs to and alterations of the Premises.

25.6 If Landlord shall elect to relet as aforesaid, then rentals received by Landlord therefrom shall be applied as follows:

- (a) to the payment of any indebtedness other than rent due hereunder from Tenant;
- (b) to the payment of all costs and expenses incurred by Landlord in connection with such reletting;
- (c) to the payment of the cost of any alterations of and repairs to the Premises; and

(d) to the payment of rent due and unpaid hereunder and the residue, if any, shall be held by Landlord and applied in payment of future rent as the same may become due and payable hereunder.

In no event shall Tenant be entitled to any excess rental received by Landlord over and above that which Tenant is obligated to pay hereunder. Should that portion of such rentals received from such reletting during any month, which is applied to the payment of rent hereunder, be less than the rent payable hereunder during that month by Tenant, then Tenant shall pay such deficiency to Landlord forthwith upon demand, and said deficiency shall be calculated and paid monthly. Tenant shall also pay Landlord as soon as ascertained and upon demand, all costs and expenses incurred by Landlord in connection with such reletting and in making any such alterations and repairs which are not covered by the rentals received from such reletting.

25.7 No re-entry or taking possession of the Premises by Landlord under this Article shall be construed as an election to terminate this Lease unless a written notice of such intention is given to Tenant or unless the termination thereof be adjudged by a court of competent jurisdiction. Notwithstanding any releting without termination by Landlord because of Tenant's default, Landlord may at any time after such releting elect to terminate this Lease because of such default.

25.8 Nothing contained in this Article shall constitute a waiver of Landlord's right to recover damages by reason of Landlord's efforts to mitigate the damages to it caused by Tenant's default; nor shall anything in this Article adversely affect Landlord's right, as in this Lease elsewhere provided, to indemnification against liability for injury or damage to persons or property occurring prior to a termination of this Lease.

25.9 If Landlord shall retain an attorney for the purpose of collecting any rental due from Tenant or enforcing any other covenant of this Lease, Tenant shall pay the reasonable fees of such attorney for his services regardless of the fact that no legal proceeding or action may have been filed or commenced.

25.10 Any unpaid rent and any other sums due and payable hereunder by Tenant shall bear interest at the maximum lawful rate per annum from the due date and until payment thereof.

25.11 The terms "**Rent**," "**rent**" and "**rental**" as used herein and elsewhere in this Lease shall be deemed to be and mean the Base Rent, all Additional Rent, rental adjustments and any and all other sums, however designated, required to be paid by Tenant hereunder.

25.12 Tenant acknowledges that late payment by Tenant to Landlord of rent will cause Landlord to incur costs not contemplated by this Lease, the exact amount of such costs being extremely difficult and impracticable to fix. Such costs include, without limitation, processing and accounting charges, and late charges that may be imposed on Landlord by the terms of any encumbrance and note secured by any encumbrance covering the Premises. Therefore, if any installment of rent due from Tenant is not received by Landlord when due more than once in any calendar year during the Term, Tenant shall pay to Landlord as additional rent an additional sum of six percent (6%) of the overdue rent as a late charge. The parties agree that this late charge represents a fair and reasonable estimate of the costs that Landlord will incur by reason of late payment by Tenant. Acceptance of any late charge shall not constitute a waiver of Tenant's default with respect to the overdue amount, nor prevent Landlord from exercising any of the other rights and remedies available to Landlord.

25.13 If Landlord shall retain a collection agency for the purpose of collecting any moneys due from Tenant arising out of an Event of Default hereunder, Tenant shall pay all fees of such collection agency for their services.

ARTICLE 26. PRIORITY OF LEASE AND ESTOPPEL CERTIFICATE

26.1 At Landlord's election, this Lease shall be either superior to or subordinate to any and all trust deeds, mortgages, or other security instruments, ground leases, or leaseback financing arrangements now existing or which may hereafter be executed covering the Premises and/or the land underlying the same or any part or parts of either thereof, and for the full amount of all advances made or to be made thereunder together with interest thereon, and subject to all the provisions thereof, all without the necessity of having further instruments executed by Tenant to effectuate the same. Tenant agrees to execute, acknowledge and deliver upon request by Landlord any and all documents or instruments which are or may be deemed necessary or proper by Landlord to more fully and certainly assure the superiority or the subordination of this Lease and to any such trust deeds, mortgages or other security instruments, ground leases, or leasebacks <u>provided that</u> as a condition to any such subordination and if this Lease shall be made subordinate to any future security instrument, or by termination of said ground leases or leasebacks, shall continue this Lease in full force and effect in the same manner as if such person or persons had been named as Landlord herein and this Lease shall continue in full force and effect as aforesaid, and Tenant shall automatically become the tenant of Landlord's successor in interest and shall attorn to said successor in interest. The words "**person**" and "**persons**" as used herein or elsewhere in this Lease shall mean individuals, partnerships, firms, associations and corporations.

See Addendum A-26.1

26.2 Landlord and Tenant shall at any time and from time to time execute, acknowledge and deliver to the other party hereto, within ten (10) business days after such party's written request therefor, a written statement certifying as follows:

(a) that this Lease is unmodified and in full force (or if there has been modification thereof, that the same is in full force as modified and stating the nature thereof);

(b) that to the best of its knowledge, there are no uncured defaults or matters which, upon the passage of time and the giving of notice, or both, would constitute a default or breach by Tenant or Landlord, as applicable (or if such exist, the specific nature and extent);

(c) that no claims or defenses exist on the part of the certifying party and no events exist that would constitute a basis for such claim or defense (or if such exist, the specific nature and extent);

(d) the date to which any rents and other charges have been paid in advance, if any;

22

(e) such other matters which are reasonably requested by the requesting party with respect to the Lease and its status, including status of construction; and

(f) in the case of Tenant's certificate, that Tenant will not enter into any agreements or modification of the Lease without the prior written consent of the lender specified by Landlord, provided such consent would not be unreasonably withheld.

If Landlord or Tenant shall fail to execute and deliver any such statement to the requesting party within ten (10) business days, the requesting party may deliver a second written notice requesting the statement. If the party required to deliver the statement fails to make such delivery within five (5) business days following such second notice, the failure shall constitute an Event of Default hereunder entitling the requesting party to pursue available remedies as set forth in this Lease.

26.3 At Landlord's election, this Lease shall be subordinate to any and all encumbrances, covenants, restrictions, conditions and easements of record now existing or which hereafter may be executed ("Record Matters") covering the Premises and/or the land underlying the same or any parts thereof without the necessity of having further instruments executed by Tenant to effectuate the same, provided that any future encumbrances shall be subject to the provision of Section 26.1 above and any other Record Matters recorded after the date of this Lease shall not materially and adversely affect Tenant's use of the Premises. Landlord hereby confirms that it has no present knowledge of the existence of any encumbrances, covenants, restrictions, conditions or easements of record which now exist, or which will be recorded in the future with respect to Parcel 6, that would materially and adversely affect Tenant's use of the Premises other than those shown in the title report for Center attached hereto as Exhibit H.

ARTICLE 27. HOLDING OVER

If, without the execution of a new lease or written extension of this Lease, and with the consent of Landlord, Tenant shall hold over after the expiration of the Term of this Lease, Tenant shall be deemed to be occupying the Premises as a tenant from month-to-month, which tenancy may be terminated as provided by law. During said tenancy, the Base Rent payable to Landlord by Tenant shall be one hundred fifty percent (150%) of the Base Rent set forth in <u>Article 3</u> of this Lease which is payable immediately preceding the date of expiration of this Lease, and upon all of the other terms, covenants and conditions set forth in this Lease so far as the same are applicable.

If Tenant shall holdover and fail to surrender the Premises upon the termination of this Lease without Landlord's consent, in addition to any other liabilities to Landlord arising therefrom, Tenant shall and does hereby agree to indemnify and hold Landlord harmless from loss or liability resulting from such failure including, but not limited to, claims made by any succeeding tenant founded on such failure.

ARTICLE 28. NOTICES

All notices, approvals, demands, consents or other communications required or permitted under this Lease shall be in writing and shall be deemed to have been given when personally served or received by certified mail, postage prepaid, or on the next business day sent by telefax, Express Mail, Federal Express or similar reputable overnight delivery service, addressed to the appropriate party at the address indicated next to each party's signature below. Notwithstanding the foregoing, notices during the initial construction of the Premises relating to construction matters shall be governed by the provisions of Exhibit C.

ARTICLE 29. LIENS

29.1 Tenant shall pay all costs for work done by it or caused to be done by it in the Premises and Tenant shall keep the Premises and the Center free and clear of all mechanics' liens and other liens of account or work done for Tenant or persons claiming under it. Notwithstanding the foregoing, Tenant shall have no responsibility or liability with respect to liens filed with respect to the Base Building, and Tenant Improvements or any other work performed by Landlord pursuant to <u>Article 8</u>, <u>Exhibit C</u> or otherwise. Tenant agrees to and shall indemnify and hold Landlord harmless against liability, loss, damage, costs, attorneys' fees, and any other expenses on account of claims of liens of laborers or materialmen for work performed or materials or supplies furnished for Tenant or persons claiming under it. If any such lien shall attach to the Premises or the Center by reason of any work performed by Tenant, Tenant shall promptly, and in any event within twenty (20) days thereafter, discharge it as a matter of record or bond over it. If necessary to accomplish same, Tenant shall furnish and record a bond to insure the protection of Landlord, the Premises, and the Center (including all buildings located thereon or of which they form a part) from loss by virtue of any such lien.

29.2 Any bond furnished by Tenant pursuant to the provisions of <u>Section 29.1</u> above shall be a lien release bond issued by a corporation authorized to issue surety bonds in the State of

2	2
2	3

California in an amount equal to one and one-half the amount of such claim of lien. The bond shall meet the requirements of Civil Code section 3143 and shall provide for the payment of any sum that the claimant may recover on the claim, together with said lien claimant's costs of suit if he recovers therein.

29.3 If a mechanics' lien which is Tenant's responsibility pursuant to <u>Section 29.1</u> above has been filed, and Tenant shall not have discharged same of record within the time permitted by that Section, Landlord may (but shall not be obligated to) pay said claim and any costs, and the amount so paid, together with reasonable attorneys' fees incurred in connection therewith shall be payable by Tenant to Landlord as Additional Rent within five (5) days after written demand therefor. Tenant's failure to pay such Additional Rent shall constitute an Event of Default of this Lease, and Landlord may, without any further notice, exercise its remedies specified in <u>Section 25</u> hereof.

29.4 Tenant shall, at least ten (10) days prior to commencing any work which might result in a lien as aforesaid, give Landlord written notice of its intention to commence such work, to enable Landlord to post, file and record a legally effective notice of non-responsibility. Landlord or its representatives shall have the right to enter into the Premises and inspect the same at all reasonable times, and shall have the right to post and keep posted thereon said notices of non-responsibility and such other notices as Landlord may deem proper to protect its interest therein.

ARTICLE 30. QUIET ENJOYMENT

Landlord agrees that Tenant, upon payment of the Base Rent, Additional Rent, and all other sums and charges required to be paid by Tenant hereunder, and the due and punctual performance of all of Tenant's other covenants and obligations under this Lease, shall have the quiet and undisturbed possession of the Premises.

ARTICLE 31. ATTORNEYS' FEES

Should either party hereto institute any action or proceeding in court to enforce any provision hereof or for damages or for declaratory or other relief hereunder, the prevailing party shall be entitled to receive from the losing party, in addition to court costs, such amount as the court may adjudge to be reasonable as attorneys' fees for services rendered to said prevailing party, and said amount may be made a part of the judgment against the losing party.

ARTICLE 32. MISCELLANEOUS

32.1 Nothing contained in this Lease shall be deemed or construed as creating a partnership or joint venture between Landlord and Tenant or between Landlord and any other party, or cause Landlord to be in any manner responsible for the debts or obligations of Tenant, or any other party. The covenants in this Lease are made between the parties to the Lease and shall not be deemed or construed as creating any rights in any other party claiming to be a third party beneficiary of this agreement.

32.2 If any provision of this Lease shall be determined to be void or voidable by any court of competent jurisdiction, such determination shall not affect any other provision of this Lease and all such other provisions shall remain in effect. It is the intention of the parties hereto that if any

provision of this Lease is capable of two constructions, one of which would render the provision void or voidable and the other of which would render the provision valid, then the provision shall have the meaning which renders it valid.

32.3 If Tenant hereunder is a corporation or partnership, the parties executing this Lease on behalf of Tenant represent and warrant to Landlord that: they are authorized to enter into this Lease; this Lease is executed in the usual course of business of Tenant and that neither the corporate Articles nor Bylaws of Tenant or any partnership agreement of Tenant, as the case may be, require the consent of its shareholders or partners, as applicable, thereto; Tenant is a valid and existing corporation or partnership, as applicable; all things necessary to qualify Tenant to do business in California have been accomplished prior to the date of this Lease; all franchise and other taxes have been paid to the date of this Lease; all forms, reports, fees, and taxes required to be filed or paid by Tenant in compliance with all Legal Requirements will be filed and paid when due.

32.4 The entire agreement between the parties hereto is set forth in this Lease, and any agreement hereafter made shall be ineffective to change, modify, alter or discharge it in whole or in part unless such agreement is in writing and signed by both parties hereto. It is further understood that there are no oral agreements between the parties hereto affecting this Lease, and that this Lease supersedes and cancels any and all previous negotiations, arrangements, brochures, agreements and understandings, if any, between the parties hereto or displayed by Landlord to Tenant with respect to the subject matter of this

24

Lease, and none of the same shall be available to interpret or construe this Lease. All negotiations and oral agreements acceptable to both parties hereto have been merged into and are included in this Lease.

32.5 Landlord reserves the absolute right to effect such other tenancies in the Center. Tenant does not rely on the fact nor does Landlord represent that any specific tenant or number of tenants shall during the term of this Lease occupy any space in any Building.

32.6 The laws of the State of California shall govern the validity, performance and enforcement of this Lease. Should either party institute legal suit or action for enforcement of any obligation herein, it is agreed that the venue of such suit or action shall be in Alameda County, California, and Tenant expressly consents to Landlord's designating Alameda County as the venue of any such suit or action.

32.7 A waiver of any breach or default shall not be a waiver of any other breach or default. Landlord's consent to or approval of, any act by Tenant requiring Landlord's consent or approval shall not be deemed to waive or render unnecessary Landlord's consent to or approval of any subsequent similar act by Tenant. The acceptance by Landlord of any rental or other payments due hereunder with knowledge of the breach of any of the covenants of this Lease by Tenant shall not be construed as a waiver of any such breach. The acceptance at any time or times by Landlord of any sum less than that which is required to be paid by Tenant shall, unless Landlord specifically agrees otherwise in writing, be deemed to have been received only on account of the obligation for which it is paid, and shall not be deemed an accord and satisfaction notwithstanding any provisions to the contrary written on any check or contained in a letter of transmittal.

32.8 Any prevention, delay or stoppage due to strikes, lockouts, labor disputes, acts of God, inability to obtain labor or materials or reasonable substitutes therefore, failure of power, including delays by PG&E in hooking up utilities to the Premises, governmental restrictions, regulations or controls, enemy or hostile governmental action, riot, civil commotion, fire or other casualty, inclement weather beyond seasonal norm and other causes of a like nature beyond the reasonable control of the party obligated to perform (any such event being "Force Majeure"), shall excuse the performance by such party for a period equal to any such prevention, delay or stoppage, except that Tenant's obligations to pay Rent and any other sums or charges specifically due and payable pursuant to this Lease shall not be affected thereby.

32.9 The term "Landlord" as used in this Lease, so far as covenants or obligations on the part of Landlord are concerned, shall be limited to mean and include only the owner or owners at the time in question of the Premises, and in the event of any transfer or transfers of title thereto, Landlord herein named (and in case of any subsequent transfers or conveyances, the then grantor) shall be automatically freed and relieved from and after the date of such transfer or conveyance of all liability as respects the performance of any covenants or obligations hereunder of the part of Landlord to be performed thereafter.

32.10 The voluntary or other surrender of this Lease by Tenant, or a mutual cancellation thereof, shall not work a merger, and shall, at the option of the Landlord terminate all or any existing subleases and subtenancies, or may, at Landlord's option, operate as an assignment to it of any or all such subleases or subtenancies.

32.11 Although the printed provisions of this Lease were prepared and drawn by Landlord, this Lease shall not be construed either for or against Landlord or Tenant, but its construction shall be at all times in accord with the general tenor of the language so as to reach a fair and equitable result.

32.12 Except as otherwise expressly provided in this Lease, any and all "approvals", "consents" and "permissions" that either party is obligated or required to provide under this Lease shall not be unreasonably withheld or delayed.

32.13 Upon Landlord's written request not more often than once per year, Tenant shall promptly furnish to Landlord, from time to time, financial statements reflecting Tenant's current financial condition. If Tenant is a publicly held company, Tenant may furnish to Landlord Tenant's most recent publicly filed annual or quarterly report to satisfy this request.

32.14 Time is of the essence with respect to the performance of each of the covenants and agreements of this Lease.

32.15 Each and all of the provisions of this Lease shall be binding upon and inure to the benefit of the parties hereto and (except as set forth in Section 32.9 above and as otherwise specifically provided elsewhere in this Lease), their respective personal representatives, successors and assigns, subject

at all times to all provisions and restrictions elsewhere in this Lease respecting the assignment, transfer, encumbering or subletting of all or any part of the Premises or Tenant's interest in this Lease.

See Addendum A-32.15

32.16 Submission of this instrument by or on behalf of Landlord for examination or execution by Tenant does not constitute a reservation of or option for lease, and this instrument shall not be effective as a lease or otherwise until executed and delivered by both Landlord and Tenant.

32.17 The captions shown in this Lease are for convenience or reference only, and shall not, in any manner, be utilized to construe the scope or the intent of any provisions thereof.

32.18 This Lease shall not be recorded, but Tenant may record a short form Memorandum of this Lease at its expense and Landlord agrees to execute such a memorandum in a form reasonably approved by Landlord upon Tenant's request. In such event, upon Landlord's written request Tenant agrees to execute a quitclaim deed at the end of the term relinquishing any interest in the Premises.

32.19 Intentionally Deleted.

32.20 All agreements herein by Tenant, whether expressed as covenants or conditions, shall be deemed to be conditions for the purpose of this Lease.

32.21 The parties represent and warrant to each other that each has not dealt with any real estate agent other than Colliers International, as to Landlord, and The Staubach Company as to Tenant. Each agrees to indemnify and hold the other harmless from and against all loss, cost and expenses incurred by reason of the breach of such representation and warranty. Landlord shall be responsible for paying all commissions due, in accordance with the terms of a separate written agreement.

32.22 The terms of this Lease are confidential and constitute proprietary information of the parties. Neither party, nor its respective employees or agents, shall disclose the terms of this Lease to any other person without the prior written consent of the other party hereto, which consent may be withheld in such party's sole discretion. However, either party may disclose the terms of this Lease to its lenders, accountants and prospective transferees have a reasonable bona fide need to know such terms, and provided that the disclosing party ensures that such lenders, accountants and prospective transferees maintain the confidentiality of such terms In addition, either party may disclose the terms of this Lease in litigation or other dispute resolution proceeding between Landlord and Tenant with respect to the Lease subject to the Lease being filed under seal if the filing of the document would otherwise make it publicly available and if the court approves of filing under seal, and: (i) pursuant to an order of a court of competent jurisdiction, provided that the disclosing party promptly notifies the other party of any motion to compel such disclosure and the disclosure order, and/or (ii) in order to comply with any applicable Securities Exchange Commission laws, rules or regulations, provided that the disclosing party notifies the other party of the fact that such disclosure will take place, subject, however, to the disclosing party in each of (i) and (ii), using commercially reasonable best efforts to limit the scope and extent of the disclosure.

32.23 The Addendum attached hereto is hereby made a part of this Lease.

See Addendum A-32.24-32.27.

WITNESS the signatures of the parties hereto, the day and year first above written.

LANDLORD: **TENANT: GREENVILLE INVESTORS, L.P.** FORMFACTOR, INC., Bv: Greenville Ventures. Inc. a Delaware corporation Title: General Partner By: /s/ WILLIAM A. DRUMMOND By: /s/ STUART L. MERKADEAU William A. Drummond Stuart L. Merkadeau Vice President Its: SVP, General Counsel Its: & Secretary ADDRESS: ADDRESS: 2020 Research 675 Hartz Avenue, Suite 300 Drive Danville, CA 94526 Livermore, CA 94550

26

EXHIBIT A

SITE PLAN [Site Plan]

EXHIBIT A-1

DEPICTION OF BUILDING 6 DRIP LINE

[Depiction of Building 6 Drip Line]

A1-1

EXHIBIT B

CENTER LEGAL DESCRIPTION AND PARCEL MAP [Legal Description of Parcel]

B-1

EXHIBIT C

WORK LETTER [Work Letter]

EXHIBIT D

INTENTIONALLY DELETED [Intentionally Deleted]

D-1

EXHIBIT E

RULES AND REGULATIONS [Rules And Regulations]

E-1

EXHIBIT F

DESCRIPTION OF TENANT'S USE OF HAZARDOUS SUBSTANCES

[Product Names of Hazardous Substances]

F-1

EXHIBIT G

COPY OF CENTER COVENANTS, CONDITIONS AND RESTRICTIONS

[Copy Of Center Covenants, Conditions And Restrictions]

G-1

[Non-Disclosure Agreement]

H-1

ADDENDUM TO LEASE

A-2.1 **Options to Renew.** Provided that no Event of Default by Tenant under this Lease exists as of the date of exercise of the applicable option or at the expiration of the initial term or preceding Option Term, and provided further that Tenant has not assigned this Lease, Tenant shall have the option to extend the initial lease term for four (4) additional, successive terms of five (5) years each (each, an "**Option Term**"). Tenant shall exercise the option, if at all, by delivering to Landlord written notice of the exercise no sooner than fifteen (15) months nor later than twelve (12) months prior to the expiration of the initial Lease Term or preceding Option Term, as applicable. Tenant's right to exercise each option shall be conditioned upon Tenant delivering to Landlord with Tenant's notice of exercise, current financial reports which evidence that Tenant's financial condition on the date of exercise is equal to or better than Tenant's financial condition on the date of exercise to provide a Letter of Credit with terms and amounts acceptable to Landlord in its business judgment to secure Tenant's obligations during the applicable Option Term.

All terms, provisions, conditions and covenants of this Lease shall remain in full force and effect during the Option Terms, provided that Tenant shall have no additional option periods and the Base Rent payable during the first Lease Year of each Option Term (and for increases during the Option Term, as applicable) shall be the market rate then prevailing as projected for the commencement of the applicable Option Term, for premises comparable in size, quality and location in comparable class R&D/Office buildings throughout the Tri-Valley/Livermore area taking into account all relevant factors (the "**market rent**"). Base Rent for the Option Term shall be determined prior to the commencement of the applicable Option Term in the following manner:

If Landlord and Tenant are unable to agree on the market rent within sixty (60) days after Tenant gives notice of its exercise of the Option Term, then Tenant shall have the right to revoke its exercise of the option by delivering written notice within ten (10) days following the expiration of such 60-day period. In the event of such revocation, Tenant shall forfeit all rights to thereafter exercise any option under this Lease and the Lease shall terminate at the end of the initial term, or then Option Term, as applicable. If Tenant does not revoke its exercise and elects to proceed with the determination of market rent, then the monthly Base Rent and Additional Rent payable during the Option Term shall be determined by appraisal in the following manner:

If Landlord and Tenant can agree on a single appraiser, then the rate set by such appraiser as set forth below shall be the Base Rent for the Option Term. If the parties cannot agree on a single appraiser, then each party, by giving written notice to the other party, shall appoint as an appraiser an experienced commercial real estate agent in the area in which the Premises are located. Said appointment shall be made within ten (10) days following the expiration of the sixty (60) day period aforesaid, and if one of the parties does not appoint an appraiser within that time, the single appraiser named shall be the sole appraiser and shall set the monthly Base Rent for the Option Term.

If the two appraisers are appointed as provided herein, each shall independently prepare an estimate of the market rent within sixty (60) days. If the higher of the two estimates so determined is within ten percent (10%) of the lower estimate, then the monthly Base Rent to be paid by Tenant during the Option Term shall be the average of the amounts determined by the appraisers. If the difference between the two estimates exceeds ten percent (10%) of the lower one, the two appraisers shall select a third appraiser meeting the qualifications set forth hereinabove within ten (10) days thereafter who will likewise independently estimate the market rate within sixty (60) days after the appointment. The average of the two closest appraisals shall be set as the monthly Base Rent.

Each party shall pay the fees of the appraiser appointed by such party and the parties will share equally the fees of any third appraiser appointed pursuant to this Section A-2.1.

Notwithstanding the above, the Base Rent payable by Tenant during each Option Term shall be in addition to all Additional Rent and other sums and charges payable by Tenant under the terms of this Lease.

Tenant acknowledges that the options granted herein are personal to Tenant and may not be assigned with an assignment of this Lease except in connection with an assignment to an entity which controls, is controlled by or is under common control with Tenant (as defined in Article 20 of this Lease) or which is a successor to Tenant by merger, consolidation or sale of substantially all of Tenant's assets with Landlord's prior written consent, not to be unreasonably withheld.

1

A-4.7. Hazardous Substances. Landlord hereby represents that it has, prior to the date of this Lease, provided to Tenant copies of all environmental reports in its possession, regarding the presence of Hazardous Substances at the Center or upon, around or under Parcel 6. Except as specifically disclosed in the reports delivered to Tenant, Landlord represents and warrants that to its actual knowledge, Landlord does not know of any Hazardous Substances in the Center. Landlord shall indemnify, defend and hold Tenant harmless for any claims, costs or liabilities (collectively, "Claims") arising out of or relating to any breach or misrepresentation by Landlord of the foregoing representation and warranty. Landlord's confidentiality obligations under Section 4.7 and its indemnity obligations pursuant to this Section A-4.7 shall survive the termination of this Lease.

A-4.8 <u>Declaration</u>. Notwithstanding the provisions of Section 4.8, Landlord shall not amend the Declaration in a manner which (i) reduces the number of Tenant's exclusive parking spaces on Parcel 6, (ii) restricts Tenant 's permitted use described in Article 4, (iii) adversely and materially affects Tenant's access to or from Parcel 6 and Lawrence Road or South Front Road or (iv) increases the share of Common Area Costs assessed against Parcel 6 or Parcel 6's proportionate share of Shared Maintenance Costs, without the prior written consent of Tenant which shall not be unreasonably withheld or delayed.

A-6 <u>Utility Use Restrictions</u>. Tenant agrees that during the term of the Lease Tenant's use of the following Building 6 utilities shall not exceed the consumption limits listed below:

- 1. Electrical Power: 1200 amps.
- 2. Water: 40 gallons per minute
- 3. Industrial Waste Line: 20 gallons per minute

In the event that Tenant's use of any of the above utility services exceed the capacities above, and the operations of other Building 6 tenants are affected thereby, Landlord may require Tenant to take necessary steps acceptable to Landlord to increase the capacities of the Building 6 utilities affected at Tenant's sole cost and expense.

A-9.3 Repairs by Tenant. Notwithstanding the provisions of Section 9.4, except to the extent necessary due to damage caused by the negligence of Tenant, its employees, agents or contractors, Tenant shall have no obligation to replace the HVAC system or any other essential building system serving Building 6 (specifically excluding any special HVAC system for Tenant's operations in the Premises, such as the HVAC serving any "clean room", the replacement of which shall be at Tenant's sole cost and expense) within the last eighteen (18) months of the Term. If any such replacement is necessary, Landlord and Tenant shall mutually agree on the type of equipment to be installed and a commercially reasonable cost sharing arrangement which will take into account the number of years of the useful life of such equipment or system which will occur following the expiration of the Term. If Tenant subsequently exercises an option to extend the Lease, however, the replacement shall be at Tenant's sole option, cost and expense and within thirty (30) days after Tenant's exercise of the option, Tenant shall reimburse Landlord for all amounts previously paid by Landlord for the system replaced.

A-9.5. <u>Tenant Equipment/Improvements</u>. The equipment Tenant initially intends to install in the Premises is described on <u>Exhibit C</u> attached hereto. If Landlord wishes to require removal of any Tenant Improvements, Landlord shall designate as a part of its approval pursuant to the terms of <u>Exhibit C</u> of the plans for Tenant's Work, any Tenant Improvements and/or Special Tenant Improvements (if any) or equipment which Landlord will require Tenant to remove at the expiration of the Term. In connection with any such required removal by Tenant, Tenant shall repair all damage caused by such removal.

A-26.1. <u>Non-Disturbance Agreement</u>. Landlord shall use commercially reasonable efforts to obtain an agreement from Landlord's existing construction lender prior to the Delivery date to not disturb Tenant's possession under this Lease so long as Tenant is not in default of its obligations hereunder.

A-32.15. <u>Restriction on Sale</u>. Notwithstanding the provisions of Section 32.15 of the Lease, during the term of this Lease and provided that Tenant is not then in default of this Lease beyond any applicable cure period, Landlord shall not sell Parcel 6 or Building 6 to an entity on Tenant's competitor list which is attached hereto as <u>Exhibit I</u> without Tenant's prior written consent, which may be withheld in Tenant's sole discretion.

2

A-32.24. <u>Building Sale Notice Rights</u>. Landlord shall provide written notice to Tenant the first time Landlord responds in writing to a new interested third party to purchase Building 6, provided that Tenant is not then in default of this Lease beyond any applicable cure period. Landlord shall only be required to notify Tenant of third party interest <u>one time</u> with respect to the Building. Tenant shall have five (5) days to indicate its interest in negotiating a purchase of the Building. Landlord may negotiate concurrently with Tenant and interested third party(ies). Landlord's obligation to notify Tenant as described herein shall in no way obligate Landlord to sell Building 6 to Tenant. Tenant's notice rights shall expire upon Landlord's execution of a sale agreement with a third party. Tenant acknowledges that the sale of Building 6 will be subject to all existing leases affecting said Building.

A- 32.25. Parking. Tenant has been allocated 30 parking stalls adjacent to the Premises on Parcel 6. The aforementioned parking stalls shall be in addition to any parking stalls within Tenant's Facility Yard. Landlord hereby acknowledges that Tenant, subject to applicable codes, may designate additional areas within Tenant's Facility Yard for parking and/or parking stalls as Tenant sees fit. Tenant is also leasing from Landlord the buildings designated as "Building 1", "Building 2" and "Building 3" on Exhibit A. So long as Tenant's lease of Building 1 is in effect, Tenant may use a portion of the parking spaces on Parcel 1 in connection with its use of Building 6; so long as the Tenant's lease of Building 3 is in effect, Tenant may use a portion of the parking spaces on Parcel 3 in connection with its use of Building 6.

A-32.26 Signage. All of Tenant's signage at the Premises and Parcel 6 must be in accordance with the City-approved master sign program for the Center. The program provides 2' x 16' signage areas at each entry structure and a 2'6" x 5'0" signage area on a monument at the street in front of each building. Notwithstanding the foregoing, Landlord shall allow Tenant to install the same signage (i.e., the same size, style, shape and type) on the exterior walls of the Premises as Tenant has installed on the exterior of Buildings 1, 2 and 3. Tenant's corporate logo and tradestyle are permitted to be used in accordance with the parameters of the sign program. Any additional signage outside the scope of the master signage program shall be subject to the approval of the Landlord (which shall not be unreasonably withheld) and the City of Livermore. Subject to City and Landlord's approval, Landlord shall permit Tenant to install a temporary sign or banner in the Center, in a location approved by Landlord, announcing the Center as Tenant's new headquarters location.

A-32.27 <u>Use of Roof</u>. Tenant acknowledges that Landlord has reserved the right to use the roof of Building 6, including the right to lease or license its use. Tenant and no employee or invitee of Tenant shall go upon the roof of the Building, except as otherwise expressly provided herein.

Tenant shall have the exclusive right to use that portion of the roof directly above the Premises, to install such mechanical equipment as may be reasonably necessary to serve the Premises as part of the Tenant Improvements approved by Landlord ("**Mechanical Equipment**"), and may make such penetrations of the roof as may be reasonably necessary in connection therewith. Tenant also shall have the exclusive right to use a portion of the roof directly above the Premises, to install a satellite dish or cluster of dishes and ancillary telecommunications equipment in connection with Tenant's business operations. Tenant's roof use shall be on the following terms and conditions set forth herein. Subject to Applicable Laws, Tenant shall have the right to install or cause to be installed rooftop equipment ("**Rooftop Equipment**") pursuant to plans and specifications which shall be subject to Landlord's prior written approval, which shall not be unreasonably withheld or delayed on the roof of the Building, in a location as Landlord and Tenant may mutually agree. There shall be no additional charge payable by Tenant to Landlord for the use of such area or for the installation of the Mechanical Equipment and Rooftop Equipment. If the Rooftop Equipment is to be installed on the roof. Tenant shall notify Landlord in writing that the Rooftop Equipment is to be installed on the roof. Tenant shall be solely responsible for complying with (or causing its vendor to comply with) the requirements of such roof warranty or roof bond in connection with the installation, maintenance, repair, replacement or removal of the Mechanical Equipment and Rooftop Equipment. Landlord

shall permit Tenant reasonable access to the designated area as reasonably necessary to install, maintain and remove the Mechanical Equipment and Rooftop Equipment, and Tenant shall indemnify Landlord and be solely responsible, at Tenant's cost and expense, for the maintenance and repair of the Mechanical Equipment and Rooftop Equipment, and Landlord shall have no responsibility with respect thereto unless the same

3

was made necessary by the negligence or willful act of Landlord or Landlord's Agents. Tenant hereby agrees to defend, indemnify and hold Landlord harmless form any mechanics or materialmen's liens upon the Premises or the Center which result from work associated with the installation of the Rooftop Equipment. Tenant shall obtain all licenses or approvals required to install and operate the Rooftop Equipment. The Rooftop Equipment shall remain the property of Tenant and upon expiration of the Lease, Tenant shall remove the Rooftop Equipment and repair the Premises and any damage to the area upon which the Rooftop Equipment was located to the original condition, normal wear and tear excepted. Landlord shall have the right to request that Tenant relocate the Rooftop Equipment, if necessary, at Landlord's sole cost and expense to facilitate Landlord's use of the roof. Tenant covenants that the Rooftop Equipment will be installed, maintained and removed in accordance with all Applicable Requirements. Tenant shall be responsible for all damage caused by the installation, maintenance, repair and/or removal of Tenant's Rooftop Equipment. Tenant's access to the roof to exercise its rights hereunder shall be subject to Landlord's prior approval, which shall not be unreasonably withheld, provided that Tenant exercises such access rights in a manner that does not void any roof warranty. Tenant's Rooftop Equipment shall not interfere with the operation of any existing roof top equipment which has been installed on the portion of the roof used by Landlord. Landlord shall not install or permit the installation of any rooftop equipment which will interfere with any Rooftop Equipment for which Tenant has submitted installation plans to Landlord or which Tenant has previously installed on the portion of the roof for Tenant's use.

4

EXHIBIT C

WORK LETTER [Work Letter]

This Work Letter sets forth the terms and conditions relating to the construction of the Premises.

SECTION 1

BUILDING AND THE PREMISES

1.1 <u>Base Building.</u> The Base Building has been constructed in accordance with the plans for such improvements listed on the plan list attached as Schedule 1 to this Exhibit C (the "Plan List") as modified by the "as-built" plan set which have been provided to Tenant. The Base Building shall include, without limitation:

a) Fully enclosed tilt-up concrete building(s) with 5" thick concrete slab and grade doors as shown on the construction drawings listed in the Plan List;

- b) Water and gas service stubbed into the Building;
- c) A sanitary sewer gut line as shown on the construction drawings;
- Four (4) 4" telephone conduits and 8' x 8' plywood terminal board in the electrical room;
 Two (2) of the four conduits shall be dedicated to Tenant for connectivity to Buildings 1, 2 and 3
- e) Fire sprinklers at roof to meet Legal Requirements for the Building shell.
- f) A 2500 amp, 480/208, 3 phase electrical service for use by all Building 6 tenants, from which Tenant may install one (1) 1600 amp breaker and meter. This service shall be installed in such a manner as to permit the installation of future electrical services for future tenants of Building 6 within the space provided for tenant electrical service within the electrical room.

Tenant acknowledges that the Base Building and Parcel 6 Site Work have been completed and agrees to accept possession of the Premises "as-is" in its existing condition except for Landlord's Work.

1.2 <u>Special Tenant Improvements.</u> Landlord shall install and construct the following improvements at its sole cost and expense (except as provided below):

- Replace the existing one-inch (1") water meter for Building 6 with a one and one-half inch (1 ½") meter (Tenant agrees to pay to Landlord 50% of the cost of such work)
- Modifications to the Common Areas as necessary to provide for adequate nitrogen delivery truck access to the Premises, including without limitation modifying curbs, sidewalks and paving.

1.3 Landlord's Work. "Landlord's Work" shall mean all work constructed and to be constructed by Landlord described in Sections 1.1 and 1.2 above.

1.4 <u>Tenant's Work</u>. **"Tenant's Work**" will include designing, providing and installing all Tenant Improvements (defined hereafter) and providing the required furnishings, fixtures and equipment for Tenant's use of the Premises. As used in this Lease, the term "**Tenant Improvements**" shall mean all

improvements set forth in the Approved Tenant Improvement Plans. Tenant shall obtain all necessary permits and approvals for and shall construct the "**Tenant Improvements**" in accordance with the Approved Tenant Improvement Plans (as defined in Section 2.4 below). Landlord will disburse the Tenant Improvement Allowance described in Section 4 below to pay for Tenant Improvement Costs (defined hereafter). All such costs of completing Tenant's Work which are in excess of the Tenant Improvement Allowance shall be paid by Tenant pursuant to the terms of the Lease and this Work Letter.

SECTION 2

TENANT IMPROVEMENT PLANS

2.1 <u>Architect/Construction Plans</u>. Tenant has retained CAS Architects, Inc. (the "Architect") to prepare the construction plans for all Tenant Improvements and Special Tenant Improvements. Tenant shall retain engineering consultants (the "Engineers") approved by Landlord (which approval shall not be unreasonably

C-1

withheld or delayed) to prepare all plans and engineering working drawings relating to the Tenant Improvements, including without limitation, all structural, HVAC, electrical, plumbing, life safety, and sprinkler work in the Premises which is not part of the Base Building. The final working plans and drawings to be prepared by Architect and the Engineers hereunder shall be known collectively as the "Tenant Improvement Plans". The scope, form and content of all plans and drawings shall be discussed in reasonable detail at each of the weekly meetings held pursuant to the terms of Section 3.2.6 below. All Tenant Improvement Plans shall be in a form suitable for bidding and construction by qualified contractors, shall meet the requirements of the City of Livermore, and shall be subject to Landlord's approval, which shall not be unreasonably withheld or delayed. Tenant and Architect shall verify, in the field, the dimensions and conditions as shown on the relevant portions of the Base Building plans, and Tenant and Architect shall be solely responsible for the same, and Landlord shall have no responsibility in connection therewith. Landlord's review of the Tenant Improvement Plans as set forth in this Section 2, shall be for its sole purpose and shall not obligate Landlord or its architect, engineers and consultants, and notwithstanding any advice or assistance which may be rendered to Tenant by Landlord or Landlord's space planner, architect, engineers, and consultants, Landlord shall have no liability whatsoever in connection therewith and shall not be responsible for any omissions or errors contained in the Tenant Improvement Plans, and Tenant's waiver and indemnity set forth in Section 16 of this Lease shall specifically apply to the Tenant Improvement Plans.

2.2 <u>Final Design Drawings</u>. Tenant and the Architect shall prepare the final design drawings and specifications for Tenant Improvements in the Premises (collectively, the "Final Design Drawings") and shall deliver the same to Landlord for Landlord's approval. The Final Design Drawings shall include a layout and designation of all offices, rooms and other partitioning. Landlord may request clarification or more specific drawings for special use items not included in the Final Design Drawings. Landlord shall advise Tenant within five (5) business days after Landlord's receipt of the Final Design Drawings for the Premises if the same are unsatisfactory or incomplete in any respect. If Tenant is so advised, Tenant shall promptly cause the Final Design Drawings to be revised to correct any deficiencies or other matters Landlord may reasonably require.

2.3 <u>Final Working Drawings</u>. After the Final Design Drawings have been approved by Landlord, Tenant shall promptly cause the Architect and the Engineers to complete the architectural and engineering drawings for the Premises, and Architect shall compile a fully coordinated set of architectural, structural, mechanical, electrical and plumbing working drawings in a form which is sufficiently complete to allow subcontractors with a reasonable level of competence and experience in the industry to bid on the work and to obtain all permits required for the construction of the Tenant Improvements (the "**Permits**") and shall submit the same (collectively, the "**Final Working Drawings**") to Landlord for Landlord's approval, which approval shall not be unreasonably withheld or delayed. Tenant shall supply Landlord with three (3) copies signed by Tenant of such Final Working Drawings. Landlord shall advise Tenant within five (5) business days after Landlord's receipt of the Final Working Drawings for the Premises if the same are unsatisfactory or incomplete in any respect. If Tenant is so advised, Tenant shall promptly cause the Final Working Drawings to be revised in accordance with such review and any disapproval of Landlord in connection therewith.

2.4 <u>Approved Tenant Improvement Plans</u>. The Final Working Drawings shall be approved by Landlord (the "Approved Tenant Improvement Plans") prior to the commencement of construction of the Tenant Improvements by Tenant. In order to expedite the permitting process, however, prior to Landlord's approval pursuant to Section 2.3 above, Tenant may submit the Final Working Drawings to the appropriate municipal authorities for all Permits necessary to allow Tenant's contractor to commence and fully complete the construction of the Tenant Improvements. Notwithstanding the foregoing, Tenant acknowledges that Landlord does not waive the right to approve the Final Working Drawings and by electing to submit the Final Working Drawings for permit prior to Landlord's approval, Tenant is assuming the risk that Landlord may require changes in such drawings after the same have been submitted for permits. Tenant hereby agrees that neither Landlord nor Landlord's consultants shall be responsible for obtaining any of the Permits or a certificate of occupancy for the Premises and that obtaining the same shall be Tenant's responsibility; provided, however, that Landlord shall cooperate with Tenant in executing permit applications and performing other ministerial acts reasonably necessary to enable Tenant to obtain any such permit or certificate of occupancy. No changes, modifications or alterations in the Approved Tenant Improvement Plans may be made without the prior written consent of Landlord, which consent shall not be unreasonably withheld or delayed, provided that if a proposed change would directly or indirectly delay the "**Substantial Completion**" of Landlord's Work as that term is defined in Article 8 of the Lease, Landlord may proceed to establish a Tenant Delay pursuant to Section 5 hereof.

C-2

SECTION 3

CONSTRUCTION OF TENANT IMPROVEMENTS

3.1.1 <u>Tenant's Contractor</u>. A general contractor shall be retained by Tenant to construct the Tenant Improvements. Such general contractor ("**Tenant's Contractor**") shall be subject to Landlord's prior approval, which approval shall not be unreasonably withheld or delayed.

3.1.2 <u>Tenant's Agents</u>. All subcontractors, laborers, materialmen, and suppliers used by Tenant (such subcontractors, laborers, materialmen, and suppliers, and Tenant's Contractor to be known collectively as "**Tenant's Agents**") must be approved in writing by Landlord, which approval shall not be unreasonably withheld or delayed. If Landlord does not approve any of Tenant's proposed subcontractors, laborers, materialmen or suppliers, Tenant shall submit other proposed subcontractors, laborers, materialmen or suppliers for Landlord's written approval. Notwithstanding the foregoing, Tenant shall retain subcontractors designated or reasonably approved by Landlord in connection with any structural, mechanical, electrical, plumbing or heating, air-conditioning or ventilation work to be performed in the Premises. Further Landlord's approval shall not be required for any of Tenant's Agents performing work or providing materials costing less than \$10,000.

3.2 Construction of Tenant Improvements by Tenant's Agents.

3.2.1 <u>Tenant's Construction Contract; Cost Budget</u>. Prior to Tenant's execution of the construction contract and general conditions with Tenant's Contractor ("**Tenant's Construction Contract**"), Tenant shall submit Tenant's Construction Contract to Landlord for its approval, which approval shall not be unreasonably withheld or delayed. Prior to the commencement of the construction of the Tenant Improvements, and after Tenant has accepted all bids for the Tenant Improvements, Tenant shall provide Landlord with a detailed breakdown, by trade, of the final costs to be incurred or which have been incurred in connection with the design and construction Contract (the "Final Costs"). If the costs of the Tenant Improvements exceed the Tenant Improvement Allowance, Tenant shall also provide Landlord with its computation of percentage that the Tenant Improvement Allowance bears to the total costs of the Tenant Improvements (the "**Allowance Percentage**")

3.2.2 <u>Tenant's Agents</u>.

A. Landlord's General Terms for Tenant's Agents and Tenant Improvement Work. Tenant's and Tenant's Agent's construction of the Tenant Improvements shall comply with the following: (i) the Tenant Improvements shall be constructed in accordance with the Approved Tenant Improvement Plans; (ii) Tenant's Agents shall submit schedules of all work relating to the Tenant Improvements to Tenant's Contractor and Tenant's Agents shall adhere to such corrected schedule; and (iii) Tenant shall abide by all rules made by Landlord with respect to storage of materials, coordination of work with the contractors of other tenants and Landlord, and any other matter in connection with this Work Letter, including, without limitation, the construction of the Tenant Improvements.

B. Indemnity. Tenant's indemnity of Landlord as set forth in Section 16.1 of this Lease shall also apply with respect to any and all costs, losses, damages, injuries and liabilities related in any way to any act or omission of Tenant or Tenant's Agents, or anyone directly or indirectly employed by any of them, or in connection with Tenant's non-payment of any amount arising out of the Tenant Improvements and/or Tenant's disapproval of all or any portion of any request for payment. Such indemnity by Tenant, as set forth in Section 16.1 of this Lease, shall also apply with respect to any and all costs, losses, damages, injuries and liabilities related in any way to Landlord's performance of any ministerial acts reasonably necessary (i) to permit Tenant to complete the Tenant Improvements, and (ii) to enable Tenant to obtain any building permit or certificate of occupancy for the Premises.

C. <u>Requirements of Tenant's Agents</u>. Each of Tenant's Agents shall guarantee to Tenant and for the benefit of Landlord that the portion of the Tenant Improvements for which it is responsible shall be free from any defects in workmanship and materials for a period of not less than one (1) year from the date of completion thereof. Each of Tenant's Agents shall be responsible for the replacement or repair, without additional charge, of all work done or furnished in accordance with its contract that shall become defective within one (1) year after the later to occur of (i) completion of the work performed by such contractor or subcontractors and (ii) the Lease Commencement Date. The correction of such work shall include, without additional charge, all additional expenses and damages incurred in connection with such removal or replacement of all or any part of the Tenant Improvements, and/or the Building and/or common areas that may be damaged or disturbed thereby. All such warranties or guarantees as to materials or workmanship of or with respect to the Tenant Improvements shall be contained in the Contract or subcontract and shall be written such that such guarantees or warranties shall incure to the benefit of both Landlord and Tenant, as their respective interests may appear, and can be directly enforced by either. Tenant covenants to give to Landlord any assignment or other assurances which may be necessary to effect such right of direct enforcement.

C-3

D. <u>Insurance Requirements</u>.

1. <u>General Coverages</u>. All of Tenant's Agents shall carry worker's compensation insurance covering all of their respective employees, and shall also carry public liability insurance, including property damage, all with limits, in form and with companies as are required to be carried by Tenant as set forth in <u>Article 14</u> of this Lease.

2. <u>Special Coverages</u>. Tenant shall carry "Builder's All Risk" insurance in an amount approved by Landlord covering the construction of the Tenant Improvements, and such other insurance as Landlord may require, it being understood and agreed that the Tenant Improvements shall be insured by Landlord pursuant to <u>Article 14</u> of this Lease upon completion thereof. Tenant's policy of Builder's All Risk insurance shall be in amounts and shall include such extended coverage endorsements as may be reasonably required by Landlord including, but not limited to, the requirement that all of Tenant's Agents shall carry Excess Liability and Products and Completed Operation Coverage insurance, each in amounts not less than \$1,000,000 per incident, \$2,000,000 in aggregate, and in form and with companies as are required to be carried by Tenant as set forth in <u>Article 14</u> of this Lease.

3. <u>General Terms</u>. Certificates for all insurance carried pursuant to this Section 3.2.2(D) shall be delivered to Landlord before the commencement of construction of the Tenant Improvements and before the equipment of Tenant's Contractor is moved onto the site. All such policies of insurance must contain a provision that the company writing said policy will give Landlord thirty (30) days' prior written notice of any cancellation or lapse of the effective date or any reduction in the amounts of such insurance. In the event that the Tenant Improvements are damaged by any cause during the course of the construction thereof, Tenant shall immediately repair the same at Tenant's sole cost and expense. Tenant's Agents shall maintain all of the foregoing insurance coverage in force until the Tenant Improvements are fully completed and accepted by Landlord, except for any Products and Completed Operation Coverage insurance required by Landlord, which is to be maintained for ten (10) years following completion of the work and acceptance by Landlord and Tenant. All policies carried under this Section 3.2.2(D) shall insure Landlord and Tenant, as their interests may appear, as well as Tenant's Contractor and Tenant's Agents. All insurance, except Workers' Compensation, maintained by Tenant's Agents shall preclude subrogation claims by the insurer against anyone insured thereunder. Such insurance shall provide that it is primary insurance as respects the Landlord and that any other insurance maintained by Landlord is excess and noncontributing with the insurance required hereunder. The requirements for the foregoing insurance shall not derogate from the provisions for indemnification of Landlord by Tenant under Section 3.2.2(B) of this Work Letter.

3.2.3 <u>Governmental Compliance</u>. The Tenant Improvements shall comply in all respects with the following: (i) all building codes and other state, federal, city or quasi-governmental laws, codes, ordinances and regulations, as each may apply according to the rulings of the controlling public official, agent or other person; (ii) applicable standards of the American Insurance Association (formerly, the National Board of Fire Underwriters) and the National Electrical Code; and (iii) building material manufacturer's specifications.

3.2.4 <u>Mechanic's Liens</u>. Tenant shall keep the Premises and the Center free and clear of mechanics' liens arising out of its construction of Tenant's Work as provided in Article 29. All provisions of Article 29 shall apply to Tenant's Work as if fully set forth into this Work Letter.

3.2.5 Inspection by Landlord. Landlord shall have the right to inspect the Tenant Improvements at all times, provided however, that Landlord's failure to inspect the Tenant Improvements shall in no event constitute a waiver of any of Landlord's rights hereunder nor shall Landlord's inspection of the Tenant Improvements constitute Landlord's approval of the same. Should Landlord reasonably disapprove any portion of the Tenant Improvements, Landlord shall specify the items disapproved. Any material defects or deviations in, and/or disapproval by Landlord of, the Tenant Improvements shall be rectified by Tenant at no expense to Landlord, provided however, that in the event Landlord determines that a defect or deviation exists or disapproves of any matter in connection with any portion of the Tenant Improvements and such defect, deviation or matter might adversely affect the mechanical, electrical, plumbing, heating, ventilating and air-conditioning or life-safety systems of the Building, the structure or exterior appearance of the Building or any other tenant's use of such other tenant's leased premises, Landlord after giving Tenant at least three (3) business days advance written notice and an opportunity to cure or otherwise demonstrate compliance, may take such action as Landlord deems necessary, at Tenant's expense and without incurring any liability on Landlord's part, to correct any such defect, deviation and/or matter is corrected to Landlord's satisfaction.

3.2.6 <u>Meetings</u>. Commencing upon the execution of this Lease, Tenant shall hold weekly meetings at a reasonable time, with the Architect and the Tenant's Contractor regarding the progress of the preparation of the Tenant Improvement Plans and the construction of the Tenant Improvements, and Landlord and/or its agents shall receive prior notice of, and shall have the right to attend, all such meetings, and, upon Landlord's request, certain of Tenant's Agents shall attend such

C-4

meetings. Such meetings shall include a detailed review of the plans, drawings and specifications prepared to date and all participants in the meeting shall make a good faith effort to raise any issues or concerns they may have regarding the scope, form or content of any plan submitted.

3.3 <u>Notice of Completion; Copy of Record Set of Plans</u>. Within ten (10) days after completion of construction of the Tenant Improvements, Tenant shall cause a Notice of Completion with respect to the Tenant Improvements to be recorded in the office of the Recorder of the County of Alameda in accordance with Section 3093 of the Civil Code of the State of California or any successor statute, and shall furnish a copy thereof to Landlord upon such recordation. If Tenant fails to do so, Landlord may execute and file the same on behalf of Tenant as Tenant's agent for such purpose, at Tenant's sole cost and expense. At the conclusion of construction, (i) Tenant shall cause the Architect and Tenant's Contractor (A) to update the Approved Working Drawings as necessary to reflect all changes made to the Approved Working Drawings during the course of construction, (B) to certify to the best of their knowledge that the "record-set" of mylar as-built drawings are true and correct, which certification shall survive the expiration or termination of this Lease, and (C) to deliver to Landlord two (2) sets of copies of such record set of drawings within ninety (90) days following issuance of a certificate of occupancy for the Premises, and (ii) Tenant shall deliver to Landlord a copy of all warranties, guaranties, and operating manuals and information relating to the improvements, equipment, and systems in the Premises.

SECTION 4

TENANT IMPROVEMENT ALLOWANCE; CONSTRUCTION COSTS

4.1 <u>Tenant Improvement Allowance</u>. Tenant shall be entitled to a one-time tenant improvement allowance (the "Tenant Improvement Allowance") in the amount of \$25.00 per square foot of the Premises GLA for the costs relating to the initial design and construction of the Tenant Improvements. In no event shall Landlord be obligated to make disbursements pursuant to this Work Letter in a total amount which exceeds the Tenant Improvement Allowance. Tenant initially desired to lease premises at the east end of Building 6 which were closer to the existing sewer and water facilities necessary for Tenant's use. Landlord acknowledges that Tenant's agreement to lease the Premises, which are located at the west end of Building 6, will increase the costs to Tenant of bringing such utilities to the Premises. Therefore, in addition to the Tenant Improvement Allowance, Landlord agrees to pay 50% of the following costs which Tenant would not have incurred had Tenant's Premises been located at the east end of Building 6: (i) Tenant's cost of running Tenant's Industrial Waste Line an additional 244 linear feet, and (ii) Tenant's cost of running Tenant's Water Supply Line an additional 267 linear feet. Landlord shall pay its share of the foregoing costs as they are incurred, in accordance with Section 4.2 below.

4.2 Disbursement of the Tenant Improvement Allowance.

4.2.1 <u>Tenant Improvement Allowance Items</u>. Except as otherwise set forth in this Work Letter, the Tenant Improvement Allowance shall be disbursed by Landlord only for the following items and costs (collectively the "Tenant Improvement Allowance Items"):

A. Payment of the fees of the Architect and the Engineers

B. The payment of plan check, permit and license fees relating to construction of the Tenant Improvements;

C. The cost of construction of the Tenant Improvements, including, without limitation, testing and inspection costs, utility usage, trash removal costs, and contractors' fees and general conditions;

D. The cost of any changes in the Base Building when such changes are required by the Tenant Improvement Plans, such cost to include all direct architectural and/or engineering fees and expenses incurred in connection therewith;

E. The cost of any changes to the Tenant Improvement Plans or Tenant Improvements required by Code; and

F. The payment of Tenant's 50% share of the costs of the new 1 ½" water meter for Building 6 described in Section 1.2 above.

4.2.2 <u>Disbursement of Tenant Improvement Allowance</u>. During the construction of the Tenant Improvements, Landlord shall make monthly disbursements of the Tenant Improvement Allowance for Tenant Improvement Allowance Items:

A. <u>Monthly Disbursements</u>. No more frequently than monthly, during the construction of the Tenant Improvements (or such other date as Landlord may designate), Tenant shall deliver to Landlord: (i) a request for payment of Tenant's Contractor, approved by Tenant, in a form to be provided by Landlord, showing the schedule, by trade, of percentage of completion of the Tenant Improvements in the Premises, detailing the portion of the work completed and the portion not completed; (ii) invoices from all of Tenant's Agents, for labor rendered and materials delivered to the Premises; (iii) executed mechanic's lien releases from all of Tenant's Agents which shall comply with the appropriate provisions, as reasonably determined by Landlord, of California Civil Code Section 3262(d); and (iv) all other information reasonably requested by Landlord. Tenant's request for payment shall be deemed Tenant's approval of the work furnished and/or the materials supplied as set forth in Tenant's payment request. After receipt of the foregoing and provided that Landlord does not dispute any request for payment based on non-compliance of any work with the Approved Tenant Improvement Plans, or due to any substandard work, or for any other reason, Landlord shall deliver a check to Tenant in an amount equal to the lesser of: (a) the Allowance Percentage of the Tenant Improvement costs for which payment is so requested less a ten percent (10%) retention (the aggregate amount of such retentions to be known as the "Final Retention"), or (b) the balance of the undisbursed Tenant Improvement Allowance (excluding the Final Retention). Landlord's payment request.

B. <u>Final Retention</u>. Subject to the provisions of this Work Letter, a check for the Final Retention shall be delivered by Landlord to Tenant within fifteen (15) days after all the following conditions have been satisfied: (i) Tenant delivers to Landlord a properly executed unconditional waiver and lien release from Tenant's Contractor in compliance with applicable sections of the California Civil Code, (ii) Landlord has determined that no substandard work exists, (iii) Architect delivers to Landlord a certificate, in a form reasonably acceptable to Landlord, certifying that the construction of the Tenant Improvements in the Premises has been substantially completed, (iv) Tenant has recorded a Notice of Completion with respect to Tenant's Work and thirty (30) days has expired from the date of such recording with no liens having been filed during such thirty day period, and (v) there is no existing Event of Default by Tenant under the Lease nor has Landlord notified Tenant of any default which would become an Event of Default under the Lease with the passage of time if not cured.

C. <u>Other Terms</u>. Landlord shall only be obligated to make disbursements from the Tenant Improvement Allowance to the extent costs are incurred by Tenant for Tenant Improvement Allowance Items. No disbursements shall be made to pay for Special Improvements unless the Tenant Improvement Allowance exceeds the costs of all other Tenant Improvements. All costs of completing Tenant's Work in excess of the Tenant Improvement Allowance shall be at Tenant's sole cost and expense.

D. Other Landlord Costs. Tenant shall also be responsible for the payment of (i) the fees incurred by Landlord for Landlord's consultants in connection with design drawing review and routine construction support related to the Tenant Improvements, (ii) the cost of documents and materials supplied by Landlord and Landlord's consultants, and (iii) all other verifiable, directly related costs, such as blueprint costs and delivery, fax and copy charges incurred by Landlord and Landlord's consultants related to the design/routine construction support of the Tenant Improvements. When the Tenant Improvement Plans have been approved by Landlord, Landlord shall submit to Tenant Landlord's estimate of the foregoing costs. The Tenant Improvement Allowance will not be used to pay the foregoing costs. Tenant shall pay such costs to Landlord from time to time within ten (10) days after receipt from Landlord of statements of such expenses.

SECTION 5

CONSTRUCTION DELAYS

5.1 <u>Tenant Delays</u>. As used in this Lease, the term **"Tenant Delay**" shall mean the period of an actual delay or delays in the Substantial Completion of Landlord's Work or in the occurrence of any of the other conditions precedent to the Delivery Date, as set forth in <u>Article 8</u> of the Lease, to the extent resulting from:

a. Tenant's failure to approve any matter requiring Tenant's approval within the time period specifically provided in this Work Letter for such approval;

b. A breach by Tenant of the terms of this Work Letter or the Lease;

c. Changes to the Base Building work described in the Plan List required by the Approved Tenant Improvement Drawings or requested in writing by Tenant; or

d. Any other acts or omissions of Tenant, or its agents, or employees,

(each a "Tenant Delay"). Landlord shall provide prompt (within 48 hours of becoming aware of any such delay) written notice to Tenant ("Delay Notice") specifying the action or inaction which Landlord contends constitutes a Tenant Delay hereunder. The period of delay, however, shall commence to run on the date of the action or inaction and not on the date of the Delay Notice. To the extent an action or inaction by Tenant specified in any Delay Notice constitutes a Tenant Delay as defined above and actually results in a delay in the Substantial Completion of the Premises (after taking into account any delays resulting from Landlord Delays and/or Force Majeure Delays described below), a Tenant Delay shall be deemed to have been established.

5.2 <u>Tenant's Lease Default</u>. Notwithstanding any provision to the contrary contained in this Lease: (i) if an Event of Default as described in Article 25 of the Lease has occurred or, (ii) a default by Tenant under this Work Letter has occurred at any time on or before the substantial completion of Landlord's Work and Tenant fails to remedy the default within such 48 hours after written notice from Landlord, then Landlord may thereafter: (i) in addition to all other rights and remedies granted to Landlord pursuant to the Lease, Landlord shall have the right to withhold payment of all or any portion of the Tenant Improvement Allowance and/or Landlord may cause Contractor to cease the construction of the Premises (in which case, Tenant shall be responsible for any Tenant Delay resulting from such work stoppage as set forth in Section 4.1 above of this Work Letter), and (ii) all other obligations of Landlord under the terms of this Work Letter shall be deferred until such time as such default is cured pursuant to the terms of the Lease.

5.3 Landlord Delay. As used herein, "Landlord Delay" shall mean: (i) any actual delay in the completion of the work Tenant is required to perform hereunder which results from any failure of Landlord to act or provide approvals within five (5) business days or (ii) the actual delay in the Substantial Completion of Landlord's Work due to any failure of Landlord, its agents, employees or contractors to perform the Base Building work or other work required to be provided by Landlord hereunder in compliance with the terms hereof and in compliance with applicable laws, rules and regulations or any other acts or omissions of Landlord, or its agents, or employees. Tenant shall provide prompt (within 48 hours of becoming aware of any such delay) written notice to Landlord ("Delay Notice") specifying the action or inaction which Tenant contends constitutes a Landlord Delay hereunder. The period of delay, however, shall commence to run on the date of the action or inaction and not on the date of the Delay Notice.

5.4 <u>Force Majeure Delays</u>. The term "**Force Majeure Delays**" shall mean delays caused by any event of force majeure described in Section 32.8 of the Lease.

5.5 <u>Substantial Completion</u>. The date set forth in Section 8.4 of the Lease for Landlord's Substantial Completion of the Base Building work shall be extended for the period of any Tenant Delays and Force Majeure Delays.

SECTION 6

MISCELLANEOUS

6.1 <u>Tenant's Representative</u>. Tenant has designated Ed Traverso and Alan Moyal as its sole representatives with respect to the matters set forth in this Work Letter, each of whom, until further notice to Landlord, shall have full authority and responsibility to act on behalf of the Tenant as required in this Work Letter.

6.2 <u>Landlord's Representative</u>. Landlord has designated William Drummond as its sole representative with respect to the matters set forth in this Work Letter, who, until further notice to Tenant, shall have full authority and responsibility to act on behalf of the Landlord as required in this Work Letter.

6.3 <u>Time of the Essence in This Work Letter</u>. Unless otherwise indicated, all references herein to a "number of days" shall mean and refer to calendar days. In all instances where Tenant is required to approve or deliver an item, if no written notice of approval is given or the item is not delivered within the stated time period, at Landlord's sole option, at the end of such period the item shall automatically be deemed approved or delivered by Tenant and the next succeeding time period shall commence.

C-7

FIRST AMENDMENT TO BUILDING 6 LEASE

THIS FIRST AMENDMENT TO BUILDING 6 LEASE (the "First Amendment") is made and entered into as of August 16, 2006 (the "First Amendment Effective Date") by and between Greenville Holding Company LLC a California limited liability company (hereafter, "Landlord"), and FormFactor, Inc., a Delaware corporation (hereafter "Tenant").

RECITALS

A. Greenville Investors L.P., a California limited partnership ("**Investors**") and Tenant entered into a Lease Agreement dated October 5, 2004 (the "**Lease**") relating to certain premises consisting of approximately 12,303 square feet located at the west end of Building 6 on Parcel 6 in the Pacific Corporate Center and "Tenant's Facility Yard" on Parcel 6 as described in said Lease (collectively, the "**Existing Premises**").

B. The Rent Commencement Date referred to in the Lease was March 15, 2005.

C. The interest of Investors in the Lease has been assigned to Landlord.

D. Tenant now desires to expand its Existing Premises to include the remainder of Building 6 consisting of 37,439 square feet of gross leasable area ("**GLA**") of additional space (the "**Additional Premises**") as depicted on the floor plan attached hereto as <u>Exhibit A</u>, and to construct improvements in the Additional Premises.

E. Landlord and Tenant desire to amend the Lease to provide for such expansion and construction on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants made herein, Landlord and Tenant agree as follows:

AGREEMENTS

1. **Premises**. Subject to the reservation set forth in the introductory Paragraph A of the Lease, Landlord hereby leases to Tenant and Tenant hereby takes from Landlord on the terms and conditions of the Lease, as amended hereby, the Additional Premises consisting of 37,439 square feet of GLA. From and after the First Amendment Effective Date, the term "Premises" as used in the Lease shall include both the Existing Premises and the Additional Premises aggregating 49,742 square feet. Commencing on the First Amendment Effective Date, all terms and provisions of the Lease, as amended hereby, specifically including the insurance and indemnity provisions, shall apply to both the Existing Premises and the Additional Premises; provided that, the monthly Base Rent payable by Tenant and Tenant's Proportionate Share of Operating and Maintenance Costs payable pursuant to Article 10 of the Lease shall not increase until the "Base Rent Adjustment Date", as defined in Paragraph 3 below.

A. Use. Landlord and Tenant hereby agree to expand the Use of the Premises as set forth in Paragraph 1(h) of the Lease to include facilities for research, development, design, manufacture (including with clean room facilities), and sale of components and devices relating to the testing and packaging of semiconductor devices. In addition, Section 4.7(B)(1) of the Lease is hereby amended by replacing the phrase in subparagraph (i): "i.e., testing of solid state electronic components and research and development related thereto" with the following: "i.e., research, development, design, manufacture (including with clean room facilities), and sale of semiconductor devices".

B. **Hazardous Substances**. Landlord and Tenant hereby agree to amend Section 4.7 of the Lease in its entirety to read as set forth on <u>Exhibit C</u> attached hereto, and to replace the list of Hazardous Substances set forth on <u>Exhibit F</u> to the Lease in its entirety with the replacement <u>Exhibit F</u> attached hereto. The parties agree that the <u>Exhibit F-1</u> referenced

1

in the replacement Exhibit F attached hereto shall refer to the existing Exhibit F-1 of the Lease.

C. **Utility Use Restrictions**. Landlord and Tenant hereby agree to delete the "Utility Use Restrictions" set forth in <u>Addendum A-6</u> to the Lease. Any increase in the existing utility capacities required by Tenant's use of Building 6 shall be made by Tenant at Tenant's sole cost and expense and shall be subject to Section 9.5 of the Lease.

D. Use of Roof. Landlord and Tenant hereby agree to amend Section A-32.27 of the Lease in its entirety to read as follows:

"A-32.27 <u>Use of Roof</u>. Landlord and Tenant agree that Tenant shall be permitted to use the roof subject to the following terms and conditions. Subject to Applicable Laws, Tenant shall have the right to install or cause to be installed mechanical equipment on the roof ("**Mechanical Equipment**"). With respect to any remaining portion of the roof on which Mechanical Equipment has not been installed by Tenant (the "**Remaining Roof**"), Landlord and Tenant agree that (i) Landlord shall have the exclusive right to use 50% of the Remaining Roof, and (ii) Tenant shall have the exclusive right to use 50% of the Remaining Roof to install a satellite dish or cluster of dishes and ancillary telecommunications equipment, or other equipment in connection with Tenant's business operations (collectively, the "**Rooftop Equipment**"), with the apportionment of the Remaining Roof to be reasonably acceptable to Landlord and Tenant.

There shall be no additional charge payable by Tenant to Landlord for the use of any portion of the roof or for the installation of any equipment on the roof.

Tenant agrees to install or cause to be installed the Rooftop Equipment pursuant to plans and specifications which shall be subject to Landlord's prior written approval, which shall not be unreasonably withheld or delayed, on the roof of the Building, in a location as Landlord and Tenant may mutually agree. If the Rooftop Equipment is to be installed on the roof, Tenant shall notify Landlord in writing that the Rooftop Equipment is to be installed on the roof. Tenant shall be solely responsible for complying with (or causing its vendor to comply with) the requirements of such roof warranty or roof bond in connection with the installation, maintenance, repair, replacement or removal of the Rooftop Equipment. Tenant shall repair any damage to the roof caused by the installation, maintenance, repair, replacement or removal of the Rooftop Equipment. Landlord shall permit Tenant reasonable access to the designated area as reasonably necessary to install, maintain and remove the Rooftop Equipment, and Tenant shall indemnify Landlord and be solely responsible, at Tenant's cost and expense, for the maintenance and repair of the Rooftop Equipment, and Landlord shall have no responsibility with respect thereto unless the same was made necessary by the negligence or willful act of Landlord or Landlord's Agents. Tenant hereby agrees to defend, indemnify and hold Landlord harmless form any mechanics or materialmen's liens upon the Premises or the Center which result from work associated with the installation of the Rooftop Equipment. Tenant shall obtain all licenses or approvals required to install and operate the Rooftop Equipment. The Rooftop Equipment shall remain the property of Tenant and upon expiration of the Lease, Tenant shall remove the Rooftop Equipment and repair the Premises and any damage to the area upon which the Rooftop Equipment was located to the original condition, normal wear and tear excepted. Tenant covenants that the Rooftop Equipment will be installed, maintained and removed in accordance with all Applicable Requirements. Tenant shall be responsible for all damage caused by the installation, maintenance, repair and/or removal of Tenant's Rooftop Equipment. Tenant's access to the roof to exercise its rights hereunder shall be subject to Landlord's prior approval, which shall not be unreasonably withheld, provided that Tenant exercises such access rights in a manner that does not void any roof warranty.

Landlord shall not install or permit the installation of any rooftop equipment on the Remaining Roof reserved for Landlord the operation of which will interfere with: (i) any Rooftop Equipment for which Tenant has submitted installation plans to Landlord or which Tenant has previously installed on the portion of the roof for Tenant's use, or (ii) the operations of the Tenant's business in Building 6 as conducted at the time of such installation. Prior to Landlord's installation of such rooftop equipment, Landlord shall provide to Tenant plans and specifications for Tenant's review. All installation work shall

be done in a manner to minimize the interference with Tenant's business operations. Further, to comply with Tenant's security requirements for the Building, Landlord shall coordinate any roof access with Tenant."

2. **Taxes**. Commencing on the First Amendment Effective Date and continuing throughout the Term of the Lease, Tenant's Proportionate Share of Taxes payable pursuant to Section 7.1 of the Lease shall be 100%. On the first day of the month after the First Amendment Effective Date, Tenant's Estimated Monthly Tax Payment shall increase to \$7,636.00.

3. **Base Rent Adjustment Date**. Commencing on the date (the "**Base Rent Adjustment Date**") which is the earlier of July 15, 2007 or the date Tenant has installed its equipment at the Additional Premises for the purpose of manufacturing its products: (i) the Monthly Base Rent in the amount of \$9,596.34 currently payable under the Lease shall increase to \$36,809.08 for the Existing Premises and the Additional Premises; and (ii) Tenant's Proportionate Share of Operating and Maintenance Costs shall be 100%.

4. **Base Rent Increases.** Pursuant to the provisions of Section 3.2 of the Lease, the monthly Base Rent increases by 4% on March 31 of each Lease Year during the Term of the Lease. The next increase of the Monthly Base Rent for the Existing Premises shall occur on March 31, 2007. The parties agree that commencing on March 31, 2008 and continuing on each March 31 during the Term of the Lease thereafter, the monthly Base Rent, as adjusted to include the Additional Premises pursuant to Paragraph 3 above, shall be increased by 4% pursuant to the provisions of Section 3.2.

5. **Parking.** Tenant acknowledges that a tenant of Building 4 has the right to use 10 parking spaces on Parcel 6 the parcel upon which Building 6 is located, in the location depicted on <u>Exhibit D</u> attached hereto. During the term of the Lease, Tenant shall have the right to use all of the remaining 121 existing parking spaces on Parcel 6. Landlord and Tenant acknowledge that Tenant's parking will be reduced by approximately 12 spaces if Tenant uses a portion of its existing spaces to expand Tenant's facility yard on Parcel 6, which will result in a total of 109 parking spaces available to Tenant.

6. **Landlord's Work**. Landlord agrees to complete Landlord's Work described in the Work Letter (Additional Premises) attached hereto as Exhibit B (the "**Work Letter (Additional Premises)**") prior to the Base Rent Adjustment Date.

7. **Approved Tenant Improvement Plans.** The Tenant Improvements for the Additional Premises shall be set forth in plans to be attached hereto as Exhibit B-1 once such plans have been prepared by Tenant and approved by Landlord. Once attached, the plans shall be deemed to be the "Approved Tenant Improvement Plans" referred to in Section 2.4 of the Work Letter (Additional Premises).

8. **Work Letter.** The Work Letter attached as <u>Exhibit C</u> to the Lease shall apply only to the Existing Premises. Landlord's Work and Tenant's Work with respect to the Additional Premises shall be constructed in accordance with the terms and conditions set forth in the Work Letter (Additional Premises). The Work Letter (Additional Premises) provides for a Tenant Improvement Allowance of (\$935,975.00) (\$25 psf of the Additional Premises),

9. **Term.** The parties hereby agree to extend the initial term of the Lease described in Section 2.1 of the Lease. The termination date of the initial term shall be extended to midnight on the fifteenth anniversary of the First Amendment Effective Date. The references to "Term" in the Lease shall include both the initial term and the two "Option Terms" described in Paragraph 10 below.

10. **Options to Renew.** The first sentence of Section A-2.1 of the Addendum to Lease is amended to read as follows: "Provided that no Event of Default by Tenant under this Lease exists as of the date of exercise of the applicable option or at the expiration of the initial term or preceding Option Term, and provided further that Tenant has not assigned this Lease, Tenant shall have the option to extend the initial lease term for two (2) additional, successive terms of five (5) years each (each, an **"Option Term**")."

11. Building Sale or Lease Notice Rights. Landlord shall provide written

3

notice to Tenant the first time Landlord responds in writing to a new interested third party to purchase or lease Building 4, 5 and/or 7, provided that Tenant is not then in default of this Lease beyond any applicable cure period. Tenant shall have five (5) days to indicate its interest in negotiating a sale or lease. Landlord may negotiate concurrently with Tenant and interested third party(ies). Landlord's obligation to notify Tenant as described herein shall in no way obligate Landlord to sell or lease Building 4, 5 or 7 to Tenant. Tenant's notice rights with respect to a Building shall expire upon Landlord's sale of the Building or execution of a lease agreement with a third party for premises in the Building.

The parties agree that with respect to any building sale or lease notice rights and obligations for Building 7 that are set forth in Addendums A-32.24 and A-32.25 of that certain Pacific Corporate Center Lease dated as of May 3, 2001 by and between Landlord and Tenant (as amended or supplemented from time to time, the "Building 1 Lease"), this Section 11 shall supersede such rights and obligations for Building 7 in Addendums A-32.24 and A-32.25 of the Building 1 Lease. Except as otherwise expressly provided in the sentence above, from and after the Effective Date of this First Amendment, all provisions of the Building 1 Lease shall remain unchanged and in full force and effect.

12. **Restriction on Sale.** Landlord and Tenant hereby agree to replace the Tenant's competitor list set forth on <u>Exhibit I</u> to the Lease in its entirety with the competitor list set forth on the replacement <u>Exhibit I</u> attached hereto.

[remainder of this page intentionally left blank]

13. **Effectiveness of Lease.** Except as otherwise expressly provided herein, from and after the Effective Date of this First Amendment, all provisions of the Lease shall remain unchanged and in full force and effect and all references to the "Lease" shall mean the Lease as amended by this First Amendment. All capitalized terms used in this First Amendment which are not defined shall have the same definition as those same terms in the Lease.

IN WITNESS WHEREOF, the parties hereto have executed this First Amendment to Building 6 Lease as of the day and year first written above.

LANDLORD:		TENANT:		
GREENVILLE HOLDING COMPANY, LLC a Delaware limited liability company		FORMFACTOR, INC., a Delaware corporation		
By: Greenville Investors II LLC, a Delaware limited liability company (dba in California as Greenville Pacific Corporate Center LLC) Its: Managing Member		By: Its:	/s/ STUART L. MERKA Stuart L. Merkadeau SVP, General Counsel &	
By: Greenville Investors LP, a California limited partnershipIts: Managing Member				
By: Greenville Ventures, Inc., a California corporationIts: General Partner				
By: /s/ JEFFREY W. ABRAMSON Jeffrey W. Abramson Its: President				
ADDRESS:	675 Hartz Avenue	ADI		7005 Southfront Road
	Danville, CA 94526			Livermore, CA 94551
Dated: 8-16-06		Date	d: <u>16 August 2006</u>	

5

EXHIBIT A

Floor Plan of Additional Premises

[Floor Plan of Additional Premises]

1

EXHIBIT B

WORK LETTER (ADDITIONAL PREMISES)

[Work Letter – Additional Premises]

This Work Letter sets forth the terms and conditions relating to the construction of the Premises.

SECTION 1

BUILDING AND THE PREMISES

1.1 <u>Base Building</u>. The Base Building has been constructed in accordance with the plans for such improvements listed on the plan list attached as Schedule 1 to this Exhibit B (the "Plan List") as modified by the "as-built" plan set which has been provided to Tenant.

Tenant acknowledges that the Base Building and Parcel 6 Site Work have been completed and agrees to accept possession of the Premises "as-is" in its existing condition except for Landlord's Work.

1.2 <u>Landlord's Work</u>. Landlord shall install, construct, repair and correct, at its sole cost and expense, "Landlord's Work" which consists of the following work on Parcel 6: the parking, landscaping and site lighting work in the southeast corner of Parcel 6 as shown on the Parcel 6 Drawing A-1.1 dated 01-12-01, a copy of which is attached hereto as <u>Schedule 1</u>, and the items set forth on <u>Schedule 2</u> attached hereto.

1.3 <u>Tenant's Work</u>. **"Tenant's Work**" will include designing, providing and installing all Tenant Improvements (defined hereafter) and providing the required furnishings, fixtures and equipment for Tenant's use of the Premises, as expanded by the Additional Premises. As used in the Lease, the term "**Tenant Improvements**" shall mean all improvements set forth in the Approved Tenant Improvement Plans. Tenant shall obtain all necessary permits and approvals for and shall construct the "**Tenant Improvements**" in accordance with the Approved Tenant Improvement Plans (as defined in Section 2.4 below). Landlord will disburse the Tenant Improvement Allowance described in Section 4 below to pay for Tenant Improvement Costs (defined hereafter). All such

costs of completing Tenant's Work which are in excess of the Tenant Improvement Allowance shall be paid by Tenant pursuant to the terms of the Lease and this Work Letter.

SECTION 2

TENANT IMPROVEMENT PLANS

Architect/Construction Plans. Tenant has retained CAS Architects, Inc. (the "Architect") to prepare the construction plans for all Tenant 2.1 Improvements and Special Tenant Improvements. Tenant shall retain engineering consultants (the "Engineers") approved by Landlord (which approval shall not be unreasonably withheld or delayed) to prepare all plans and engineering working drawings relating to the Tenant Improvements, including without limitation, all structural, HVAC, electrical, plumbing, life safety, and sprinkler work in the Premises which is not part of the Base Building. The final working plans and drawings to be prepared by Architect and the Engineers hereunder shall be known collectively as the "Tenant Improvement Plans". The scope, form and content of all plans and drawings shall be discussed in reasonable detail at each of the weekly meetings held pursuant to the terms of Section 3.2.6 below. All Tenant Improvement Plans shall be in a form suitable for bidding and construction by qualified contractors, shall meet the requirements of the City of Livermore, and shall be subject to Landlord's approval, which shall not be unreasonably withheld or delayed. Tenant and Architect shall verify. in the field. the dimensions and conditions as shown on the relevant portions of the Base Building plans, and Tenant and Architect shall be solely responsible for the same, and Landlord shall have no responsibility in connection therewith. Landlord's review of the Tenant Improvement Plans as set forth in this Section 2, shall be for its sole purpose and shall not obligate Landlord to review the same, for quality, design, code compliance or other like matters. Accordingly, notwithstanding that any Tenant Improvement Plans are reviewed by Landlord or its architect, engineers and consultants, and notwithstanding any advice or assistance which may be rendered to Tenant by Landlord or Landlord's space planner, architect, engineers, and consultants, Landlord shall have no liability whatsoever in connection therewith and shall not be responsible for any omissions or errors contained in the Tenant Improvement Plans, and Tenant's waiver and indemnity set forth in Article 16 of the Lease shall specifically apply to the Tenant

Improvement Plans.

2.2 <u>Final Design Drawings</u>. Tenant and the Architect shall prepare the final design drawings and specifications for Tenant Improvements in the Premises (collectively, the "**Final Design Drawings**") and shall deliver the same to Landlord for Landlord's approval. The Final Design Drawings shall include a layout and designation of all offices, rooms and other partitioning. Landlord may request clarification or more specific drawings for special use items not included in the Final Design Drawings. Landlord shall advise Tenant within five (5) business days after Landlord's receipt of the Final Design Drawings for the Premises if the same are unsatisfactory or incomplete in any respect. If Tenant is so advised, Tenant shall promptly cause the Final Design Drawings to be revised to correct any deficiencies or other matters Landlord may reasonably require.

2.3 <u>Final Working Drawings</u>. After the Final Design Drawings have been approved by Landlord, Tenant shall promptly cause the Architect and the Engineers to complete the architectural and engineering drawings for the Premises, and Architect shall compile a fully coordinated set of architectural, structural, mechanical, electrical and plumbing working drawings in a form which is sufficiently complete to allow subcontractors with a reasonable level of competence and experience in the industry to bid on the work and to obtain all permits required for the construction of the Tenant Improvements (the "**Permits**") and shall submit the same (collectively, the "**Final Working Drawings**") to Landlord for Landlord's approval, which approval shall not be unreasonably withheld or delayed. Tenant shall supply Landlord with three (3) copies signed by Tenant of such Final Working Drawings. Landlord shall advise Tenant within five (5) business days after Landlord's receipt of the Final Working Drawings for the Premises if the same are unsatisfactory or incomplete in any respect. If Tenant is so advised, Tenant shall promptly cause the Final Working Drawings to be revised in accordance with such review and any disapproval of Landlord in connection therewith.

2.4 <u>Approved Tenant Improvement Plans</u>. The Final Working Drawings shall be approved by Landlord (the "**Approved Tenant Improvement Plans**") prior to the commencement of construction of the Tenant Improvements by Tenant. In order to expedite the permitting process, however, prior to Landlord's approval pursuant to Section 2.3 above, Tenant may submit the Final Working Drawings to the appropriate municipal authorities for all Permits necessary to allow Tenant's contractor to commence and fully complete the construction of the Tenant Improvements. Notwithstanding the foregoing, Tenant acknowledges that Landlord does not waive the right to approve the Final Working Drawings and by electing to submit the Final Working Drawings for permit prior to Landlord's approval, Tenant is assuming the risk that Landlord may require changes in such drawings after the same have been submitted for permits. Tenant hereby agrees that neither Landlord nor Landlord's consultants shall be responsible for obtaining any of the Permits or a certificate of occupancy for the Premises and that obtaining the same shall be Tenant's responsibility; provided, however, that Landlord shall cooperate with Tenant in executing permit applications and performing other ministerial acts reasonably necessary to enable Tenant to obtain any such permit or certificate of occupancy. No changes, modifications or alterations in the Approved Tenant Improvement Plans may be made without the prior written consent of Landlord, which consent shall not be unreasonably withheld or delayed.

SECTION 3

CONSTRUCTION OF TENANT IMPROVEMENTS

3.1 <u>Tenant's Selection of Contractors</u>.

3.1.1 <u>Tenant's Contractor</u>. A general contractor shall be retained by Tenant to construct the Tenant Improvements. Such general contractor ("**Tenant's Contractor**") shall be subject to Landlord's prior approval, which approval shall not be unreasonably withheld or delayed.

3.1.2 <u>Tenant's Agents</u>. All subcontractors, laborers, materialmen, and suppliers used by Tenant (such subcontractors, laborers, materialmen, and suppliers, and Tenant's Contractor to be known collectively as "**Tenant's Agents**") must be approved in writing by Landlord, which approval shall not be unreasonably withheld or delayed. If Landlord does not approve any of Tenant's proposed subcontractors, laborers, materialmen or suppliers, Tenant shall submit other proposed subcontractors, laborers, materialmen or suppliers for Landlord's written approval. Notwithstanding the foregoing, Tenant shall retain subcontractors designated or reasonably approved by Landlord in connection with any structural, mechanical, electrical, plumbing or heating, air-conditioning or ventilation work to be performed in the Premises. Further Landlord's approval shall not be required for any of Tenant's Agents performing work or providing materials costing less than \$10,000.

3.2 <u>Construction of Tenant Improvements by Tenant's Agents</u>.

3.2.1 <u>Tenant's Construction Contract; Cost Budget</u>. Prior to Tenant's execution of the construction contract and general conditions with Tenant's Contractor ("**Tenant's Construction Contract**"), Tenant shall submit Tenant's Construction Contract to Landlord for its approval, which approval shall not be unreasonably withheld or delayed. Prior to the commencement of the construction of the Tenant Improvements, and after Tenant has accepted all bids for the Tenant Improvements, Tenant shall provide Landlord with a detailed breakdown, by trade, of the final costs to be incurred or which have been incurred in connection with the design and construction of the Tenant Improvements to be performed by or at the direction of Tenant or Tenant's Contractor, which costs form a basis for the amount of Tenant's Construction Contract (the "Final Costs"). If the costs of the Tenant Improvements exceed the Tenant Improvement Allowance, Tenant shall also provide Landlord with its computation of percentage that the Tenant Improvement Allowance bears to the total costs of the Tenant Improvements (the "**Allowance Percentage**")

3.2.2 <u>Tenant's Agents</u>.

A. Landlord's General Terms for Tenant's Agents and Tenant Improvement Work. Tenant's and Tenant's Agent's construction of the Tenant Improvements shall comply with the following: (i) the Tenant Improvements shall be constructed in accordance with the Approved Tenant Improvement Plans; (ii) Tenant's Agents shall submit schedules of all work relating to the Tenant Improvements to Tenant's Contractor and Tenant's Agents of any changes which are necessary thereto, and Tenant's Agents shall adhere to such corrected schedule; and (iii) Tenant shall abide by all rules made by Landlord with respect to storage of materials, coordination of work with the contractors of other tenants and Landlord, and any other matter in connection with this Work Letter, including, without limitation, the construction of the Tenant Improvements.

B. <u>Indemnity</u>. Tenant's indemnity of Landlord as set forth in <u>Section 16.1</u> of the Lease shall also apply with respect to any and all costs, losses, damages, injuries and liabilities related in any way to any act or omission of Tenant or Tenant's Agents, or anyone directly or indirectly employed by any of them, or in connection with Tenant's non-payment of any amount arising out of the Tenant Improvements and/or Tenant's disapproval of all or any portion of any request for payment. Such indemnity by Tenant, as set forth in <u>Section 16.1</u> of the Lease, shall also apply with respect to any and all costs, losses, damages, injuries and liabilities related in any way to Landlord's performance of any ministerial acts reasonably necessary (i) to permit Tenant to complete the Tenant Improvements, and (ii) to enable Tenant to obtain any building permit or certificate of occupancy for the Premises.

C. <u>Requirements of Tenant's Agents</u>. Each of Tenant's Agents shall guarantee to Tenant and for the benefit of Landlord that the portion of the Tenant Improvements for which it is responsible shall be free from any defects in workmanship and materials for a period of not less than one (1) year from the date of completion thereof. Each of Tenant's Agents shall be responsible for the replacement or repair, without additional charge, of all work done or furnished in accordance with its contract that shall become defective within one (1) year after the later to occur of (i) completion of the work performed by such contractor or subcontractors and (ii) the Lease Commencement Date. The correction of such work shall include, without additional charge, all additional expenses and damages incurred in connection with such removal or replacement of all or any part of the Tenant Improvements, and/or the Building and/or common areas that may be damaged or disturbed thereby. All such warranties or guarantees as to materials or workmanship of or with respect to the Tenant Improvements shall be contained in the Contract or subcontract and shall be written such that such guarantees or warranties shall incure to the benefit of both Landlord and Tenant, as their respective interests may appear, and can be directly enforced by either. Tenant covenants to give to Landlord any assignment or other assurances which may be necessary to effect such right of direct enforcement.

D. <u>Insurance Requirements</u>.

1. <u>General Coverages</u>. All of Tenant's Agents shall carry worker's compensation insurance covering all of their respective employees, and shall also carry public liability insurance, including property damage, all with limits, in form and with companies as are required to be carried by Tenant as set forth in <u>Article 14</u> of the Lease.

2. <u>Special Coverages</u>. Tenant shall carry "Builder's All Risk" insurance in an amount approved by Landlord covering the construction of the Tenant Improvements, and such other insurance as Landlord may require, it being understood and agreed

3

that the Tenant Improvements shall be insured by Landlord pursuant to <u>Article 14</u> of the Lease upon completion thereof. Tenant's policy of Builder's All Risk insurance shall be in amounts and shall include such extended coverage endorsements as may be reasonably required by Landlord including, but not limited to, the requirement that all of Tenant's Agents shall carry Excess Liability and Products and Completed Operation Coverage insurance, each in amounts not less than \$1,000,000 per incident, \$2,000,000 in aggregate, and in form and with companies as are required to be carried by Tenant as set forth in <u>Article 14</u> of the Lease.

3. <u>General Terms</u>. Certificates for all insurance carried pursuant to this Section 3.2.2(D) shall be delivered to Landlord before the commencement of construction of the Tenant Improvements and before the equipment of Tenant's Contractor is moved onto the site. All such policies of insurance must contain a provision that the company writing said policy will give Landlord thirty (30) days' prior written notice of any cancellation or lapse of the effective date or any reduction in the amounts of such insurance. In the event that the Tenant Improvements are damaged by any cause during the course of the construction thereof, Tenant shall immediately repair the same at Tenant's sole cost and expense. Tenant's Agents shall maintain all of the foregoing insurance coverage in force until the Tenant Improvements are fully completed and accepted by Landlord, except for any Products and Completed Operation Coverage insurance required by Landlord, which is to be maintained for ten (10) years following completion of the work and acceptance by Landlord and Tenant. All policies carried under this Section 3.2.2(D) shall insure Landlord and Tenant, as their interests may appear, as well as Tenant's Contractor and Tenant's Agents. All insurance, except Workers' Compensation, maintained by Tenant's Agents shall preclude subrogation claims by the insurer against anyone insured thereunder. Such insurance shall provide that it is primary insurance as respects the Landlord and that any other insurance maintained by Landlord is excess and noncontributing with the insurance required hereunder. The requirements for the foregoing insurance shall not derogate from the provisions for indemnification of Landlord by Tenant under Section 3.2.2(B) of this Work Letter. 3.2.3 <u>Governmental Compliance</u>. The Tenant Improvements shall comply in all respects with the following: (i) all building codes and other state, federal, city or quasi-governmental laws, codes, ordinances and regulations, as each may apply according to the rulings of the controlling public official, agent or other person; (ii) applicable standards of the American Insurance Association (formerly, the National Board of Fire Underwriters) and the National Electrical Code; and (iii) building material manufacturer's specifications.

3.2.4 <u>Mechanic's Liens</u>. Tenant shall keep the Premises and the Center free and clear of mechanics' liens arising out of its construction of Tenant's Work as provided in <u>Article 29</u> of the Lease. All provisions of Article 29 shall apply to Tenant's Work as if fully set forth into this Work Letter.

3.2.5 Inspection by Landlord. Landlord shall have the right to inspect the Tenant Improvements at all times, provided however, that Landlord's failure to inspect the Tenant Improvements shall in no event constitute a waiver of any of Landlord's rights hereunder nor shall Landlord's inspection of the Tenant Improvements constitute Landlord's approval of the same. Should Landlord reasonably disapprove any portion of the Tenant Improvements, Landlord shall specify the items disapproved. Any material defects or deviations in, and/or disapproval by Landlord of, the Tenant Improvements shall be rectified by Tenant at no expense to Landlord, provided however, that in the event Landlord determines that a defect or deviation exists or disapproves of any matter in connection with any portion of the Tenant Improvements and such defect, deviation or matter might adversely affect the mechanical, electrical, plumbing, heating, ventilating and air-conditioning or life-safety systems of the Building, the structure or exterior appearance of the Building or any other tenant's use of such other tenant's leased premises, Landlord after giving Tenant at least three (3) business days advance written notice and an opportunity to cure or otherwise demonstrate compliance, may take such action as Landlord deems necessary, at Tenant's expense and without incurring any liability on Landlord's part, to correct any such defect, deviation and/or matter is corrected to Landlord's satisfaction.

3.2.6 <u>Meetings</u>. Commencing upon the execution of the First Amendment, Tenant shall hold weekly meetings at a reasonable time, with the Architect and the Tenant's Contractor regarding the progress of the preparation of the Tenant Improvement Plans and the construction of the Tenant Improvements, and Landlord and/or its agents shall receive prior notice of, and shall have the right to attend, all such meetings, and, upon Landlord's request, certain of Tenant's Agents shall attend such meetings. Such meetings shall include a detailed review of the plans, drawings and specifications prepared to date and all participants in the

4

meeting shall make a good faith effort to raise any issues or concerns they may have regarding the scope, form or content of any plan submitted.

3.3 <u>Notice of Completion; Copy of Record Set of Plans</u>. Within ten (10) days after completion of construction of the Tenant Improvements, Tenant shall cause a Notice of Completion with respect to the Tenant Improvements to be recorded in the office of the Recorder of the County of Alameda in accordance with Section 3093 of the Civil Code of the State of California or any successor statute, and shall furnish a copy thereof to Landlord upon such recordation. If Tenant fails to do so, Landlord may execute and file the same on behalf of Tenant as Tenant's agent for such purpose, at Tenant's sole cost and expense. At the conclusion of construction, (i) Tenant shall cause the Architect and Tenant's Contractor (A) to update the Approved Working Drawings as necessary to reflect all changes made to the Approved Working Drawings during the course of construction, (B) to certify to the best of their knowledge that the "record-set" of mylar as-built drawings are true and correct, which certification shall survive the expiration or termination of this Lease, and (C) to deliver to Landlord two (2) sets of copies of such record set of drawings within ninety (90) days following issuance of a certificate of occupancy for the Premises, and (ii) Tenant shall deliver to Landlord a copy of all warranties, guaranties, and operating manuals and information relating to the improvements, equipment, and systems in the Premises.

SECTION 4

TENANT IMPROVEMENT ALLOWANCE; CONSTRUCTION COSTS

4.1 <u>Tenant Improvement Allowance</u>. Tenant shall be entitled to a one-time tenant improvement allowance (the "Tenant Improvement Allowance") in the amount of \$25.00 per square foot of the Premises GLA for the costs relating to the initial design and construction of the Tenant Improvements. In no event shall Landlord be obligated to make disbursements pursuant to this Work Letter in a total amount which exceeds the Tenant Improvement Allowance.

4.2 Disbursement of the Tenant Improvement Allowance.

4.2.1 <u>Tenant Improvement Allowance Items</u>. Except as otherwise set forth in this Work Letter, the Tenant Improvement Allowance shall be disbursed by Landlord only for the following items and costs (collectively the "Tenant Improvement Allowance Items"):

- A. Payment of the fees of the Architect and the Engineers
- B. The payment of plan check, permit and license fees relating to construction of the Tenant Improvements;

C. The cost of construction of the Tenant Improvements, including, without limitation, testing and inspection costs, utility usage, trash removal costs, and contractors' fees and general conditions;

D. The cost of any changes in the Base Building when such changes are required by the Tenant Improvement Plans, such cost to include all direct architectural and/or engineering fees and expenses incurred in connection therewith; and

- E. The cost of any changes to the Tenant Improvement Plans or Tenant Improvements required by Code.
- F. Any costs payable to Landlord pursuant to Section 4.2.2 (D) below.

4.2.2 <u>Disbursement of Tenant Improvement Allowance</u>. During the construction of the Tenant Improvements, Landlord shall make monthly disbursements of the Tenant Improvement Allowance for Tenant Improvement Allowance Items:

A. <u>Monthly Disbursements</u>. No more frequently than monthly, during the construction of the Tenant Improvements (or such other date as Landlord may designate), Tenant shall deliver to Landlord: (i) a request for payment of Tenant's Contractor, approved by Tenant, in a form to be provided by Landlord, showing the schedule, by trade, of percentage of completion of the Tenant Improvements in the Premises, detailing the portion of the work completed and the portion not completed; (ii) invoices from all of Tenant's Agents, for labor rendered and materials delivered to the Premises; (iii) executed mechanic's lien releases from all of Tenant's Agents

which shall comply with the appropriate provisions, as reasonably determined by Landlord, of California Civil Code Section 3262(d); and (iv) all other information reasonably requested by Landlord. Tenant's request for payment shall be deemed Tenant's approval of the work furnished and/or the materials supplied as set forth in Tenant's payment request. After receipt of the foregoing and provided that Landlord does not dispute any request for payment based on non-compliance of any work with the Approved Tenant Improvement Plans, or due to any substandard work, or for any other reason, Landlord shall deliver a check to Tenant in an amount equal to the lesser of: (a) the Allowance Percentage of the Tenant Improvement costs for which payment is so requested less a ten percent (10%) retention (the aggregate amount of such retentions to be known as the "**Final Retention**"), or (b) the balance of the undisbursed Tenant Improvement Allowance (excluding the Final Retention). Landlord's payment of such amounts shall not be deemed Landlord's approval or acceptance of the work furnished or materials supplied as set forth in Tenant's payment request.

B. <u>Final Retention</u>. Subject to the provisions of this Work Letter, a check for the Final Retention shall be delivered by Landlord to Tenant within fifteen (15) days after all the following conditions have been satisfied: (i) Tenant delivers to Landlord a properly executed unconditional waiver and lien release from Tenant's Contractor in compliance with applicable sections of the California Civil Code, (ii) Landlord has determined that no substandard work exists, (iii) Architect delivers to Landlord a certificate, in a form reasonably acceptable to Landlord, certifying that the construction of the Tenant Improvements in the Premises has been substantially completed, (iv) Tenant has recorded a Notice of Completion with respect to Tenant's Work and thirty (30) days has expired from the date of such recording with no liens having been filed during such thirty day period, and (v) there is no existing Event of Default by Tenant under the Lease nor has Landlord notified Tenant of any default which would become an Event of Default under the Lease with the passage of time if not cured.

C. <u>Other Terms</u>. Landlord shall only be obligated to make disbursements from the Tenant Improvement Allowance to the extent costs are incurred by Tenant for Tenant Improvement Allowance Items. No disbursements shall be made to pay for Special Improvements unless the Tenant Improvement Allowance exceeds the costs of all other Tenant Improvements. All costs of completing Tenant's Work in excess of the Tenant Improvement Allowance shall be at Tenant's sole cost and expense. Notwithstanding anything to the contrary herein, Tenant shall not be required to commence or complete the installation of office cubicles down on Tenant's Plans to obtain the Final Retention if the cost of Tenant's Improvements excluding the office cubicle installation work exceeds the Tenant Improvement Allowance.

D. Other Landlord Costs. Tenant shall also be responsible for the payment of up to \$10,000 of the following costs: (i) the fees incurred by Landlord for Landlord's consultants in connection with design drawing review and routine construction support related to the Tenant Improvements, (ii) the cost of documents and materials supplied by Landlord and Landlord's consultants, and (iii) all other verifiable, directly related costs, such as blueprint costs and delivery, fax and copy charges incurred by Landlord and Landlord's consultants related to the design/routine construction support of the Tenant Improvement. When the Tenant Improvement Plans have been approved by Landlord, Landlord shall submit to Tenant Landlord's estimate of the foregoing costs. The Tenant Improvement Allowance shall be used to pay the foregoing costs. Landlord shall submit invoices for Landlord's actual and documented costs to Tenant and pay the same as incurred from the Tenant Improvement Allowance.

SECTION 5

CONSTRUCTION DELAYS

5.1 <u>Tenant Delays</u>. As used in this Lease, the term "**Tenant Delay**" shall mean the period of an actual delay or delays in the Substantial Completion of Landlord's Work or in the occurrence of any of the other conditions precedent to the Delivery Date, as set forth in <u>Article 8</u> of the Lease, to the extent resulting from:

a. Tenant's failure to approve any matter requiring Tenant's approval within the time period specifically provided in this Work Letter for such approval;

b. A breach by Tenant of the terms of this Work Letter or the Lease;

c. Changes to the Base Building work described in the Plan List required by the Approved Tenant Improvement Drawings or requested in writing by Tenant; or

d. Any other acts or omissions of Tenant, or its agents, or employees,

6

(each a "Tenant Delay"). Landlord shall provide prompt (within 48 hours of becoming aware of any such delay) written notice to Tenant ("Delay Notice") specifying the action or inaction which Landlord contends constitutes a Tenant Delay hereunder. The period of delay, however, shall commence to run on the date of the action or inaction and not on the date of the Delay Notice. To the extent an action or inaction by Tenant specified in any Delay Notice constitutes a Tenant Delay as defined above and actually results in a delay in the Substantial Completion of the Premises (after taking into account any delays resulting from Landlord Delays and/or Force Majeure Delays described below), a Tenant Delay shall be deemed to have been established.

5.2 <u>Tenant's Lease Default</u>. Notwithstanding any provision to the contrary contained in this Lease: (i) if an Event of Default as described in Article 25 of the Lease has occurred or, (ii) a default by Tenant under this Work Letter has occurred at any time on or before the substantial completion of Landlord's

Work and Tenant fails to remedy the default within such 48 hours after written notice from Landlord, then Landlord may thereafter: (i) in addition to all other rights and remedies granted to Landlord pursuant to the Lease, Landlord shall have the right to withhold payment of all or any portion of the Tenant Improvement Allowance and/or Landlord may cause Contractor to cease the construction of the Premises (in which case, Tenant shall be responsible for any Tenant Delay resulting from such work stoppage as set forth in Section 5.1 above of this Work Letter), and (ii) all other obligations of Landlord under the terms of this Work Letter shall be deferred until such time as such default is cured pursuant to the terms of the Lease.

5.3 Landlord Delay. As used herein, "Landlord Delay" shall mean: (i) any actual delay in the completion of the work Tenant is required to perform hereunder which results from any failure of Landlord to act or provide approvals within five (5) business days or (ii) the actual delay in the Substantial Completion of Landlord's Work due to any failure of Landlord, its agents, employees or contractors to perform the Base Building work or other work required to be provided by Landlord hereunder in compliance with the terms hereof and in compliance with applicable laws, rules and regulations or any other acts or omissions of Landlord, or its agents, or employees. Tenant shall provide prompt (within 48 hours of becoming aware of any such delay) written notice to Landlord ("Delay Notice") specifying the action or inaction which Tenant contends constitutes a Landlord Delay hereunder. The period of delay, however, shall commence to run on the date of the action or inaction and not on the date of the Delay Notice.

5.4 <u>Force Majeure Delays</u>. The term "**Force Majeure Delays**" shall mean delays caused by any event of force majeure described in Section 32.8 of the Lease.

5.5 <u>Substantial Completion</u>. The date of July 15, 2007 for Landlord's Substantial Completion of the Landlord's Work shall be extended for the period of any Tenant Delays and Force Majeure Delays.

SECTION 6

MISCELLANEOUS

6.1 <u>Tenant's Representative</u>. Tenant has designated Ed Traverso and Alan Moyal as its sole representatives with respect to the matters set forth in this Work Letter, each of whom, until further notice to Landlord, shall have full authority and responsibility to act on behalf of the Tenant as required in this Work Letter.

6.2 <u>Landlord's Representative</u>. Landlord has designated William Drummond as its sole representative with respect to the matters set forth in this Work Letter, who, until further notice to Tenant, shall have full authority and responsibility to act on behalf of the Landlord as required in this Work Letter.

6.3 <u>Time of the Essence in This Work Letter</u>. Unless otherwise indicated, all references herein to a "number of days" shall mean and refer to calendar days.

7

SCHEDULE 1 TO EXHIBIT B

PARCEL 6 DRAWING A-1.1 DATED 01-12-01

[Parcel 6 Drawing]

8

SCHEDULE 2 TO EXHIBIT B

ADDITIONAL LANDLORD'S WORK

[Landlord's Additional Work]

9

EXHIBIT B-1

DESCRIPTION OF TENANT IMPROVEMENT PLANS

[Tenant Improvement Plans – Descriptions]

LEASE PROVISIONS REGARDING HAZARDOUS SUBSTANCES

[Hazardous Substances – Lease Provisions)

From and after the First Amendment Effective Date, Section 4.7 of the Lease is amended its entirety and the following Section 4.7 substituted therefor:

4.7 Hazardous Substances:

A. **Hazardous Substance; Reportable Uses:** As used herein, the terms "**Hazardous Substance**" and "**HS**" shall mean any product, substance, chemical, material or waste whose presence, nature, quantity and/or intensity of existence, use, manufacture, disposal, transportation, spill, release or effect, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentially injurious to the public health, safety or welfare, the environment, or the Premises; (ii) regulated or monitored by any governmental authority; or (iii) a basis for potential liability of Landlord to any governmental agency or third party under any applicable statute. Hazardous Substance shall include, but not be limited to, hydrocarbons, petroleum, gasoline, crude oil or any products or by-products thereof. "**Reportable Use**" shall mean (i) the installation or use of any above or below ground storage tank, (ii) the generation, possession, storage, use, transportation, or disposal of a Hazardous Substance that requires a permit from, or with respect to which a report, notice, registration or business plan is required to be filed with, any governmental authority, and (iii) the presence in, on or about the Premises of a Hazardous Substance with respect to which any Applicable HS Requirements (as defined in subparagraph (F) hereinafter) require that a notice be given to persons entering or occupying the Premises or neighboring properties.

B. Tenant's Use of Hazardous Substances:

(1) <u>Notice of Use of Hazardous Substances</u>. Tenant may, without Landlord's prior consent, but upon notice to Landlord and in compliance with all Applicable HS Requirements and all other provisions of this <u>Section 4.7</u>, at Tenant's sole cost and expense, (i) operate a business on the Premises which is substantially similar to the business it is operating at its facilities in Livermore, California as of the Commencement Date, i.e. research, development, design, manufacture (including with clean room facilities), and sale of components and devices relating to the testing and packaging of semiconductor devices ("**Permitted Use**"), and (ii) use any ordinary and customary Hazardous Substances reasonably required to be used by Tenant in the normal course of the Permitted Use.

(2) <u>Tenant's HS Use</u>. Tenant shall have the right to use the Hazardous Substances listed on Exhibit F without Landlord's prior consent and without the requirement of additional insurance. Tenant shall use all such Hazardous Substances in accordance with all Applicable HS Requirements and in compliance with all other provisions of this Section 4.7, specifically including the notice requirements and restrictions set forth below. Tenant's use of the substances referenced in Exhibit F may be referred to herein as "**Tenant's HS Use**".

(3) <u>Control of HS Hazards.</u>

(a) <u>Plans for Designated HS Areas</u>. Tenant shall use, store, or otherwise manage HS only in areas designated by Tenant for such use ("**Designated HS Areas**"). Prior to commencement of Tenant's HS Use on the Premises, and prior to modification of or addition to any Designated HS Areas, Tenant shall provide Landlord with written plans (such as architectural or engineering plans) regarding the design and planned operation of the Designated HS Areas. The plans shall include descriptions of the types and quantities of HS that will be used, stored, or otherwise managed in Designated HS Areas, the maximum design capacity of each Designated HS Area and descriptions of all equipment and structures that will be used to control environmental, health, and safety hazards associated with the HS, including, for example, secondary containment structures

11

and air pollution control equipment. Tenant will also provide copies of all permits and other approvals required to be obtained to lawfully operate Tenant's business and Hazardous Substances on the Premises.

(b) <u>Commencement of Tenant's HS Use</u>. Tenant shall not commence Tenant's HS Use until Landlord has approved the plans submitted by Tenant pursuant to subparagraph (a) above, which approval shall not be unreasonably withheld or delayed. Landlord may (but without any obligation to do so) condition its approval upon Tenant's taking such measures as Landlord, at its reasonable discretion, deems necessary to protect itself, the public, the Premises, the Center, and the environment against damage, contamination, injury, and/or liability, including, but not limited to the installation (and, at Landlord's option, removal on or before Lease expiration or earlier termination) of reasonably necessary protective equipment, structures, or modifications to the Premises. Tenant's plans shall be deemed approved, however, if Tenant's plans comply with the requirements of subparagraph (a) and the minimum standards set forth on Exhibit F-1.

(c) <u>Modification/Expansion of Designated HS Areas</u>. Tenant shall not modify or add to the Designated HS Areas until Landlord has approved the plans submitted by Tenant pursuant to subparagraph (a) above for the modification or addition, which approval shall not be unreasonably withheld or delayed. Tenant's plans shall be deemed approved, however, if Tenant's plans for the modification or addition comply with the requirements of subparagraph (a) and the minimum standards set forth on <u>Exhibit F-1</u>.

- (4) <u>Notice of HS Use.</u> Tenant shall notify the Landlord in writing at least five (5) business days prior to any of the following:
 - (a) the date Tenant first commences Tenant's HS Use on the Premises; or

(b) the date Tenant commences to store or use any Hazardous Substance which is not listed on Exhibit F (a "**New HS**"), if the quantity of the New Hazardous Substance exceeds either (i) 55 gallons of liquid, 500 pounds of solid, 200 cubic feet of compressed gas at standard temperature and pressure, or (ii) the applicable Threshold Planning Quantity listed in 40 CFR Part 355.

After receipt of a notice pursuant to subparagraph (b) above, if Tenant's use of the New HS in the Premises is materially more dangerous than Tenant's use of Hazardous Substances listed on <u>Exhibit F</u>, Landlord may require Tenant to obtain a policy of pollution liability insurance in a commercially reasonable form and amounts and with such insurer as may be reasonably approved by Landlord. For any insurance policy requirement, Landlord shall be named as an additional insured under such policy. Tenant shall deliver a certificate of any insurance required prior to bringing the Hazardous Substance into

the Premises and Tenant shall maintain such insurance in effect until the closure requirements set forth in subparagraph (H) below have been satisfied or the New HS use ceases.

(5) <u>Contents of New HS Notice</u>. Each notice of a New HS shall specify the names and quantities of any New HS that Tenant intends to place on the Premises which exceeds the quantities described in subparagraph 4(b) above together with a copy of all permits and other approvals required to be obtained to lawfully use, store, or otherwise manage the New HS on the Premises. Tenant's notice shall also provide Landlord with information regarding the Designated HS Areas where the New HS will be used, stored, or otherwise managed, the new aggregate quantities of all Hazardous Substances in Designated HS Areas, and the maximum design capacities of the Designated HS Areas (if changed or modified from the Designated HS Areas as initially approved consistent pursuant to <u>Section 4.7(B)(3)(b)</u> above).

(6) Increase in HS Quantities. If, at any time during the Term, Tenant intends to increase the quantity of existing Hazardous Substances and/or add New HS

12

such that the aggregate quantity of all Hazardous Substances in any Designated HS Area on the Premises exceeds the maximum design capacity for the Designated HS Area, Tenant shall not increase quantities or add New HS until Landlord has consented to the modification of or addition to the Designated HS Areas, pursuant to Section 4.7(B)(3)(c) above.

(7) <u>Restrictions on Quantity or Use of HS</u>. Notwithstanding any other provision of this Lease, but subject to Tenant's right to engage in a Permitted Use consistent with the standards of <u>Exhibit F-1</u>, Tenant's use of Hazardous Substances at the Premises is subject to the following restrictions:

- (a) Tenant shall not, without Landlord's consent, use any HS in quantities such that Tenant would be subject to requirements for preparation of a Risk Management Plan, as set forth in 40 CFR Part 68 (as such requirements exist on the date of execution of this Lease without regard to amendments which may be enacted after the date hereof) and such HS use is materially more dangerous than the HS use presently being carried on by Tenant.
- (b) Tenant shall not, without Landlord's consent, use any HS which emits odors unless the odors can be controlled to the extent they are not present at objectionable levels in any areas exterior to the Premises that are accessible to other tenants of the Center or the general public. In the absence of any legal thresholds for identifying objectionable odors, other odor standards may be used, provided they are generally accepted as being scientifically valid.
- (c) Tenant shall not, without Landlord's consent, use any HS in a manner that would result in "Significant Emissions". Significant Emissions are defined as air emissions originating from the Premises for which under applicable federal or state law (i) notices or warnings must be given to other occupants of the Center or the general public based upon their proximity to the Building, as opposed to entry therein, or (ii) other occupants of the Center or the general public must receive special training and/or use personal protective equipment.

C. **Plans/Reports**: Within ten (10) days after Tenant submits the same to any governmental authority, Tenant shall provide Landlord with copies of all hazardous materials business plans, permits and all other plans, reports and correspondence pertaining to storage/management of Hazardous Substances at the Premises, except waste manifests and routine monitoring reports.

D. **Duty to Inform Landlord:** If Tenant knows, or has reasonable cause to believe, that a Hazardous Substance has come to be located in, on, under or about the Premises or Building or Center, other than as previously permitted or consented to by Landlord or there has been a spill, release or discharge of any Hazardous Substances in the Premises (other than discharges permitted, authorized or otherwise approved by the applicable governmental agencies regulating the same), Tenant shall immediately give Landlord written notice thereof, together with a copy of any statement, report, notice, registration, application, permit, business plan, license, claim, action, or proceeding given to, or received from, any governmental authority or third party concerning the presence, spill, release, discharge of, or exposure to, such Hazardous Substance including but not limited to all such documents as may be involved in any Reportable Use involving the Premises. Tenant shall not cause or permit any Hazardous Substance to be spilled or released in, on, under or about the Premises (including, without limitation, through the plumbing, storm, or sanitary sewer system).

13

E. **Indemnification:** Tenant shall indemnify, protect, defend and hold Landlord, its agents, employees, lenders, and the Premises and Center, harmless from and against any and all damages, liabilities, judgments, costs, claims, liens, expenses, penalties, loss of permits and attorneys' and consultants' fees arising out of or involving any Hazardous Substance to the extent brought into the Premises and/or Center by or for Tenant, its employees, agents or contractors. Tenant's obligations under this Section 4.7(E) shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Tenant, and, except as otherwise provided in Section 4.7(G), the cost of investigation (including reasonable consultants' and attorneys' fees and testing), removal, remediation, restoration and/or abatement thereof, or of any contamination therein involved, and shall survive the expiration or earlier termination of this Lease. No termination, cancellation or release agreement entered into by Landlord and Tenant shall release Tenant from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Landlord in writing at the time of such agreement.

F. **Tenant's Compliance with Requirements:** Tenant shall, at Tenant's sole cost and expense fully, diligently and in a timely manner, comply with all "**Legal Requirements**", which term is used in this Lease to mean all laws, rules, regulations, ordinances, directives, covenants, easements and restrictions of record, permits, the requirements of any applicable fire insurance underwriter or rating bureau, relating in any manner to the Premises or Center (including but not limited to matters pertaining to (i) industrial hygiene, (ii) environmental conditions on, in, under or about the Premises, including soil and groundwater conditions, and (iii) the use, generation, manufacture, production, installation, maintenance, removal, transportation, storage, spill or release of any Hazardous Substance, which foregoing (ii) and (iii) Legal Requirements may be referred to as "**Applicable HS Requirements**"), now in effect

or which may hereafter come into effect. Tenant shall, within twenty (20) business days after receipt of Landlord's written request made from time to time, provide Landlord with copies of all documents and information, including but not limited to permits, registrations, manifests, applications, reports and certificates, evidencing Tenant's compliance with all Applicable HS Requirements specified by Landlord, and shall within five (5) business days after receipt, notify Landlord in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, complaint or report pertaining to or involving failure by Tenant or the Premises to comply with any Legal Requirements. Tenant shall be obligated to disclose to Landlord which Hazardous Substances are used at the Premises and how such Hazardous Substances are being handled (but in no event shall Tenant be required to disclose information regarding formulations or manufacturing processes or procedures related to such Hazardous Substances) notwithstanding that such information may be proprietary information or a trade secret. Landlord agrees to keep as confidential all such proprietary information delivered to Landlord (including, without limitation, Exhibit F) and which Tenant designates in writing as confidential, provided that Landlord may disclose the same when required by law or in litigation between Landlord and Tenant regarding such information or to Landlord's lenders or to prospective purchasers provided such parties have also agreed to keep the same confidential.

G. **Compliance with Law Governing Hazardous Substances:** Landlord, Landlord's agents, employees, contractors and designated representatives, and the holders of any mortgages, deeds of trust or ground leases on the Premises ("**Lenders**") shall have the right to enter the Premises at any time in case of an emergency, and otherwise at reasonable times (but not more often than annually for inspection of Tenant's "clean room" on the Premises, if any, or more often than quarterly for inspection of other parts of the Premises), and upon no less than 10 days' notice, unless an emergency exists, for the purpose of inspecting the condition of the Premises and for verifying compliance by Tenant with this Lease and all Legal Requirements, and Landlord shall be entitled to employ experts and/or consultants in connection therewith(provided that such experts and/or consultants are not engaged in a business competitive with Tenant, or consult or give advice to any competitor of Tenant listed on <u>Exhibit I</u>) to advise Landlord with respect to Tenant's activities, including but not limited to Tenant's installation, operation, use, monitoring, maintenance, or removal of any Hazardous Substance on or from the Premises ("**Landlord's Consultants**"). Prior to engaging any Landlord's Consultants,

14

Landlord shall provide Tenant with written notice of the name of the proposed consultant and Tenant shall have five (5) business days to object to the engagement based upon Tenant's reasonable belief that engagement of the particular individual as Landlord's Consultant, and the consequent access to Tenant's facilities and proprietary information and trade secrets, could result in competitive injury to Tenant. Landlord shall not engage with a consultant as to whom Tenant has objected. Tenant shall cooperate with Landlord's Consultants inspecting the Premises, including responding to interviews (for a time period not to exceed four (4) hours for the initial site visit and two (2) hours for site visits thereafter). Landlord's Consultants shall at all times be escorted by Tenant, unless Tenant agrees otherwise. This and all rights to enter except in the event of an emergency are subject to Landlord, Landlord's agents, employees, contractors, designated representatives, prospective purchasers and/or Lenders, as the case may be, executing Tenant' s standard non-disclosure agreement in the form attached hereto as <u>Exhibit H</u>. The costs and expenses of any such inspections shall be paid by the party requesting same and in no event shall be borne by or passed along to Tenant unless requested by Tenant, subject only to the proceeding sentence. If the inspection is performed due to a violation of Applicable HS Requirements, Tenant shall, upon request, reimburse Landlord or Landlord's Lender, as the case may be, as additional rent, for the costs and expenses of such inspections.

H. **Closure Requirements:** Prior to any termination of the Lease, Tenant, at its sole cost and expense(except as to those costs and expenses arising out of actions undertaken by Landlord or by a third party on behalf of Landlord), shall satisfy the following closure requirements with respect to the Hazardous Substances Tenant has used in the Premises during the Term:

(1) Comply with all applicable federal, state and local closure requirements with respect to Hazardous Substances;

(2) Prepare a closure plan (the "**Closure Plan**") that specifies the final disposition of all Hazardous Substances and equipment which may be contaminated with Hazardous Substances; cleaning and decontamination activities, and confirmation sampling (e.g. wipe samples, soil/ground water samples and/or indoor air quality samples, to the extent warranted by the site conditions then existing).

(3) At least sixty (60) days prior to the Lease termination, provide to Landlord a copy of the Closure Plan for review and reasonable approval. Landlord may, after consultation with Tenant, require modification of the Closure Plan to include additional activities, including sampling activities, if the site conditions indicate that there is a reasonable probability that "Significant Residual Contamination" is present. **Significant Residual Contamination** shall mean residual contamination which: (i) exceeds standards or guidance levels typically used by regulatory agencies in California for evaluating potential threats to human health or the environment; or (ii) would result in notification requirements under applicable state law of potential health risks to individuals on the Premises, other tenants of the Center, and/or the general public; or (iii) would result in potential environmental liability to Tenant or Landlord; or (iv) would result in the need for conducting any type of additional decontamination activities prior to leasing the Premises to a new tenant. If Landlord fails to request modification of the Closure Plan within ten (10) business days after its receipt thereof, Tenant's Closure Plan shall be deemed accepted.

(4) Notify Landlord of closure schedule and allow access to Landlord and/or Landlord's Consultants for inspections prior to commencing and following completion of the cleaning/decontamination activities.

(5) Notify Landlord of all sample analysis results, if any. Landlord may require additional closure activities if sampling results sampling results disclose Significant Residual Contamination.

(6) Prepare and provide to Landlord closure report documenting closure activities consistent with the Closure Plan and sample results, if any, following completion of all closure activities.

Closure shall be deemed to be complete upon Landlord's reasonable approval of the closure report and, if applicable, Landlord's receipt of a copy of the written closure approval from the local environmental agency with jurisdiction over the Hazardous Substances at the Premises.

See Addendum A-4.7.

16

EXHIBIT D

LOCATION OF PARKING SPACES ALLOCATED TO BUILDING 4 TENANT

[Building 4 - Location of Parking Spaces]

17

EXHIBIT F (Replacement)

DESCRIPTION OF TENANT'S USE OF HAZARDOUS SUBSTANCES

[Product Names of Hazardous Substances]

18

EXHIBIT I (Replacement)

LIST OF COMPETITORS

[List of Competitors]

AMST Co., Ltd. Cascade Microtech, Inc. Feinmetall GmbH Japan Electronic Materials Corporation Kulicke and Soffa SV Probe, Inc. Micronics Japan Co., Ltd. Phicom Corporation Tessera Tokyo Cathode Laboratory Co., Ltd. Tokyo Electron Ltd. Wentworth

19

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO 15 U.S.C. SECTION 7241, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Igor Y. Khandros, certify that:

- 1. I have reviewed the quarterly report on Form 10-Q of FormFactor, Inc., a Delaware corporation, for the period ended September 30, 2006, as filed with the Securities and Exchange Commission;
- 2. Based on my knowledge, the quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by the quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in the quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in the quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which the quarterly report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in the quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by the quarterly report based on such evaluation; and

(d) Disclosed in the quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2006

/s/ IGOR Y. KHANDROS

Igor Y. Khandros Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO 15 U.S.C. SECTION 7241, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Ronald C. Foster, certify that:

- 1. I have reviewed the quarterly report on Form 10-Q of FormFactor, Inc., a Delaware corporation, for the period ended September 30, 2006, as filed with the Securities and Exchange Commission;
- 2. Based on my knowledge, the quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by the quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in the quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in the quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which the quarterly report is being prepared;

(b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in the quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by the quarterly report based on such evaluation; and

(d) Disclosed in the quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2006

/s/ RONALD C. FOSTER

Ronald C. Foster Chief Financial Officer

CERTIFICATION OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of FormFactor, Inc., a Delaware corporation, for the period ended September 30, 2006, as filed with the Securities and Exchange Commission, each of the undersigned officers of FormFactor, Inc. certifies pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to his respective knowledge:

- (1) the quarterly report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) the information contained in the quarterly report fairly presents, in all material respects, the financial condition and results of operations of FormFactor, Inc. for the periods presented therein.

Date: November 7, 2006

/s/ IGOR Y. KHANDROS

Igor Y. Khandros Chief Executive Officer

Date: November 7, 2006

/s/ RONALD C. FOSTER

Ronald C. Foster Chief Financial Officer